REGULATIONS FOR ADMISSION

to Degree Programmes of Higher Education (Bachelor's Degree Programmes, Specialist's Degree Programmes, and Master's Degree Programmes) at St Petersburg University in 2022

Section 1. General Regulations for Admission to Degree Programmes of Higher Education (Bachelor's Degree Programmes, Specialist's Degree Programmes, and Master's degree programmes)

1. General Provisions

- 1.1. These Regulations for Admission to Degree Programmes of Higher Education (Bachelor's Degree Programmes, Specialist's Degree Programmes, and Master's Degree Programmes) at St Petersburg University in 2022 (hereinafter referred to as these Regulations) set out the admissions policy concerning citizens of the Russian Federation (hereinafter referred to as citizens; persons; applicants) as well as foreign nationals, stateless citizens, including compatriots residing abroad (hereinafter referred to as foreign nationals; persons; applicants), to degree programmes of higher education (bachelor's degree programmes, specialist's degree programmes, and master's degree programmes) (hereinafter referred to as degree programmes of higher education; degree programmes; programmes of higher education) at St Petersburg State University (hereinafter referred to as St Petersburg University), on the government-funded basis or on the fee-paying (contractual) basis paid for by individuals and/or legal entities. These Regulations also specify the details of holding admissions tests for physically challenged and disabled applicants.
- 1.2. These Regulations have been developed in compliance with the following documents: Federal Law No. 273-FZ 'On Education in the Russian Federation' dated 29 December 2012; Decree No 2150 'On Establishing an Education Quota for Foreign Nationals and Stateless Citizens in the Russian Federation' of the Government of the Russian Federation dated 8 October 2013; 'Procedures for Admission to Degree Programmes of Higher Education Bachelor's Degree Programmes, Specialist's Degree Programmes, and Master's Degree Programmes' approved by Order No 1076 of the Ministry of Science and Higher Education dated 21 October 2020 (hereinafter referred to as the Procedures for Admissions); Charter of Federal State Budgetary Institution of Higher Education 'St Petersburg State University'.
- 1.3. Admissions of foreign applicants, including those having equal rights with Russian citizens when applying for degree programmes of higher education, shall be governed by Section 3 of these Regulations. A Russian citizen with other non-Russian citizenship shall only be considered a Russian citizen while defining conditions for admission to St Petersburg University, except cases provided for by international treaties of the Russian Federation or federal laws.
- 1.4. An order on enrolment concerning a Russian Federation national being at the same time a foreign national that has failed to inform St Petersburg University thereof when applying for admission can be duly cancelled due to such applicant's failure to provide complete and/or accurate information of themselves to the St Petersburg University Admissions Committee (in accordance with the Sub-Clause 1, Part 2, Article 61 of Federal Law No 273-FZ 'On Education in the Russian Federation' dated 29 December 2012, such an enrolled person shall be subject to expulsion for the violation of the St Petersburg University Regulations that led to their illegal enrolment to St Petersburg University).
- 1.5. The responsibility for organising admissions to degree programmes of higher education shall rest with the St Petersburg University Admissions Committee (hereinafter referred to as the Admissions Committee) chaired by the rector of St Petersburg University. The Provisions for the Admissions Committee shall define the Admissions Committee's composition, competencies, and operating procedures.
- 1.6. The Admissions Committee shall be responsible for publishing regulations for admissions procedures to degree programmes of higher education at St Petersburg University, including those that regulate the Admissions Committee activities, as well as any required information on the St Petersburg University official website http://www.spbu.ru (hereinafter referred to as 'the St Petersburg University official website') in the Admission Committee's web page http://www.abiturient.spbu.ru. The specified

resource herein shall serve as the only official resource of information on admissions policy for degree programmes of higher education at St Petersburg University and the Admissions Committee's activities.

1.7. There are full-time and part-time enrolment types to the first year of bachelor's degree programmes, specialist's degree programmes, and master's degree programmes at St Petersburg University.

2. Applying for Admission

- 2.1. Admissions to St Petersburg University degree programmes of higher education at all levels, types and conditions of enrolment shall be conducted on a competitive basis upon personal applications and admissions tests results (unless otherwise specified by the laws of the Russian Federation and these Regulations).
- 2.2. Particular categories of persons eligible to enter St Petersburg University without taking any admissions tests to government-funded places within the special quota that enjoy a priority right to be enrolled are specified by the laws of the Russian Federation.
- 2.3. The following applicants shall have the right to be enrolled on bachelor's degree programmes and specialist's degree programmes without taking the admissions test:
- 2.3.1 winners and awardees of the final round of the All-Russian School Olympiad (hereinafter referred to as the winners and awardees of the All-Russian Olympiad), members of Russian national teams, duly formed in the manner defined by the federal executive agency in charge of the development of the governmental policy and legal regulation for general secondary education, who have participated in international olympiads in general education subjects (hereinafter referred to as the members of Russian national teams) in specialities and/or fields of study corresponding to the profile of the All-Russian School Olympiad within four (4) years following the year of the relevant olympiad;
- 2.3.2. winners and awardees of Olympic Games, Paralympic Games and Deaflympic Games, world champions, European champions, first-place finishers in the world or European championships in sports admitted to the Olympic Games, Paralympic Games and Deaflympic Games, to specialities and/or fields of study in physical culture and sports.
- 2.4. The correspondence between fields of study and specialities open for the admissions at St Petersburg University and the profile of the All-Russia School Olympiad, international olympiad in general education subjects is set forth in Appendix 8 of these Regulations.

Persons specified in Clauses 2.3 and 2.5.1 of these Regulations shall have the right within four subsequent years after the year of the relevant olympiad to exercise a certain right provided for by Clause 2.5.2 of these Regulations if the general education admissions test corresponds to the profile of the olympiad or the area within physical culture and sports. The correspondence between the All-Russia School Olympiad disciplines and the general education admissions tests is set forth in Appendix 4. The correspondence between school olympiad profiles held in accordance with the Procedures for School Olympiads (hereinafter referred to as 'the Procedures for School Olympiads') adopted by the Ministry of Education of the Russian Federation and the general education admissions tests has been set by the Procedure of providing special rights to the winners and awardees of school olympiads held in accordance with the Procedures for School Olympiads (Appendix 9 hereto).

- 2.5. By the decision of the Admissions Committee that is responsible for the admissions to bachelor's or specialist's degree programmes, the winners and awardees of school olympiads held in 2021/2022, 2020/2021, 2019/2020, 2018/2019, 2017/2018 school years and included in the list of the school olympiads for the relevant school year, approved by the Ministry of Education of the Russian Federation may enjoy one of the following particular rights (only those who were in 10th or 11th form students at the time of the olympiad):
- 2.5.1. being enrolled without any admissions tests to a bachelor's (specialist's) degree programme in the field of study correspondent to the olympiad's profile; the correspondence between St Petersburg University fields of study and the olympiad's profile shall be specified by the decision of the Admissions Committee;

- 2.5.2. being equal to the persons with maximum grades at the Unified State Examination in any general education discipline correspondent to the olympiad's profile, or to the persons who have successfully passed practical, creative admissions test and/or profession-specific admissions test in the manner specified by the Admissions Committee.
- 2.6. Decisions of the Admissions Committee mentioned in Clauses 2.5.1. and 2.5.2. herein are displayed in Appendix 9 to these Regulations.
- 2.7. The following categories of applicants shall be eligible for the admission to degree programmes of higher education to the government-funded places within the limits of the particular quotas: orphaned children or children deprived of parental care as well as persons coming from the category of orphaned children or children deprived of parental care, disabled children, persons with 1st and 2nd group of disability, those who are disabled from childhood, disabled combatants due to traumas or diseases resulted from their military service, and combat veterans specified in Sub-Clauses 1 4, Clause 1, Article 3 of Federal Law No 5-FZ 'On Veterans' dated 12 January 1995.
- 2.8. The following persons are eligible for priority enrolment when applying for admission to bachelor's or specialist's degree programmes:
- 2.8.1. Orphaned children and children deprived of parental care as well as persons coming from the category of orphaned children or children deprived of parental care;
 - 2.8.2. Disabled children, persons with 1st/2nd group of a disability;
- 2.8.3. Persons under 20 (twenty) years of age with only one parent who is a 1st group invalid, if the average family income is lower than the minimum subsistence level specified for the constituent entity of the Russian Federation where these people reside;
- 2.8.4. Persons who were exposed to radiation because of the disaster at the Chernobyl Nuclear Power Plant and are eligible to enjoy the rights under Federal Law No 1244-1 'On Social Protection of Persons Exposed to Radiation Because of the Disaster at the Chernobyl Nuclear Power Plant' dated 15 May 1991;
- 2.8.5. Children of military combatants killed on duty or died as the result of an injury (a wound, a trauma, a contusion) or diseases caused by their military service, including participation in counterterrorist operations and/or other anti-terrorism efforts;
- 2.8.6. Children of deceased (fallen) Heroes of the Soviet Union, Heroes of the Russian Federation and Full Cavaliers of the Order of Glory;
- 2.8.7. Children of employees of law enforcement agencies, Federal Service of the National Guard Forces of the Russian Federation, institutions and bodies of the penal enforcement system, Ministry of the Russian Federation for Civil Defence, Emergency Situations and Liquidation of Consequences of Natural Disasters, drugs and psychotropic substances control agencies, customs agencies, Investigative Committee of the Russian Federation who have fallen (died) as the result of an injury or other health damage received by them in the line of duty or as the result of a disease received during their service in the said institutions and agencies, as well as children dependent upon them;
- 2.8.8. Children of public prosecution officers, who have fallen (died) as the result of an injury or other health damage received by them during their public prosecution service or after dismissal as a result of personal injury related to their employment activity;
- 2.8.9. Persons on contractual military service whose term of continuous military service is three years or more, as well as conscripts with recommendations for admission from their commanders issued in the manner set forth by the federal executive authority or the federal state authority where military service is provided for by the federal law;
- 2.8.10. Persons who have been on contractual military service for three years or longer in the Armed Forces of the Russian Federation, other troops, military units and bodies, appointed at military posts and dismissed from military service at the grounds specified in Sub-clauses 'b' –'d' of Clause 1, Sub-clause 'a' of Clause 2 and Sub-clauses 'a' 'c' of Clause 3 of Article 51 of the Russian Federal Law No. 53-FZ 'On Military Duty and Military Service' dated 28 March 1998;
- 2.8.11. Disabled veterans, combatants as well as combat veterans belonging to persons specified in Sub-clauses 1-4 of Clause 1 of Article 3 of the Federal Law No. 5-FZ 'On Veterans' dated 12 January 1995;

- 2.8.12. Persons who personally participated in nuclear weapon or combat radioactive substances testing in the atmosphere, and underground nuclear weapon testing, in manoeuvres using such armoury and combat radioactive substances before the date of the actual ending of the mentioned tests and schemes, direct participants in the elimination of radiation accidents at nuclear facilities on surface vessels and submarines and other military sites, direct participants in carrying out work in collecting and disposing radioactive substances, as well as direct participants in eliminating consequences of accidents (military personnel and civilians employed by the Armed Forces of the Russian Federation, military personnel of the Ministry of Internal Affairs of the Russian Federation or federal state authorities, military personnel and employees of the National Guard Forces, persons who served in railroad troops and other military formations, employees of the law enforcement agencies of the Russian Federation, and the Ministry of the Russian Federation for Civil Defence, Emergency Situations and Liquidation of Consequences of Natural Disasters);
- 2.8.13. Military personnel, employees of the Federal Service of the National Guard Forces of the Russian Federation, employees of law enforcement agencies of the Russian Federation, penal enforcement system, the Ministry of the Russian Federation for Civil Defence, Emergency Situations and Liquidation of Consequences of Natural Disasters who have accomplished missions in the conditions of the armed conflict in the Chechen Republic and adjacent territories belonging to the armed conflict area and the said military personnel performing missions during counter-terrorist operations in the North Caucasus region;
- 2.8.14. Persons who have graduated from educational organisations, vocational educational organisations under the supervision of the federal state bodies and implementing non-degree programmes that aim to train junior students to military or state service.
- 2.9. according to the legislation of the Russian Federation, persons eligible for admission to bachelor's or specialist's degree programmes on the government-funded places at St Petersburg University without taking admissions tests as specified in Clauses 2.3 and 2.5.1 herein shall exercise the provided right of being admitted to only one educational institution of higher education and the only one-degree programme upon their choice (regardless the number of the grounds confirming the respective particular right). Under the provided specific right, the applicant must specify a particular degree programme of the highest priority they wish to be admitted. If such persons apply for admission at other educational organisations that implement degree programmes of higher education or to other bachelor's or specialist's degree programmes, they shall participate in the general competitive selection according to these Regulations. The individual rights during admissions to bachelor's or specialist's degree programmes provided in compliance with Federal Law No 273-FZ dated 29 December 2012 shall only be granted to Russian citizens. It shall not be granted to foreign nationals, including compatriots residing abroad, unless otherwise has been provided by the legislation of the Russian Federation or international treaties of the Russian Federation.
- 2.10. Admission to degree programmes of higher education at St Petersburg University shall be carried out on the grounds of a personal application addressed by the applicant to the St Petersburg University Rector. Applicants shall submit personal applications to the Admissions Committee. The application must contain the Insurance Number of the Individual Ledger Account (Insurance Number of Compulsory Pension Insurance) (hereinafter referred to as Insurance Number of the Individual Ledger Account) (if applicable).
- 2.11. Applicants shall either submit their applications electronically through the applicant's Personal Account or use 'Online Admission' Super-Service through the federal state information system the 'Unified Portal of State and Municipal Services (functions)' (hereinafter referred to as the UPSMS). To submit the application through the Personal Account, the applicants must register on the St Petersburg University official website within the deadlines set in the respective sections of these Regulations and obtain a login and a password to their Personal Account. Logins and passwords must not be disclosed to third parties. The responsibility for their Personal Account login and password security after they have obtained them shall rest with the applicant. To submit their application form, the applicant must type information in all boxes required, print the application out, and confirm the completeness and accuracy of the provided information by putting their handwritten signature.

- 2.12. The applicant shall be entitled to submit their application and other required documents as follows:
 - 2.12.1. in person;
 - 2.12.2. through general post service operators;
- 2.13.3. electronically, through the applicants' Personal Account on the St Petersburg University official website.
 - 2.13.4. electronically, through the Unified Portal of State and Municipal Services (UPSMS).

Requirements to the documents submitted by the applicants electronically and the details of the procedure of submitting the documents for the competitive selection are specified in the respective sections of these Regulations.

- 2.13. Admission applications and other documents submitted in the manner not specified by Clause 2.12. of these Regulations, including those submitted via email or fax, shall not be accepted and considered by the Admissions Committee.
- 2.14. According to the civil legislation of the Russian Federation, the applicant may have the right to capacitate any third party to submit (withdraw) documents and perform other actions that do require personal attendance during admission to degree programmes of higher education.
- 2.15. An application receipt shall be issued to the applicant (their trustee) submitting the documents in person.
- 2.16. Applicants submitting their documents via general post service shall send their documents to: St Petersburg University Admissions Committee, 7/9 Universitetskaya Embankment, St Petersburg, 199034, Russian Federation
- 2.17. The applications either sent by general post service operators or submitted electronically via the St Petersburg University official website or submitted through the UPSMS shall be accepted within the same period as those submitted in person.
- 2.18. When submitting documents by general post service, the applicant shall attach to their application other required documents specified in Clauses 2.23 2.29 herein.
- 2.19. The following documents shall confirm that the application documents sent by the applicant to the Admissions Committee by general post service have been accepted:
- 2.19.1. a return receipt signed by a St Petersburg University authorised official and carrying the date stamp of the destination post office, as well as by a signed list of enclosures having the date stamp of the post office from which the letter was sent;
- 2.19.2. an invoice (or a copy thereof) signed by a St Petersburg University authorised official (when the postal item has been delivered by express mail directly to the Admissions Committee's address).
- 2.20. The Admissions Committee shall be responsible for updating the information on the St Petersburg University official website daily at least before 6 pm Moscow Standard Time to inform the applicants in the manner set forth in Clause 1.6 herein as follows:
- 2.20.1.a list of persons whose submitted applications have been received by the Admissions Committee before the end of the previous working day and have been admitted to the competitive selection (with the specified type of competitive selection): who shall have the right to be enrolled without taking admissions tests; who shall have the right to be enrolled to the government-funded places within the admissions quotas; who shall have the priority right to be enrolled to the relevant types of study (full-time and, where applicable, part-time evening) and the relevant conditions (government-funded places or fee-paid places paid for by individuals and/or legal entities);
- 2.20.2. a list of persons whose submitted applications have been received by the Admissions Committee before the end of the previous working day and have not been admitted to the competitive selection on the grounds specified in Clause 2.32 herein;
- 2.21. The admission applications must be submitted in Russian. Documents in foreign languages must be submitted, translated into Russian, and acknowledged in the prescribed manner. Documents obtained in a foreign country must be submitted either legalised in the manner prescribed by the legislation of the Russian Federation or with Apostil affixing (except cases when the legalisation and

apostil are not required according to the provisions of the legislation of the Russian Federation and/or international treaties). By signing their application, the applicant shall confirm that:

- 2.21.1. they have read the instructions to provide accurate data and originals of documents with the admission application;
- 2.21.2. they have read these Regulations as well as other regulations and information specified in Part 2, Article 55 of Federal Law No 273-FZ 'On Education in the Russian Federation' dated 29 December 20212;
 - 2.21.3. when applying for admission to the government-funded places:
- 2.21.3.1.in bachelor's degree programmes or specialist's degree programmes, they do not possess any bachelor's degree, specialist's degree, or master's degree diploma, except those persons with higher education who apply for specialist's degree programmes in the field of study of Arts according to Part 22, Article 83 of Federal Law No 273-FZ 'On Education in the Russian Federation' dated 29 December 2012;
- 2.21.3.2. in master's degree programmes, they do not possess any specialist's degree diploma, except those applicants who have a professional specialist's degree of higher education that is confirmed by conferment on them a specialist's qualification;
 - 2.21.4. when applying for admission to bachelor's or specialist's degree programmes:
- 2.21.4.1. they apply for parallel admission to not more than five (5) educational organisations of higher education, including St Petersburg University;
- 2.21.4.2. they apply for parallel admission at St Petersburg University to not more than three (3) specialities and/or fields of study of education;
- 2.21.5. when being entitled to the admission to bachelor's degree programmes or specialist's degree programmes to the government-funded places without taking admissions tests under Part 4, Article 71 of Federal Law No 273-FZ 'On Education in the Russian Federation' or being entitled to the admission without taking admissions tests pursuant to the results of school olympiads, included in the list of the Ministry of Education of the Russian Federation:
- 2.21.5.1. they submit their admission application based on the respective particular right only to St Petersburg University;
- 2.21.5.2. they exercise their right for admission to only a particular degree programme when submitting their application to several degree programmes at St Petersburg University.
- 2.22. The applicant shall also provide their handwritten signature in their admission application to confirm their awareness of audio and video recording during admissions tests. The applicant's admission application and other documents required for the admission shall be accepted if the applicant has provided written consent for processing their personal data.
- 2.23. Applicants to St Petersburg University degree programmes of higher education are eligible to apply for admission (to take part in the competitive selection) to different types (full-time and, if applicable, part-time) and conditions (government-funded places or places on the fee-paying (contractual) basis paid for by individuals and/or legal entities) of study. The application submitted to different types and conditions of enrolment within the same educational programme shall be considered an application to take part in the competitive selection for a one-degree programme.
- 2.24. The applicant, when applying for the competitive selection for government-funded places in several degree programmes of higher education (groups of degree programmes taking into account the manner of the competitive selection organisation) as well as to different types of enrolment within the same degree programme (a group of degree programmes), must specify the admission priority for each degree programme (as well as admission priority within the group of degree programmes open for competitive selection) and, if applicable, for each type of enrolment. The applicant shall be eligible before the submission deadline to adjust their admission priority of degree programmes and the enrolment types and conditions.
- 2.25. To take part in the competitive selection for St Petersburg University's degree programmes of higher education, an applicant must submit the following documents:
- 2.25.1. A copy of an identity document with identification of their citizenship (including a copy of the passport of the Russian citizen identifying their Russian identity abroad or a copy of the document

identifying the foreign national in the Russian Federation or the identity of the stateless person in the Russian Federation according to the provisions of Federal Law No 115-FZ 'On Legal Status of Foreign Citizens in the Russian Federation' dated 25 July 2002.

- 2.25.2. An application for admission to a degree programme of the appropriate level in the format established by the Admissions Committee;
- 2.25.3. Two (2) photographs 3x4 cm or a digital image (the photographs and images must be taken in the current calendar year) for persons applying for admissions upon the results of the competitive selection held at the discretion of St Petersburg University;
- 2.25.4. An origin or a copy of the standard education credential, including education credentials obtained in a foreign country accompanied by a certificate confirming the recognition of the foreign education, except the cases when such recognition is not required according to the legislation of the Russian Federation and/or international treaties (hereinafter referred to as 'standard credential');
- 2.25.5. Documents confirming their achievements whose results shall be considered when applying for admission (at the discretion of the applicant);
- 2.25.6. A document confirming their registration in the system of the individual (personified) accounting (if applicable).
- 2.26. When submitting an admission application through the UPSMS, an applicant's identity document with citizenship identification and a document confirming their registration in the individual (personified) accounting system shall be considered accepted if the available information in the UPSMS has approved the information specified in them.
- 2.27. Persons eligible for enrolment without taking admissions tests as provided for by the legislation of the Russian Federation must provide a document confirming their eligibility for such a particular right when submitting their admission applications.

Persons participating in competitive selection within the particular admission quota or within a sponsored admission quota must provide a document confirming their eligibility for the respective right for admission within a quota when submitting their admission applications.

- 2.28. Persons eligible for the priority right must provide a document confirming their respective rights when submitting their admission applications.
- 2.29. Following the Procedures for School Olympiads, the winners and awardees of the 2021/2022, 2020/2021, 2019/2020, 2018/2019, 2017/2018 school olympiads held following the Procedures for School Olympiads and included in the List of School Olympiads duly approved by the Russian Federation Ministry of Education are eligible to submit, when applying for admission, an electronic image of the school olympiad winner's (awardee's) diploma, together with its original or a copy, generated by services of the Russian Council of School Olympiads official portal.
- 2.30. To take admissions tests for the enrolment to bachelor's or specialist's degree programmes held at the discretion of St Petersburg University, physically challenged or disabled persons must provide documents confirming their disability or the condition. To be provided with any special support when taking admissions tests, physically challenged or disabled persons or those with health restrictions must provide documents confirming their disability or the condition.
- 2.30. Disabled children, persons with 1st/2nd group of disability, persons disabled since childhood; persons disabled as a result of war injuries or diseases received during their military service eligible for admission on a government-funded basis within the established admission quota must submit, at their own discretion, either the original or a copy of the statement of physical disability.

Orphaned children, children deprived of parental care as well as those from the category of orphaned children and children deprived of parental care, war veterans from the category specified in Sub-clauses 1 - 4 Clause 1 Article 3 of Federal Law No 5-FZ 'On Veterans' dated 12 January 1995 entitled for the enrolment to government-funded places within the established quota must provide a copy of the document confirming their respective status.

- 2.32. The applicant shall be excluded from the admission to the competitive selection if they failed to:
 - 2.32.1. Submit a complete set of application documents;
 - 2.32.2. Put their handwritten signature on the admission application;

- 2.32.3. Provide legible copies of the documents (including legible scanned copies of the documents when submitting them electronically through the St Petersburg University official portal);
 - 2.32.4. Provide accurate information;
- 2.32.5. Comply with the requirements for competitive selection for degree programmes established by the legislation of the Russian Federation, Procedures for Admission and these Regulations;
- 2.32.6. Provide documents for the competitive selection within the application submission period specified by these Regulations;
- 2.32.7. Pass the admissions tests with the minimum score of points specified by the Admissions Committee confirming the successful completion of the admissions tests in a discipline when applying for a programme with the respective type (full-time or part-time) and conditions (government-funded places or places on the fee-paying (contractual) basis paid for by individuals and/or legal entities) of enrolment.
- 2.33. St Petersburg University shall undertake to validate the accuracy of the provided information in the admission application and the authenticity of the submitted documents, among other things, by contacting the relevant information systems, public and municipal administration bodies and organisations.

If applicants have submitted illegible copies of documents or the copies with significant parts of text missing (in particular, the seal, the authorised representative's signature, document details, etc., are missing or illegible), such documents shall neither be accepted nor considered.

Pursuant to Decree No 697 of the Government of the Russian Federation dated 14 August 2013, the applicants applying for admission to the specialities and fields of study included in the list of those specialities and areas of study that require mandatory preliminary medical examinations in the manner prescribed for entering an employment or labour contract within the relevant position or speciality must undergo required preliminary medical examinations in the manner prescribed for entering an employment or labour contract within the appropriate position, profession or speciality. When applying for admission to respective specialities or fields of study, the applicants are entitled to provide a medical statement (medical professional advisory opinion).

- 2.34. Applicants who have submitted consciously forged documents to the Admissions Committee shall bear liability under the legislation of the Russian Federation.
- 2.35. Applicants shall have the right to file a written application for document revocation at any point of admission procedure. An applicant not included in the list of the enrolled persons shall have the right to file a standard application for the revocation of their document origins.

Before the deadline of the admission procedures to the government-funded places with particular admission conditions, St Petersburg University shall return applicants' documents upon their written application filed in person:

- 2.35.1. by the end of the current working day in case the revocation application is submitted at least two hours before the end of the working day;
- 2.35.2. in the course of the first two hours of the next working day in case the revocation application is submitted within two hours before the end of the working day.
- 2.36. The documents shall be collected upon the provision of an application receipt duly and directly issued to the applicant or their representative. If the revocation application indicates a document-collecting person, whose documents are being revoked, or their authorised representative, the set of application documents shall be given to this person.

Should the revocation application instruct to send submitted documents by general post service, St Petersburg University shall return only the originals of the submitted documents.

- 2.37. Applicants may cancel their participation in the competitive selection:
- 2.37.1. considering their admission application as a whole;
- 2.37.2. considering a particular and specified in the admission application degree programme with a specific competitive selection procedure or considering a group of degree programmes with competitive selection procedure taking into account types and conditions of study:
- 2.37.2.1. Before the end of the submission period by submitting a renewed admission application with the respective degree programme (a competitive group) omitted;

2.37.2.2. After the submission period deadline but before the enrolment, consents submission deadline upon submitting the consent for enrolment to a different degree programme.

The following persons shall be excluded from the competitive selection: those who have submitted a statement of their refusal to participate in the competitive selection or have withdrawn their documents after the document submission deadline set herein, as well as those who have received scores below the minimum score of points sufficient for the successful completion of the relevant admissions tests during the admissions tests or the additional admissions tests sessions.

- 2.38. It shall not be allowed to change the type of competition based on documents submitted to the Admissions Committee after the admission applications submission deadline.
- 2.39. Unless otherwise specified by the legislation of the Russian Federation and these Regulations, **6 pm, Moscow Standard Time** of the date defined by these Regulations shall be the deadline for accepting documents required both for participating in the competitive selection and for the enrolment.

3. Procedure of Defining the Number of Places for Admission to Degree Programmes of Higher Education

- 3.1. St Petersburg University complies with the admission quotas set for St Petersburg University by the respective federal executive authority while holding admissions of applicants to degree programmes of higher professional education to government-funded places.
- 3.2. Within the quoted government-funded places, St Petersburg University shall hold sponsored admission of the persons based on the contracts concluded between these persons and the following sponsors of the sponsored training:
- 3.2.1. federal government authorities, government authorities of constituent entities of the Russian Federation, local self-governance authorities;
 - 3.2.2. state and municipal institutions, unitary enterprises;
 - 3.2.3. government-owned corporations;
 - 3.2.4. government-owned companies;
- 3.2.5. organisations included in the consolidated register of organisations of the military-industrial complex, formed in accordance with Part 2, Article 21 of Federal Law No 488-FZ 'On Industrial Policy in the Russian Federation' dated 31 December 2014;
- 3.2.6. economic societies whose equity includes a share of the Russian Federation, a constituent entity of the Russian Federation or a municipal entity;
- 3.2.7. joint-stock companies whose stock has been owned or entrusted to a government-owned corporation;
- 3.2.8. subsidiary economic societies of the organisations specified in Clauses 3.2.4, 3.2.6, 3.2.7 of these Regulations;
- 3.2.9. organisations established by government-owned corporations or transferred to government-owned corporations in compliance with the provisions of federal laws in connection with the specified corporations;
- 3.2.10. organisations recognised as producers of agricultural goods in compliance with the provisions of Part 1 Article 3 Federal Law No 264-FZ 'On Development of Agriculture' dated 29 December 2006 in fields of study and specialities correspondent to agriculture and engineering.
- 3.3. Apart from admissions within admission quotas, St Petersburg University accepts students on a fee-paying (contractual) basis to places paid for by individuals and/or legal entities. The Admissions Committee shall define the number of such places.
- 3.4. The quantity of the places within each competitive selection for the admission to a separate degree programme of higher education as well as within unified competitive selection for the admission to a group of degree programmes regarding all types and conditions of study open for the admissions at St Petersburg University shall be specified by the Admissions Committee:
- 3.4.1. At least before 1 November 2021 concerning the number of places for admissions to bachelor's, specialist's or master's degree programmes;

- 3.4.2. At least before 1 June 2022 concerning the number of places for admissions within a particular admission quota, within a sponsored quota.
- 3.5. The tuition fee for the places on the fee-paying (contractual) basis paid for by individuals and/or legal entities and duly approved in the manner prescribed at St Petersburg University shall be announced by the Admissions Committee at least before 1 June 2022.
- 3.6. St Petersburg University may announce additional admissions to vacant places. Additional admissions and enrolments shall be carried out following the provisions of these Regulations. Thus, the information about additional admissions shall be published at least before 15 August 2022 in the manner prescribed by Clause 1.6 herein. Enrolments upon the additional admissions results must be finalised at least before the beginning of the academic year.
- 3.7. St Petersburg University complies with the provisions of the federal legislation and bylaws, including Section 12 of these Regulations.
- 3.8. Persons applying for admissions within admission quotas without taking admissions tests or upon the results of admissions competitive selection shall be enrolled to the vacant places within admission quotas or the sponsored quota.

4. Procedures for Admissions Tests

- 4.1. The list and forms of St Petersburg University admissions tests and admissions syllabi for degree programmes of higher education must be approved by the decision of the Admissions Committee. If it is allowed to select disciplines when taking admissions tests for admission to bachelor's or specialist's degree programmes, applicants can choose one or several disciplines.
- 4.2. The same list of admissions tests shall be specified for admission to all types (full-time or part-time) and conditions (government-funded places or places on the fee-paying (contractual) basis paid for by individuals and/or legal entities) of enrolment to each degree programme.
- 4.3. All admissions tests held at St Petersburg University to degree programmes of higher education shall:
 - 4.3.1. be marked on a 100-point scale;
- 4.3.2. be held either in Russian or in Russian and a foreign language, or in a foreign language (depending on the discipline of the admissions test and language of instruction of the implemented degree programme) in accordance with the information published in the manner prescribed by Clause 1.6. herein;
- 4.3.3. contain elements of testing the creative (professional) skills and/or knowledge of applicants in the written form and/or provide audio- or video recording of the admissions test;
 - 4.3.4. be held only with the use of distance teaching and learning technologies.
- 4.4. When conducting admissions to degree programmes of higher education, the Admissions Committee at least before 1 November 2021 shall announce the minimum score of points sufficient for successful completion of each admissions test for every separate degree educational programme or for a group of degree programmes within a unified competitive selection.
- 4.5. The schedule of all admissions tests for degree programmes of higher education (with information on the subject, date, time, place of examination, consultation (if any), results announcement day specified) shall be approved by the Admissions Committee and made available to the applicants at least before 1 June 2022. The schedule shall not indicate information of admissions tests board members. The results of admissions tests shall be announced within the time appointed by the admissions tests schedule.
- 4.6. Persons who have failed to attend admissions tests upon the justifiable reason (illness or other circumstances confirmed by appropriate documents) shall be allowed to undergo respective admissions tests during either the next admissions test period in case the admissions tests are held as the applicant groups are formed or before the admissions tests final deadline, or on the reserved for admissions tests day based on a written application with the appropriate confirming documents filed with the Admission Committee. Such an application with the documents must be submitted before the

deadline established by these Regulations. The applicant is allowed to undergo only one admissions test on one reserved for the admissions test day.

- 4.7. When admissions tests for the chosen type and/or condition of study to degree programmes of higher education are organised in several streams, the applicant is allowed to take each admission test only once, i.e. in one of the streams.
- 4.8. All admissions tests shall be held in compliance with the Regulations for Admissions Tests for Degree Programmes of Higher Education at St Petersburg University in 2022 available in Appendix 1 hereto.
- 4.9. St Petersburg University shall ensure the anonymity of all applicants' examination papers during their assessment in the manner their authorship cannot be identified.
- 4.10. Should they disagree with the obtained score of points for their admissions tests, or should they suspect these Regulations have been violated during admissions test sessions, the applicant shall have the right to file a properly written appeal (an appeal) with the St Petersburg University Appeals Panel in the manner prescribed by the Regulations for Review of Written Works, Appeal Submission and Appeal Proceedings that constitute Appendix 2 hereto.

5. Consideration of the Applicants' Individual Achievements

- 5.1. Applicants are entitled to submit any information concerning their individual achievements before the deadline established in Clauses 7.3 and 8.9 of these Regulations. Their results shall be considered during the admission procedure.
- 5.2. Individual achievements shall be taken into consideration and scored by points for individual achievements either to add them to the competitive selection score of points under Clause 6.5 hereto or to account for them while comparing personal achievements quantity in case other competitive selection criteria are equal.
- 5.3. Procedures for considering individual achievements of applicants applying for admission to bachelor's or specialist's degree programmes are set forth in Chapter 7 herein. The list and the procedures for evaluating the individual achievements of the applicants applying for admission to master's degree programmes are available in Appendix 3.

6. Enrolment Procedures

- 6.1. Enrolment to degree programmes of higher education to the respective type and/or conditions of study shall be carried out based on competitive selection results within admission quotas to government-funded places or places on the fee-paying (contractual) basis paid by individuals and/or legal entities (depending on the types and conditions specified in the admission application submitted) established for corresponding degree programme of higher education to the respective type and/or condition of study.
- 6.2. Persons to be enrolled are as follows: those who have taken part in the competitive selection for degree programmes of higher education in respective type and/or condition of the study, those who have provided their written consent for the enrolment, the original of the standard document (if applying for the admission to the places within admission quotas), and other required for the enrolment documents before the deadline established by these Regulations and successfully accomplished admissions tests and obtained the maximum score points both for their admissions tests and for their achievements, as well as the persons who are entitled to be enrolled without taking admissions tests as provided for by the legislation of the Russian Federation or who apply for admission to places on the fee-paying basis paid by individuals and/or legal entities, as well as those who provided signed contracts for rendering chargeable educational services.
- 6.3. Enrolment to degree programmes of higher education at St Petersburg University shall proceed under the order issued by the authorised official based on the decision of the Admissions Committee upon the completion of admissions for the respective degree programmes.

- 6.4. Upon the results of document consideration and/or admissions tests, the Admissions Committee shall form the ranked lists of applicants (hereinafter referred to as the lists of applicants) as follows:
- 6.4.1. the lists of applicants eligible for admission without any admissions tests for each competitive selection to places within admission quotas and to places on the fee-paid basis paid for by individuals and/or legal entities;
- 6.4.2. the lists of applicants who have successfully completed the admissions tests for each competitive selection.
- 6.5. The total number of competitive selection scores of points shall be calculated as the total number of points assigned for each admissions test plus the points assigned for individual achievements.
- 6.6. The lists of applicants shall contain either the Individual Insurance Number of Personal Ledger Account or the unique code assigned to the applicants (in case of the Individual Insurance Number of Personal Ledger Account is not applicable). Information for each applicant shall include the total score of points indicating the points assigned for every admissions test and the points assigned for individual achievements, as well as the information on the possessed particular right for enrolment, confirmation entitling not to take admissions tests (for those entitled), information of the provided consent for enrolment, as well as of the provided original of the standard credential. The application lists shall indicate neither the applicants' first names, middle names, nor family names.
- 6.7. The application lists of the applicants for enrolment shall be published in the manner prescribed by Clause 1.6 of these Regulations, including the UPSMS, and updated if necessary at least five (5) times until the day following the deadline for enrolment consent acceptance.
- 6.8. A person applying for admission to degree programmes of higher education to government-funded places shall be enrolled at St Petersburg University with a student status only to a particular degree programme and only to one type of enrolment, to which a competitive selection is held separately.
- 6.9. To be enrolled on degree programmes of higher education at St Petersburg University, at least before 6 pm Moscow Standard Time of the working day specified by these Regulations as the deadline for submission of the written enrolment consent, the applicant shall submit:
- 6.9.1. to places within admission quotas the written enrolment consent with the original of the standard credential;
- 6.9.2. to places on the fee-paying basis paid for by individuals and/or legal entities the written enrolment consent, the origin of the standard credential or its duly legalised copy, or its copy with the provision of the original. Should the admission application be submitted through UPSMS, the credential must be confirmed by the details available in the federal information system 'Federal Register of Information of Education or/and Qualification Credentials, Credentials of Training'.
- 6.10. Persons applying for admission to places on the fee-paying (contractual) basis paid for by individuals and/or legal entities shall be enrolled only upon signing the contract for rendering chargeable educational services based on the applicant's enrolment consent.
- 6.11. The application lists shall indicate the availability of the original standard credential and the enrolment consent at St Petersburg University that have been provided in the manner prescribed by these Regulations.
- 6.12. The applicants shall be enrolled in the following order: from the beginning to the end of the list of applicants until there are no vacant places for admission left.
- 6.13. The information that the applicants have provided their enrolment consents and the originals of the standard credentials, information about withdrawals of the originals of the standard credentials upon written applications from the applicants, as well as about their refusals to be enrolled shall be entered by the authorised personnel into St Petersburg Admission support information system for respective degree programmes of higher education.
- 6.14. Information on the applicants' enrolment shall be made known to the applicants by publishing the lists of the enrollees per each degree programme, admission to which has been open at St Petersburg University, that must indicate types and conditions of study in the manner envisaged by Clause 1.6. herein. The above-mentioned lists shall only contain either the Individual Insurance Number of Personal Ledger Account (if applicable) or the unique code assigned to the applicant (if the Individual

Insurance Number of Personal Ledger Account is not available), the total score of competitive selection points, the score of points for the completion of admissions tests and the score of points for the individual achievements. The applicants' first names, middle names, and family names shall not be indicated in the lists.

Section 2. The Particulars of Admissions Organisation to the Degree Programmes of Higher Education at Various Levels

7. The Particulars of Admissions Organisation to Bachelor's and Specialist's Degree Programmes

- 7.1. The following persons shall be allowed for admissions to study degree programmes at either bachelor or specialist level: those with credentials either of secondary (complete) general education, or of secondary vocational education and qualification, or of higher education and qualification. Credentials of primary vocational education confirming secondary (complete) general education or credentials of primary vocational education based on the secondary (complete) general education shall be equalled to the credentials of secondary vocational education and qualification. Persons with higher education and qualification credentials shall be admitted to study bachelor's or specialist's degree programmes only to the places on the fee-paying basis paid for by individuals and/or legal entities.
- 7.2. The admissions to St Petersburg University bachelor's and specialist's degree programmes shall be held in the manner as follows:
- 7.2.1. on the basis of the marked on a 100-point scale USE results obtained in general education disciplines correspondent to the degree programme to which the admission is held;
- 7.2.2. on the basis of the results of admissions tests held at the discretion of St Petersburg University in general education disciplines correspondent to the degree programmes. The categories of persons that are eligible to take these tests are as follows:
 - 7.2.2.1. physically challenged and disabled applicants, including disabled children;
 - 7.2.2.2. international applicants;
- 7.2.2.3. applicants with education credentials obtained from an international organisation (with general education disciplines the applicants have not undertaken USE in 2022)
- 7.2.2.4. applicants with higher education (for applicants applying for admission on the basis of secondary vocational education, the USE shall be considered admissions test);
- 7.2.3. upon the results of additional practical creative admissions tests and/or additional profession-specific admissions tests in a degree programme that requires applicants to possess either particular creativity, physical, and/or psychological characteristics;
- 7.2.4. upon the results of admissions test of professional education level for those who apply for degree programmes orientated towards the applicants with higher education.
- 7.3. The acceptance of personal applications for competitive selection and other documents from persons applying for admission to bachelor's and/or specialist's degree programmes shall start on **20 June 2022** and end on:
- 7.3.1. **7 July 2022,** for persons applying for admissions to degree programmes requiring completion of practical creative admissions tests and/or profession-specific additional admissions tests;
- 7.3.2. **10 July 2022,** for persons with higher education taking admissions tests held at the discretion of St Petersburg University;
- 7.3.3. **25 July 2022,** for persons applying for admission to enrolment without taking admissions tests held at the discretion of St Petersburg University, including those applying to admission without taking admissions tests;
- 7.3.4. **10 August 2022**, for persons applying for admissions to degree programmes to places both, paid for by individuals and/or legal entities and requiring only the USE results in general education disciplines, correspondent to the degree programme, to which admission is held; as well as for persons who have submitted their admission applications and documents within the period specified in Clauses 7.3.1 and 7.3.2 hereof, applying for the admission only to government-funded places and who have

successfully passed practical creative admissions tests and/or profession-specific additional admissions tests, as well as admissions tests held at the discretion of St Petersburg University (applications shall be accepted only to degree programmes for which the respective admissions tests have been passed successfully).

- 7.4. Persons applying for admission to bachelor's or specialist's degree programmes are eligible to submit applications for taking part in competitive selection in as few as three-degree programmes or specialities by submitting their application:
 - 7.4.1. in person
 - 7.4.2. through general post service operators;
- 7.4.3. electronically, through the applicants' Personal Account on the St Petersburg University official website or through the UPSMS.
- 7.5. Should the applicant for bachelor's or specialist's degree programme(s) choose to submit the specified documents electronically, the following documents must be scanned and attached to the registration form through the applicant's Personal Account:
- 7.5.1. a copy of an identity document with the indication of their citizenship (the following pages must be scanned: the applicant's photograph, family name, first name, middle name, gender, date and place of birth, citizenship, document series and number, information about the issuing authority and the applicant's place of registration);
- 7.5.2. an electronic image of the application for the relevant degree programme containing the applicant's handwriting signature;
- 7.5.3. the applicant's photograph (taken in the current calendar year, with a vertical resolution of at least 800 pixels and horizontal resolution of at least 600 pixels);
 - 7.5.4. the electronic image of their standard education credential;
- 7.5.5. the electronic image of their Individual Insurance Number of Personal Ledger Account (if applicable);
- 7.5.6. electronic images of the documents confirming the applicant's particular rights and/or privileges when applying for admission (if applicable);
- 7.5.7. electronic images of the documents confirming the applicant's individual achievements (if applicable)
 - 7.5.8. other documents required by these Regulations.
- 7.6. An electronic image of a document in an interplatform open format for electronic documents PDF (PortabelDocumentFiles) must ensure the visual identity with its original on a 1:1 scale. The quality of the provided electronic images must ensure the legibility of the entire document and the identification of their details. If the hard copy of any document consists of two or more pages, its electronic image must be provided in one file. It is required to make scans of documents with a resolution of at least 300 dpi. It is not allowed to provide illegible scanned images of the documents as well as images that miss significant sections of the document (texts, signatures, seals or stamps, etc.).
- 7.7. The applicant applying for admission to bachelor's or specialist's degree programmes at St Petersburg University must provide either the original or a copy of one of the documents recognised as their education credential as specified in Clause 7.1. herein taking into account provisions of Clause 4 of these Regulations.
- 7.8. Should several actual USE resulting scores of points or resulting scores of the USE points plus general education admissions test points held at the discretion of St Petersburg University be available, the highest resulting score of points shall be recognised for the admissions test result.
- 7.9. The USE resulting scores of points recognised as the admissions tests results in general education disciplines correspondent to a degree programme open for admission, as well as resulting score of points for the admissions tests held at the discretion of St Petersburg University must not be less than the minimum score of points that has been established sufficient for the successful passing admissions tests in corresponding disciplines by the Admissions Committee.
- 7.10. The Admissions Committee shall appoint at least three (3) admissions tests from the List of the admissions tests for pursuing higher education degree in bachelor's or specialist's degree programmes approved by the Ministry of Science and Higher Education of the Russian Federation

(hereinafter referred to as the List of admissions tests), including a mandatory admissions test in Russian and an admissions test in a general education discipline specified as the mandatory one in the List of admissions tests for the respective degree programme (speciality).

- 7.11. Should there additional practical creative and/or profession-specific admissions tests be held in respective degree programmes open for admission, the Admissions Committee shall appoint at least two (2) admissions tests from the List of the admissions tests.
- 7.12. Admissions test held at the discretion of St Petersburg University shall start **after 8 July 2022** and end on **25 July 2022**. Additional admissions tests shall be carried in several stages as the admissions test groups of applicants, who have submitted required documents, are formed. The applicants shall submit applications for taking admissions tests on the reserved for admissions tests the day before 12 am (Moscow Standard Time) on 21 July 2022.
- 7.13. An applicant shall take the general education admissions test held at the discretion of St Petersburg University only once regardless of the degree programme, the type (full-time, evening part-time) or the condition (government-funded places or places on the fee-paying basis paid for by individuals and/or legal entities) of study they apply for. The resulting score of points obtained shall be considered for admissions to all degree programmes and types and conditions of study to which the applicant has submitted their application for competitive selection.
- 7.14. An applicant shall take the additional practical creative or profession-specific admissions test specified for the respective degree programme and held at the discretion of St Petersburg University only once regardless of the degree programme, the type (full-time, evening part-time) or the condition (government-funded places or places on the fee-paying basis paid for by individuals and/or legal entities) of study they apply for. The resulting score of points obtained for practical creative or profession-specific admissions tests in different disciplines when applying for a particular degree programme shall not be recognised as a resulting score of points should the applicant apply for a different degree programme.
- 7.15. When admitting persons to bachelor's or specialist's programmes, St Petersburg University shall assign points for the individual achievements as follows:
- 7.15.1. for the status of a winner and awardee of Olympic Games, Paralympic Games and Deaflympic Games, a world champion, a European champion, the first-place finisher in a world or European championship in sports admitted to the Olympic Games, Paralympic Games and Deaflympic Games -10 points;
- 7.15.2. for the status of a world champion, a European champion, the first-place finisher in a world or European championship in sports not admitted to the Olympic Games, Paralympic Games, or Deaflympics **5 points**;
- 7.15.3. for a gold, silver, or bronze insignia of the All-Russia Physical Culture and Sports Complex 'Ready for Labour and Defence' (GTO) (hereinafter respectively referred to as GTO badge, GTO complex) received by the applicant in accordance with Order No 16 'On the Procedure for Awarding Persons Who Fulfilled the Test Standards of the All-Russia Physical Culture and Sports Complex 'Ready for Labour and Defence' (GTO) the Corresponding Insignia of the All-Russia Physical Culture and Sports Complex 'Ready for Labour and Defence' (GTO)' approved by the Ministry of Sports, Tourism and Youth Policy of the Russian Federation dated 14 January 2016 if the applicant has been awarded a GTO badge for the fulfilment of the GTO complex, designed for the age group of the population of the Russian Federation to which the applicant belongs(ed) in 2021 and/or 2022 provided that the GTO badge has been awarded with the accompanied GTO badge certificate or the Extracts from the order on awarding the GTO badge of the Ministry of Sports, Tourism and Youth Policy, confirmed by the signature of an authorised officer of the public administration body of the Russian Federation constituting entity 2 points;
 - 7.15.4. other sports achievements:
- 7.15.4.1. membership in a Russian national Olympic team in 2021 or 2022 (the core, under-21, junior and youth teams as well as the reserve squad) -3 points;
- 7.15.4.2. membership in a Russian national non-Olympic team in 2021 and 2022 (the core, under-21, junior and youth teams as well as the reserve squad) -1 point;

- 7.15.4.3. for the insignia 'Unarmeyskoye Valour', a certificate to the badge 'Unarmeyskoye Valour' issued in accordance with these Regulations for the procedure for registering, issuing, and awarding the badges of 'Unarmeyskoye Valour' of three degrees of VVPOD 'UNARMIA' approved by the decision of the General Staff of the VVPOD 'UNARMIA' dated 9 February 2018, for the first, second or third place in the All-Russia and international stages of military sports games, sports Olympiads 1 point;
 - 7.15.4.4. for the status of a winner of the All-Russia military and sports game 'Pobeda' -1 point;
- 7.15.5. for the education credentials or education plus qualification credentials with distinction (a certificate of secondary general education with distinction, a certificate of secondary (complete) general education with distinction, a certificate of secondary (complete) general education for the gold (silver) medal awardees. A diploma of secondary vocational education with distinction, a diploma of primary vocational education for gold (silver) medal awardees) obtained from the educational institutions of the Russian Federation **3 points**;
- 7.15.6. for the awards from school olympiads (that grant no particular rights and/or particular privilege when applying for admission to relevant types and conditions of study) and other intellectual and/or creative contests, or sports events held in compliance with Part 2 Article 77 of Federal Law No 273-FZ aimed to identifying and supporting persons presented outstanding abilities:
- 7.15.6.1. a diploma of the winner of the regional (3rd) round of the All-Russian School Olympiad in a discipline corresponding to the first or second discipline in the list of admissions tests specified for admission to the relevant bachelor's or specialist's degree programmes (see Appendix 4 hereto to find out which disciplines of the All-Russian School Olympiad correspond to which disciplines of the admissions tests) during four subsequent years after the respective Olympiad was held, **5 points**;
- 7.15.6.2. a diploma of an awardee of the regional (3rd) round of the All-Russian School Olympiad in a discipline corresponding to the first or second discipline in the list of admissions tests specified for admission to the relevant bachelor's or specialist's degree programmes during four subsequent years after the respective Olympiad was held **4 points**;
- 7.15.6.3. a diploma of a winner of a school olympiad included in the list of the Ministry of Education during four subsequent years after the respective Olympiad was held -5 points;
- 7.15.6.4. a diploma of an awardee of a school olympiad included in the list of the Ministry of Education during four subsequent years after the respective Olympiad was held -4 points;
- 7.15.6.5. a diploma of a winner of the St Petersburg University School Olympiad in the set of disciplines 'A Modern Manager' (mathematics, social studies, English) during four subsequent years after the respective Olympiad was held **5 points**;
- 7.15.6.6. a diploma of an awardee of the St Petersburg University School Olympiad in the set of disciplines 'A Modern Manager' (mathematics, social studies, English) during four subsequent years after the respective Olympiad was held **3 points**;
 - 7.15.6.7. a diploma of a winner of the All-Russia School Contest 'Bolshaya Peremena' 5 points;
- 7.15.6.8. a diploma of a winner or an awardee of the National Technological Initiative 'Kruzhkovoie Dvizheniie' Olympiad (not considered for individual achievements points in compliance with Clause 7.15.6.3. or 7.15.6.4. herein) during four subsequent years after the respective Olympiad was held **2 points**;
- 7.15.7. a diploma of a world champion in professional skills among persons with a disability or additional support needs 'Abilympics' -10 points;
- 7.16. For persons having several individual achievements out of those set forth in Clauses 7.15.4 and 7.15.6., points shall be assigned for one personal achievement which is the basis for assigning more points. The total score of points assigned for individual achievements shall not exceed 10 points. The applicant shall give any information on their achievements (specifying the achievement type and confirmation documents details) in their personal application.
- 7.17. The admissions to St Petersburg University bachelor's and specialist's degree programmes within admission quotas shall be held in the following stages:
- 7.17.1. the stage of the priority enrolment (without admissions tests to places within a particular admission quota, to places within the sponsored admissions quota);

- 7.17.2. the stage of regular enrolment (enrolment upon regular admissions tests).
- 7.17. The list of applicants eligible for admission without any admissions tests shall specify reasons for applicants' admission without any admissions tests, as well as the number of points assigned for individual achievements and the priority right to enrolment (if any). The lists of applicants eligible for admission without any admissions tests shall be ranked as follows:
- 7.18.1 in accordance with the status of persons entitled to admission without taking any admissions tests in the following order:
 - 7.18.1.1. members of national teams taking part in international olympiads;
 - 7.18.1.2. winners of the final round of the All-Russian School Olympiad;
 - 7.18.1.3. awardees of the final round of the All-Russian School Olympiad;
 - 7.18.1.4. winners and awardees in sports specified in Clause 7.15.1. Of these Regulations
- 7.18.1.5. winner of school olympiads included in the list of the Ministry of Education of the Russian Federation;
- 7.18.1.6. awardees of school olympiads included in the list of the Ministry of Education of the Russian Federation;
- 7.18.2. for persons specified in each Sun-clauses of Clause 7.18.1. 7.18.1. of these Regulations in the descending order of the points assigned for individual achievements;
- 7.18.3. upon the equality of the criteria specified in Clauses 7.18.1. and 7.18.2. of these Regulations the higher place shall be given to the applicants with the priority enrolment right according to Clauses 2.8.1 2.8.13 herein;
- 7.18.4. Upon the equality of the criteria specified in Clauses 7.18.1. and 7.18.3. of these Regulations, the higher place shall be given to the applicants with the priority enrolment right according to Clauses 2.8.14. herein.
- 7.18.5. Upon the equality of the criteria specified in Clauses 7.18.1. 7.18.4. of these Regulations the higher place shall be given to the applicants with the following individual achievements to be considered should applicants be equal according to other ranking criteria:
 - 7.18.5.1. laureates of the Intel International Science and Engineering Fair (Intel ISEF);
- 7.18.5.2. persons with certificates confirming their accomplishment of the St Petersburg University open online courses;
- 7.18.5.3. persons with certificates confirming their accomplishment of the St Petersburg University non-degree programmes;
 - 7.18.5.4. winners of the All-Russia Dictation of Victory at either federal or regional levels;
- 7.18.5.5. winners of the All-Russia contest among general education educational organisations 'History of Self-Governance in my Region' at either federal or regional levels;
- 7.18.5.6. persons with the certificate confirming their successful accomplishment of 'IT SCHOOL SAMSUNG' programme;
 - 7.18.5.7. winners of the school accelerator SberZ;
 - 7.18.5.8. winners or awardees of the final stage of 'Russia in Digital World' School Olympiad;
 - 7.18.5.9. winners or awardees of the Baltic Science and Engineering Fair;
- 7.18.5.10. winners or awardees of the intellectual talk-show 'Legko Skazat' held by St Petersburg University in the framework of the Federal Target Programme 'The Russian Language';
- 7.18.5.11. winners and awardees of the competence track of the National Technological Initiative Kruzhkovoie Dvizheniie Olympiad 'Talant 20.35';
- 7.18.5.12. winners and awardees of the competence track of the National Technological Initiative Kruzhkovoie Dvizheniie Olympiad 'Talant NTO';
- 7.18.5.13. winners and awardees of the St Petersburg University School Olympiad in technoentrepreneurship;
- 7.18.5.14. winners and awardees of the School Olympiad in the History of Engineering held by St Petersburg University and the Central Rail Road Transport Museum of the Russian Federation;
- 7.18.5.15. winners and awardees of the International Contest Literary Ties of Belarus and Russia: from a Dialogue to Understanding' organised by St Petersburg University supported by the Federal

Agency for the Commonwealth of Independent States Affairs, compatriots residing abroad and within the framework of international humanitarian cooperation in the Republic of Belarus.

- 7.19. The applicants who have successfully completed the admissions tests shall be ranked on the competition lists in order:
 - 7.19.1. of decrease of the total score of competition points;
- 7.19.2. if the total score of competition points is equal of decrease of the score of competition points assigned upon the results of the admissions tests;
- 7.19.3. if the total number of competition points assigned following the results of the admissions tests is equal of decrease of the score of points assigned following the results of particular admissions tests, in accordance with the priority ranking specified by the Admissions Committee for the admissions tests. The priority ranking shall be made available when publishing the list of degree programmes open for admission at St Petersburg University in the manner prescribed by Clause 1.6. herein;
- 7.19.4. Should the applicants be equal under the criteria specified in Clauses 7.19.1. and 7.19.3. of these Regulations, the higher place shall be given to the applicants with the priority enrolment right according to Clauses 2.8.1 2.8.13 herein;
- 7.19.5. Should the applicants be equal under the criteria specified in Clauses 7.19.1. 7.19.4. of these Regulations, the higher place shall be given to the applicants with the priority enrolment right according to Clause 2.8.14. herein;
- 7.19.6. Should the applicants be equal under the criteria specified in Clauses 7.19.1. 7.19.5. of these Regulations, the higher place shall be given to those applicants specified in Clause 7.18.5. herein with individual achievements to be considered upon the equality of the applicants according to other ranking criteria:
- 7.20. The enrolment to St Petersburg University bachelor's or specialist's degree programmes within admission quotas (to regular government-funded places) shall be held within the following deadlines:
- 7.20.1. **27 July 2022** as the established date for publishing the lists of applicants in accordance with the procedure set forth by Clause 1.6 of these Regulations;
 - 7.20.2. The priority enrolment stage:
- 7.20.2.1. **28 July 2022** as the established date for both accepting enrolment consent applications for enrolment from the persons applying for either admission without admissions tests, or within the admissions quotas, or within sponsored quotas and submitting originals of their standard credentials;
- 7.20.2.2. **30 July 2022** as the established date for issuing and publishing orders for the enrolment of those persons who are eligible for admission without any admissions tests, either to the places within admission quotas or to the places within the sponsored quota who have submitted their enrolment consent applications and the originals of their standard documents;
 - 7.20.3. **The main stage of enrolment** to degree programmes regular competition places:
- 7.20.3.1. **3 August 2022** as the established date for accepting enrolment consent applications from those who apply for regular competition places and who have submitted originals of their standard credentials;
- 7.20.3.2. **9 August 2022** as the established date for issuing and publishing orders on the enrolment of those applicants who have been included in the lists of the persons applying for admission to regular competition places and who have submitted the originals of their standard credentials.
- 7.21. Applicants shall be **enrolled** with St Petersburg University student status of bachelor's or specialist's degree programmes **to places paid for** by legal entities and/or individuals within the following deadlines:
- 7.21.1. 11 August 2022 as the established date for announcing and publishing the name lists of those persons who have successfully completed the admissions tests and been sorted by the score of points received in descending order accounting for the specified by the Admissions Committee availability of the fee-paid (contractual) places, and who have submitted their enrolment consent applications to particular bachelor's or specialist's degree programme to the places paid for by individuals and/or legal entities in the manner set forth in Clause 1.6 of these Regulations;

- 7.21.2. **15 August 2022** as the established date for submitting the enrolment consent applications by those persons who have signed contracts for rendering chargeable educational services;
- 7.21.3. **16 August 2022** as the established date for issuing and publishing the orders on enrolment since 1 September 2022 of those persons who have successfully completed the admissions tests and submitted their enrolment consent applications and concluded the contracts for rendering chargeable educational services, and who have been included in the list specified in Clause 7.21.1. herein.
- 7.22. Should the admission be organised for a totality of the degree programmes within one field of study or an enlarged group of the field of study, the applicants shall take part in the competitive selection for the admission to study each bachelor's degree programme that comprises a particularly competitive group.
- 7.23. In the case specified in Clause 7.22 of these Regulations, the procedures of both launching the certain degree programme and distributing the applicants (enrolling to) within the certain degree programme shall be governed by the Procedures for distributing within (enrolment to) the bachelor's degree programmes admission to which is organised according to the totality of the degree programmes in the framework of the field of study (a single competitive selection for a group of degree programmes) (Appendix 5 hereto).
- 7.24. When applying for admission to the government-funded places, the applicant shall be allowed to submit only two (2) enrolment consent applications. Should there the enrolment consent application be submitted after the previous enrolment consent application for admission to the government-funded places was submitted, the applicant before filing their next enrolment consent application shall submit their enrolment refusal application to cancel their previous enrolment consent application. The applicant's enrolment refusal application shall serve as the grounds for their withdrawal from the number of enrollees.

8. The Particulars of Admissions Organisation to Master's Degree Programmes

- 8.1. Persons with higher education confirmed by a diploma of their higher education and qualification shall be admitted to pursue master's degree programmes. Persons with either a master's or a specialist's diploma that confirms their qualification of a specialist shall be admitted to pursue master's degree programmes only on places on the fee-paying basis paid for by individuals and/or legal entities.
- 8.2. Applicants for admission to master's degree programmes shall also have an opportunity to submit their applications and other required documents either in person or via general post service operator, or electronically through the Personal Account on the St Petersburg University official website, or through the UPSMS. The deadlines for acceptance of applications and other documents for competitive selection, the deadlines for admissions tests and enrolment differ depending on the type of enrolment to a master's degree programme and on the forms of admissions tests.
- 8.3. Applicants to St Petersburg University degree programmes of higher education are eligible to apply for admission to participate in the competitive selection for an unlimited quantity of master's degree programmes, open for separate admissions at St Petersburg University.
- 8.4. When submitting their admission applications, the applicants for master's degree programmes shall provide the original or a copy (digital image) of their standard higher education credential as their standard credential according to the provision of Clause 4 of these Regulations. Should an applicant lack the above-mentioned documents at the point of application or at the point of the deadline specified in Clause 8.9. herein due to the circumstances beyond the applicant's control, the applicant shall be allowed to provide a standard transcript certified by the seal of the educational institution of higher education that must carry information of the completed curriculum and successfully passing the State Final Assessment in a degree programme of higher education specifying the disciplines mastered, assessment grades awarded, and the date of issuance of the standard education credential.
- 8.5. When applying for admission to the competitive selection, the applicants shall be allowed to submit their documents specified in Clause 8.7 herein electronically through their Personal Account through the St Petersburg University official website. To do so, the applicants shall undertake to register on the St Petersburg University official website and receive their personal login and password, ensuring

their access to the Personal Account. Logins and passwords must not be disclosed to third parties. The responsibility for their Personal Account login and password security after they have obtained them shall rest with the applicant.

- 8.6. Applications for admission to master's degree programmes shall be filled in electronically. To submit their application form, the applicant must type information in all boxes required, print the application out, and confirm the completeness and accuracy of the provided information by putting their handwritten signature. Applicants shall submit their signed admission application with the required documents to the Admissions Committee either in person, or through the general post service operator, or electronically through the Personal Account on the St Petersburg University website in accordance with the provisions set forth in Clause 8.7. herein. Admission applications and other documents submitted in the manner not specified herein, including those submitted via email or fax, shall not be accepted and considered by the Admissions Committee.
- 8.7. Should the applicant for a master's degree programme choose to submit the specified documents through their Personal Account on the St Petersburg University official website, the following documents must be scanned and attached to the registration form:
- 8.7.1. the electronic image of an identity document with the indication of their citizenship (the following pages must be scanned: the applicant's photograph, family name, first name, middle name, gender, date and place of birth, citizenship, document series and number, information about the issuing authority and the applicant's place of registration);
 - 8.7.2. the electronic image of the admission application for respective degree programmes;
 - 8.7.3. the electronic image of their standard education credential with a supplement;
- 8.7.4. the electronic image of their Individual Insurance Number of Personal Ledger Account (if applicable);
- 8.7.5. the applicant's photograph (taken in the current calendar year, with a vertical resolution of at least 800 pixels and horizontal resolution of at least 600 pixels);
- 8.7.6. the electronic image of the documents for the competitive selection of documents (portfolio) in compliance with the admissions test syllabus;
- 8.7.7. the electronic images of the documents confirming the applicant's individual achievements (if applicable);
 - 8.7.8. other required documents.
- 8.8. An electronic image of a document in the interplatform open format for electronic documents PDF (PortabelDocumentFiles) must ensure the visual identity with its original on a 1:1 scale. The quality of the provided electronic images must ensure the legibility of the entire document and the identification of their details. If the hard copy of any document consists of two or more pages, its electronic image must be provided in one file. It is required to make scans of documents with a resolution of at least 300 dpi. It is not allowed to submit illegible scanned images of the documents as well as images that miss significant sections of the document (texts, signatures, seals or stamps, etc.).
 - 8.9. The admission applications shall be accepted on:
- 8.9.1. **1 March 2022** if submitted electronically, through the applicants' Personal Account on the St Petersburg University official site;
 - 8.9.2. 1 July 2022 if submitted in person or through the general post service operator.
- 8.10. If submitted electronically through either the Personal Account on the St Petersburg University official website, or the UPSMS, or general post service operator, the deadline for admission applications and other documents for the competitive selection shall be on:
- 8.10.1. **20 July 2022**, **at 6 pm (Moscow Standard Time)** for full-time master's degree programmes;
- 8.10.2. **9 September 2022**, at 6 pm (Moscow Standard Time) for part-time master's degree programmes.
 - 8.11. For full-time master's degree programmes:
- 8.11.1. the admissions tests shall be conducted from **22 July 2022 to 6 August 2022**. Should there be any coincidence of the admissions tests dates for different programmes at St Petersburg University, applications for passing the admissions tests on a reserved for tests day shall be accepted on

- 30 July 2022 until noon, Moscow Standard Time. Should there be any other justifiable reasons (illness, etc.) on 2 August 2022 until noon, Moscow Standard Time.
 - 8.11.2. applicants shall be enrolled as students within the following time frame:
- 8.11.2.1. **8 August 2022:** with publishing in the manner prescribed by Clause 1.6 of these Regulations the complete name list of applicants sorted in descending order by the score of points they have received (to be specified) calculated as the sum of the points for each admissions test and the points for individual achievements, the named list of those persons who have successfully completed the admissions tests and whose enrolment the Admissions Committee may consider under various conditions of the competitive selection (regular competition, sponsored quotas, etc.) and conditions of study:
 - 8.11.2.2. **15 August 2022 until 6 pm, Moscow standard time**:
- 8.11.2.2.1. the deadline for accepting the enrolment consent applications and submitting the original standard education credentials for the applicants for the government-funded places;
- 8.11.2.2.2. the deadline for accepting the enrolment consent applications, submitting standard documents (the original of the document or its duly certified copy with the provision of the original), and concluding the contract for rendering chargeable educational services for the applicants for the places on the fee-paying basis paid for by individuals and/or legal entities;
- 8.11.2.3. **17 August 2022** as the established date for issuing and publishing the orders on enrolment of those persons who have provided documents specified in Clause 8.11.2.2. of these Regulations.
- 8.12. Applicants to full-time master's degree programmes that have successfully completed all the admissions tests but lost the competition for government-funded places shall have the right to submit, together with the original or a copy of their standard education credential and the enrolment consent application, an application to participate in the competition for places on the fee-paying (contractual) basis on **15 August 2022** until 6 pm Moscow Standard Time. They shall also have the opportunity to conclude the contract for rendering chargeable educational services.
 - 8.13. For **part-time** master's degree programmes:
 - 8.13.1. admissions tests shall be conducted from 10 September 2022 to 15 September 2022;
 - 8.13.2. The following are the deadlines for enrolment:
- 8.13.2.1. **16 September 2022** as the established date for publishing in the manner prescribed by Clause 1.6. of these Regulations the completer ranked name list of the applicants sorted in the descending order of the obtained competitive score points (to be specified) calculated as the sum of the points for each admissions test and the points for individual achievement of those persons who have successfully passed admissions tests and whose enrolment the Admissions Committee may consider upon different admissions tests results and study conditions;
- 8.13.2.2. **21 September 2022 at 6 pm** (Moscow Standard Time) as the established date for submitting enrolment consent applications and standards credentials (originals of the document or its duly certified copy, or its copy with the provision of its original) and concluding the contract for rendering chargeable educational services;
- 8.13.2.3. **22 September 2022** as the established date for issuing and publishing the orders on enrolment of those persons who have provided documents specified in Clause 8.13.2.2. of these Regulations.
- 8.14. Based on the applicant's personal request, the Admissions Committee shall make a decision to recognise the results of the respective international certification examinations provided by the applicant together with the admission application and other required documents as the results of the admissions test of foreign language skills and the results of the admissions test aimed at assessing applicants' foreign language competencies and professional competence required for successful completion of the selected master's degree programme. The applicants must submit the said documents at least before the deadline for submitting the documents specified in Clause 8.10 of these Regulations. The said decision shall be made known to the applicants at least before the day the admissions tests begin in the manner prescribed by Clause 1.6 herein.

- 8.15. In the competitive list, the applicants who have successfully passed the admissions tests shall be ranked as follows:
- 8.15.1. in the descending order of the competitive points calculated as the sum of the score points for each admissions test and individual achievements
 - 8.15.2. in the descending order of the scored points assigned for the admissions tests;
- 8.15.3. in the descending order of the scored points assigned for separate admissions tests in accordance with the priority of the admissions tests established by the Admissions Committee;
- 8.15.4. based on individual achievements if other criteria considered and specified in Clauses 8.15.1. 8.15.3. of these Regulations are equal:
- 8.15.4.1. based on the average score of the standard education credential provided by the applicant when applying for the competitive selection. The calculation of the average score shall be carried out based on the supplement to the standard education credential. If the supplement to the standard education credential is missing, the ranking score average for a particular applicant shall be zero (0).
- 8.15.4.2. based on the score of points assigned for the motivation letter and the essay in accordance with the syllabus of the admissions test in the form of competitive selection of documents (portfolio);
- 8.15.4.3. based on the score of points assigned for the section 'Information about Publications' in accordance with the syllabus of the admissions test in the form of competitive selection of documents (portfolio);
- 8.15.4.4. based on the score of points assigned for the section 'Documents and Materials Confirming the Applicant's Participation in Scientific Events' in accordance with the syllabus of the admissions test in the form of competitive selection of documents (portfolio);
- 8.15.4.5. based on the number of publications on the topics correspondent to the enlarged group of fields of study (specialities) within which the admission to a degree programme is held in the journals indexed in the Web of Science Core Collection and Scopus scientometric databases information of which has been provided by the applicant;
- 8.15.4.6. based on the number of publications on the topics correspondent to the enlarged group of fields of study (specialities) within which the admission to a degree programme is held in the journals indexed in the Russian Science Citation Index scientometric databases upon the information provided by the applicant;
- 8.15.4.7. based on the number of scientific events on the topic correspondent to the enlarged group of fields of study and specialities within which the admission to a degree programme is held upon the information provided by the applicant;
- 8.15.4.8. based on the status of a diploma holder of the St Petersburg University 'Petropolian Science (Re)Search' Open International Olympiad for students and young scientists or the 'I am a Professional' All-Russia University Olympiad;
- 8.15.4.9. based on the average score of points obtained for the final state assessment according to the information contained in the standard education credential.
- 8.16. The applicants must submit their documents to have them ranked before the deadline stipulated in Clause 8.9. of these Regulations. The Admissions Committee shall have the authority to introduce other additional ranking criteria. The information on the additional ranking criteria shall be made available in the manner prescribed in Clause 1.6. herein.
- 8.17. Should the admission be organised for a totality of master's degree programmes within one field of study, the applicants shall take part in the competitive selection for the admission to study each of the master's degree programmes that comprise a particularly competitive group.
- 8.18. In the case specified in Clause 8.17. of these Regulations, the procedures of both launching the certain degree programme and distributing within (enrolling to) a particular degree programme shall be specified by the Procedures for distribution within (enrolment to) the Master's degree programmes admission to which is organised according to the totality of the degree programmes in the framework of the field of study (a single competitive selection for a group of degree programmes) (Appendix 6 hereto).

9. The Particulars of Admissions Organisation to Bachelor's or Specialist's Degree Programmes for Persons with Higher Education

- 9.1. Persons with higher education that is confirmed by a degree diploma of higher education and qualification and applying for admission to bachelor's or specialist's degree programmes shall be enrolled at St Petersburg University to the places on the fee-paying (contractual) basis paid for by individuals and/or legal entities.
- 9.2. The procedures of the admission of applicants with higher education to bachelor's or specialist's degree programmes, acceptance of documents for the admission to study, organisation of admissions tests, formation of competitive lists, and enrolment to study shall be governed by the provisions of Chapter 7 of these Regulations observing the same deadlines and the same procedures applied to the admissions of those persons who apply for the admission to bachelor's or specialist's degree programmes to the places on the fee-paying basis paid for by individuals and/or legal entities and pursue degree programmes of the appropriate level for the first time.
- 9.3. The applicants with higher education applying for the admission must take admissions tests correspondent to the professional education as follows:
- 9.3.1. persons applying for admission to degree programmes orientated towards applicants with higher education must take admissions test in the form of competitive selection of the documents (portfolio);
- 9.3.2. persons applying for admission to other bachelor's or specialist's degree programmes must take admissions tests in those disciplines in which the general education admissions test are usually held, as well as admissions tests of creative and/or profession-specific character in those fields of study (specialities) that require specific creative abilities, physical and/or psychological qualities in the applicants.

Section 3. Particulars of Admissions Organisation for Foreign Nationals

10. The foreign nationals categories admitted to the competitive selection, particulars of their submitting admission application and other required documents, their participation in admissions tests and enrolment

- 10.1. This section hereof governs the particulars of the admissions organisation for foreign nationals, stateless citizens, including compatriots residing abroad (hereinafter referred to as foreign citizens, persons, applicants) and for those who have equal to citizens of the Russian Federation rights when applying for admission to degree programmes at St Petersburg University both to the government-funded places and to the contractual places under contracts for rendering chargeable educational services.
- 10.2. Admission of foreign citizens to degree programmes on the government-funding basis shall be carried out:
- 10.2.1. to the places within the quotas established by Decree No 2150 'On Establishing an Education Quota for Foreign Nationals and Stateless Citizens in the Russian Federation' of the Government of the Russian Federation dated 18 December 2020 based on the references issued by the Ministry of Science and Higher Education of the Russian Federation;
 - 10.2.2. in compliance with international treaties of the Russian Federation;
- 10.2.3. in compliance with federal laws, including Federal Law No 99-FZ 'On State Policy of the Russian Federation Concerning Compatriots Residing Abroad' dated 24 May 1999.
- 10.3. Admissions of foreign nationals at St Petersburg University specified in Clauses 10.2.2. and 10.2.3. of the Regulations for Admission to St Petersburg State University to government-funded places shall be carried out on a competitive basis on the equal with citizens of the Russian Federation conditions defined by these Regulations for Admission.
- 10.4. Terms and procedures for the admissions of foreign nationals specified in Clause. 10.2.1 herein, shall be set forth in the Regulations for an Open Competition among Foreign Nationals, Stateless

Citizens, and Compatriots Residing Abroad Applying for Admission to St Petersburg State University to Government-Funded Places approved by the Admissions Committee and published in the manner prescribed by Clause 1.6 herein.

- 10.5. Admissions of foreign nationals to the contractual places under contracts for rendering chargeable educational services shall be carried out on a competitive basis to the places within the quantity and on the conditions defined by the Admissions Committee as provided for by these Regulations for Admission.
- 10.6. Admissions of foreign nationals to the St Petersburg University degree programmes under the St Petersburg University agreements of interuniversity cooperation, including agreements for double-degree programmes, shall be carried out in compliance with the provisions of Clause 10.4 of these Regulations for Admission. If these agreements provide for admissions to the places on the feepaying basis paid for by individuals and/or legal entities, the admission shall be carried following the procedures set forth for persons specified in Clause 10.5. herein.
- 10.7. The list of degree programmes, the list of admissions tests with the minimum score of points required for their successful accomplishment, as well as the admissions tests syllabi shall be made available to the applicants in the manner prescribed by Clause 1.6. herein.
- 10.8. Enrolment to St Petersburg University of the foreign nationals specified in Clause 10.2.1. of these Regulations shall be carried out based on the references issued by the Ministry of Science and Higher Education for the admission to degree programmes correspondent to the fields of study (specialities) specified in the reference issued by the Ministry of Science and Higher Education.
- 10.9. When submitting their admission applications to St Petersburg University, the foreign nationals, as set forth in Section 2 of the Regulations for Admission, shall provide documents specified in Clause 2.25 herein, documents confirming that they relate to the categories of foreign nationals specified in Clauses 10.2 and 10.5 herein, as well as the following documents:
- 10.9.1. a copy of the visa to enter the Russian Federation if the foreign national has arrived in the Russian Federation on an entry visa indicating the purpose of their stay as 'study at St Petersburg University';
 - 10.9.2. a copy of their Landing Card;
 - 10.9.3. a confirmation that they have read the Guide of the Rules of Stay in the Russian Federation;
- 10.10. All translations into Russian must be made in the name of the entry visa holder (if the entry visa is not available, in the name specified in the identity document).
- 10.11. To participate in the competitive selection procedure, the applicants shall submit their applications specified in Clause 10.9 herein electronically through the St Petersburg University official website. To do so, the applicants must undertake to register on the St Petersburg University official website and receive their personal login and password to access their Personal Account. The applicants must not disclose their login and password to third parties. From the moment they have obtained their login and password, the applicant shall bear responsibility for their security.
- 10.12. The applicant shall complete the admission application electronically. To submit their admission application, the applicant must fill in all boxes required, print the completed application out, and confirm the completeness and accuracy of the data provided by putting their handwritten signature. The applicant shall submit their signed admission application with the required documents to the Admissions Committee either in person, or through the general post service operator, or electronically in the manner prescribed by Clause 10.13 herein.
- 10.13. Should they choose to submit their admission application with the required documents electronically, the applicant shall provide documents as scanned copied and attach the following copies to the registration form in their Personal Account:
- 10.13.1. an electronic image of the document confirming their identity and citizenship (scanned copies of the pages with the applicant's photograph, first name, middle name, family name, sex, date and place of birth, citizenship, reference series and number of the respective document, the public authority agency that issued the document, as well as details of the applicant's place of registered residence);

- 10.13.2. an electronic image of the applicant's admission application for enrolment to the respective degree programmes;
 - 10.13.3. an electronic image of the standard education credential with a supplement;
- 10.13.4. an applicant's photograph (it must be taken in the current calendar year with a resolution of at least 800 pixels in height and 600 pixels in width);
- 10.13.5. electronic images of the documents to have them considered during the competitive selection of the documents (portfolio) according to the syllabus of the admissions test;
- 10.13.6. electronic images of the documents confirming the applicant's individual achievements (if applicable);
 - 10.13.7. other required documents.
- 10.14. The applicant shall confirm, apart from the provisions set forth in Clauses 2.21 and 2.22 herein, that they have read the guidelines of the rules of stay in the Russian Federation for foreign nationals adopted by Federal Law No 115-FZ 'On the Legal Status of Foreign Citizens in the Russian Federation' dated 25 July 2002 and other legislation of the Russian Federation by putting their handwritten signature under their admission application.
- 10.15. The foreign nationals admitted under the provisions of Clauses 10.2.2 and 10.2.3 herein shall be subject to take the same admissions tests as established for citizens of the Russian Federation when applying for admission to St Petersburg degree programmes of higher education of the respective level.
- 10.16. Enrolment of the foreign nationals admitted under the provisions of Clauses 10.2.2 and 10.2.3 herein to the government-funded places as well as to the places on the fee-paying basis paid for by individuals and/or legal entities shall be carried out complying with the procedures and terms established for the enrolment of citizens of the Russian Federation to St Petersburg University degree programmes of higher education of the respective level.
- 10.17. The established period for submitting documents by foreign nationals specified in Clause 10.5 herein applying for the admission to bachelor's, specialist's, or master's degree programmes shall be from **7 February 2022 to 17 June 2022** (to master's degree programme 'Advanced Mathematics' to 22 May 2022).
- 10.18. St Petersburg University shall consider documents submitted by foreign nationals applying for admission to degree programmes within three (3) working days from the moment the documents have been submitted. Given the complete set of documents specified in Clause 10.9 herein, the applicant shall be included in the list of foreign nationals admitted to the competitive selection procedure for degree programmes. The said list must be made available in the manner prescribed by Clause 1.6 herein. If the name of the foreign national is missing in the list of admitted to the competitive selection procedure persons after three (3) working days from the moment they submitted their documents, it shall mean that the set of the documents was incomplete and/or they didn't meet standards for the documents.
- 10.19. For the foreign nationals specified in Clause 10.5 herein, the admissions test shall be held in accordance with the admissions test syllabi published on the St Petersburg University official website. The applicants must submit all documents and materials required by the admissions tests syllabi held in the form of the competitive selection of documents (portfolio) before the deadline specified for accepting documents in Clause 10.7. herein.
- 10.20. To have their level of Russian competence recognised when applying for admission to the degree programmes implemented in Russian during the admissions test in the form of the competitive selection of the documents (portfolio), the applicants shall confirm that they have completed educational programmes of the previous level or stage (not lower than the general education level) in Russian as the confirmation of their TORFL-1 or TORFL-2 level competence.
- 10.21. The ranked lists with the ratings of the applicants applying for admission to bachelor's (specialist's) or master's degree programmes specified in Clause 10.5 herein shall be published on the St Petersburg University official website at least before **19 July 2022**.

- 10.22. Enrolment to degree programmes of higher education of the foreign nationals specified in Clause 10.5 herein shall be carried out at least before 31 August 2022 based on the decision of the Admissions Committee upon the results of admissions tests to the respective degree programmes.
- 10.23. Foreign nationals applying for admission to the places on the fee-paying basis paid for by individuals and/or legal entities shall be enrolled only after they have concluded contracts for rendering chargeable educational services.
- 10.24. Should citizens of the Republic of Belarus be admitted to bachelor's or specialist's degree programmes with equal with citizens of the Russian Federation rights, the citizens of the Republic of Belarus shall have the right to provide the results of the centralised testing underwent by them in 2021 or 2022 (hereinafter referred to as centralised testing) held in the Republic of Belarus The centralised testing results certificate shall be submitted before the deadline specified by Clause 7.3.3 herein and shall be recognised as general education tests results if the applicant did not take USE in the year when the centralised testing results certificate was obtained.

The centralised testing results marked on the 100-point scale shall be considered the result of the admissions test marked on an identical 100-point scale. The correspondence between disciplines of the centralised testing and general education admissions tests has been set in Appendix 7 to these Regulations. To be admitted to the competitive selection procedure to bachelor's or specialist's degree programmes that requires taking general education admission tests in 'Literature', 'History', Informatics and ICT' and other disciplines that do not correspond to the disciplines of the centralised testing and if there are no other disciplines of choice, citizens of the Republic of Belarus must either provide the USE results in specified general education disciplines or, under provisions of Clause 7.2.2.2 of these Regulations, undergo admissions tests held in these disciplines at the discretion of St Petersburg University.

Appendix 1 to the Regulations for Admission to Degree Programmes of Higher Education (Bachelor's Degree Programmes, Specialist's Degree Programmes) at St Petersburg University in 2022

REGULATIONS for Admissions Tests to Degree Programmes of Higher Education at St Petersburg University in 2022

1. General Provisions

- 1.1. These Regulations for Admissions Tests to Degree Programmes of Higher Education at St Petersburg University in 2022 (hereinafter referred to as Regulations for Admissions Tests) shall specify the procedures for additional practical, creative admissions tests and/or profession-specific admissions tests held at the discretion of St Petersburg University when admitting applicants to bachelor's, specialist's, or master's degree programmes in 2022.
- 1.2. These Regulations for Admissions Tests comply with Federal Law No 273-FZ 'On Education in the Russian Federation' dated 29 December 2012, Order No 1076 'Procedures for Admission to Bachelor's, Specialist's, or Master's Degree Programmes of Higher Education' of the Ministry of Science and Higher Education of the Russian Federation dated 21 August 2020 (hereinafter referred to as the Procedures for Admission), the Charter of Federal State Budgetary Institution of Higher Education 'St Petersburg State University', the Regulations for Admission to Bachelor's, Specialist's, or Master's Degree Programmes of Higher Education at St Petersburg University in 2022 (hereinafter referred to as the Regulations).
- 1.3. St Petersburg University ensures the anonymity of all examination papers of the applicants during their assessment so that the authorship cannot be identified. During admissions tests in the form of the competitive selection of the documents (portfolio), anonymity shall be ensured in the manner prescribed by the provisions of Section 5 herein.
- 1.4. The syllabi of the admissions tests held at the discretion of St Petersburg University for the persons specified in Clause 7.2.2 of the Regulations developed at St Petersburg University comply with the federal state educational standard of secondary general education.
- 1.5. Written examination, oral examination, oral and written examination, and competitive selection of documents as well as creative competition and other forms according to the decision of the Admissions Committee shall be the forms of admissions tests held at the discretion of St Petersburg University.
- 1.6. The St Petersburg University Admissions Committee shall be responsible for defining the form of the additional practical creative and/or profession-specific admissions test.
- 1.7. Admissions test held at the discretion of St Petersburg University for persons applying to a particular bachelor's or specialist's degree programme shall be held in general education disciplines specified by the Admissions Committee in compliance with Order No 666 'On Approval of the List of Admissions Tests for Admissions to Degree Programmes of Higher Education Bachelor's or Specialist's Degree Programmes'.
- 1.8. The Admission Committee shall be responsible for defining the list and the forms of admissions tests held at St Petersburg University for master's degree programmes.
- 1.9. To prepare examination materials and organise admissions tests for master's degree programmes, St Petersburg University shall appoint examination boards in the manner prescribed at St Petersburg University for organising admissions tests.
 - 1.10. The examination boards shall include the St Petersburg University research and teaching

staff and other employees, research and teaching staff of educational, research, and other organisations whose activities correspond to the field of study of the degree programme. The chairs of the examination boards shall be appointed out from the qualified research and teaching staff and other employees of educational, research and other organisations whose activities correspond to the field of study of the degree programme.

2. Procedures for Admissions Tests Organisation

- 2.1. The St Petersburg University 2022 schedule of admissions tests for admissions to degree programmes of higher education shall be approved in the manner prescribed by these Regulations.
- 2.2. Before the admissions tests take place, except for the competitive selection of the documents (portfolio), the consultations for the applicants shall be scheduled to brief them on the structure and syllabi of the admissions tests, procedures of their organisation, requirements to examination papers or answers, as well as to the criteria of their assessment.
- 2.3. The applicants must not carry into the examination rooms any personal belongings with them except materials and belongings permitted for use. The applicants must switch off any communication devices and other appliances, except for the personal computer (a laptop), video camera and a microphone, to ensure the video conferencing in advance.
 - 2.4. The applicants must not bring to the examination rooms:
- 2.4.1. Personal computers (including portable computers, pocket computers, etc.) and other computing machinery (including calculators) and other devices, except for those specified in Clause 2.3 of the Regulations for Admissions Tests;
 - 2.4.2. cell phones and other means of communication;
- 2.4.3. any stationary, except for the items permitted for use in the prescribed manner Clause 2.6.2. of the Regulations for Admissions Tests;
- 2.4.4. any reference materials (including learner's guides, monographs, lecture notes, geographic maps, atlases, etc.) unless their usage is specified by the syllabus of the relevant admissions test.
- 2.5. Should admissions test be held in a different from a competition of documents (portfolio) form, the system of video and audio recording and video conferencing shall be used (with both mandatory use of a video camera and a microphone by the applicant during the admissions test and provision St Petersburg University with the video recording of the admissions test) ensuring passing the admissions test through the St Petersburg University distance learning system when connected via the Internet, as well as providing control of compliance by the persons taking the admissions test with the procedures for admissions tests (proctoring).
- 2.6. To ensure compliance of the admissions test organisation with Clause 2.5 of the Regulations for Admissions Tests:
- 2.6.1. the examination board members, other authorised St Petersburg University personnel and/or authorised personnel of other organisations shall control the identities of applicants to ensure that the very person who applied for the admissions test is taking part in the admissions test. To do so, the applicants before the admissions test starts shall present on a video camera the original of their identity document (a page with their family name, first name, middle name and a photograph);

in the manner prescribed by Clause 1.6. of these Regulations, at least before the period of admissions tests starts, the Admission Committee shall announce the following: the information about requirements to the equipment for holding admissions tests, procedures of obtaining access to the St Petersburg University system of distance learning by the applicants, admissions tests tasks, the order of entering (typing in) the answers to the admissions tests tasks by the applicants, the algorithm the applicants have to follow before the admissions tests start to ensure personal identification and control of compliance with the regulations for admissions tests during the admissions tests, information about the possibility of earlier completion of the admissions test, whether it is allowed/not allowed to leave the examination room for short breaks, and the procedures of leaving the examination room in case such leaves are allowed.

3. Procedures for Holding Admissions Tests

- 3.1. At least two members of the respective examination board shall grade each applicant's oral part of the admissions test (interviewing) or the oral element of the test. To prepare for the oral answer, the applicant shall be given 45 minutes (20 minutes to work with the text at a foreign language examination). During the admissions test, the applicant may be asked additional questions concerning the topic of their examination card. To prepare for answering the examination card questions, the applicant shall use the blank form for oral task answers. The examiners shall keep records of the correctness and completeness of the answers to the questions from the examination card and additional questions.
- 3.2. When holding the oral part of the admissions test or the oral element of the test, the board shall keep a record of questions they ask the applicant, brief commentary of the examiners (synopsis) of the answers to them and award grades for each part of the admissions test syllabus.
- 3.3. The final grade for the oral part of the admissions test or the oral element of the test shall be put in a figure and in cursive into the sheet of the oral (oral and written) answer signed by at least two examiners.
- 3.4. The syllabus of the admissions test shall specify the duration of the admissions test conducted in written form.
- 3.5. The duration of the practical creative and/or professional-specific admissions test may be specified for each particular case in accordance with the syllabus of the respective admissions test.
- 3.6. A written examination paper with notes that can help identify the paper's authorship shall not be admitted to the assessment by the examination board. The applicant shall be awarded the lowest grade (zero score). In the above-mentioned case, a statement of the violation and the applicant's failure at their admissions test upon a non-justifiable reason shall be drawn out, and its electronic copy shall be sent to the applicant.
 - 3.7. Each applicant must perform their admissions test tasks independently.
- 3.8. During the admissions test, no interaction between the applicants and any other persons shall not be allowed except the members of the examination boards, authorised St Petersburg University personnel and/or authorised personnel of other organisations on issues directly concerning the admissions test organisation.
- 3.9. Should an applicant during the admissions test carry or use any written and/or printed materials, devices (including means of communication) prohibited for use or violate provisions of Clauses 3.7 and 3.8 of the Regulations for Admissions Tests, they shall be withdrawn from the admissions test and awarded the lowest grade (zero score) If such violations are established after the admissions test session, the applicant is awarded the lowest grade (zero score) regardless the content of the examination paper. In the above-mentioned case, a statement of the violation and the applicant's failure at their admissions test upon a non-justifiable reason shall be drawn out, and its electronic copy shall be sent to the applicant.

4. Organisation of Admissions Tests for Physically Challenged and Disabled Applicants

- 4.1. Admissions tests for physically challenged and disabled applicants shall be held with an account of their psycho-physical characteristics, their individual capabilities and health condition.
- 4.2. In their admission application, a physically challenged and disabled applicant can specify the need for an increased duration of the admissions test session. The duration of the admissions test session for physically challenged and disabled applicants, who have specified such a need, shall be increased by 60 minutes.

5. Regulations for the Competitive Selection of Documents (Portfolio)

5.1. To pass an admissions test in the form of competitive selection of documents (portfolio),

applicants, along with the documents required for admission to the competitive selection, shall submit the documents and materials that confirm their aspiration to pursue the chosen degree programme, their achievements in academic, research and practical activities.

- 5.2. The list of documents (portfolio) for the competitive selection shall be specified by the syllabus of the admissions test for each degree programme outlined by the applicant in their admission application.
- 5.3. All documents specified by the syllabus of the admissions test held in the form of the competitive selection of documents (portfolio) shall be submitted within the competitive applications acceptance periods specified in the Regulations for Admission.
- 5.4. In case their standard education credential is absent upon the circumstances beyond their control at the point of conducting an admissions test in the form of competitive selection of documents (portfolio), graduates from educational organisations shall provide a standard transcript certified by the seal of the educational organisation specifying their completion of the curriculum, successful passing of the State Final Assessment in the degree programme, the disciplines studied, assessment results, and the date of issuance of the standard credential.
- 5.5. To participate in the competitive selection for the admission to degree programmes with admissions test in the form of the competitive selection of documents (portfolio), the following documents may be considered depending on the syllabi of the relevant admissions tests:
- 5.5.1. A motivation letter containing a well-reasoned description of the aspiration and ability to study the chosen degree programme. The motivation letter must contain necessary information about professional experience and/or work of the applicant; information confirming the necessity of obtaining the knowledge, skills, and competencies specified by the chosen programme; prospects and/or plans of implementing the acquired knowledge, skills, and competences in the future professional activity;
- 5.5.2. an essay (a creative or research paper) on one specified topic. List of topics, standards applied to the length, and essay assessment criteria shall be specified in the syllabi of the respective admissions tests held in the form of competitive selection of documents (portfolio);
- 5.5.3. copies of publications in scientific sources (periodicals, collections of papers, and monographs). An applicant shall attach a copy of the title page of the journal with its masthead and the link to the research article in the Web of Science Core Collection and Scopus, Russian Science Citation Index (Elibrary) scientometric databases to the text of their publications;
- 5.5.4. documents confirming the applicants' participation in conferences, seminars, round tables and other scientific and research events. Abstracts of research papers and an event programme can be such documents;
- 5.5.5. diplomas of a winner, awardee, laureate or participant of open-all-Russia and regional competitions of student research papers and student olympiads of various levels;
- 5.5.6. documents and materials confirming both the applicant's participation in research projects supported by grants and the obtained by the results;
- 5.5.7. documents and materials confirming the applicant's work (including the efficiency) in research laboratories, research teams, etc.;
- 5.5.8. a service record statement or a copy of their labour contract certified by an authorised employee of the organisation confirming their length of service and practical experience;
- 5.5.9. statements confirming scholarships (presidential, governmental or sponsored) from educational institutions carrying out the educational activity;
 - 5.5.10. certificates confirming their foreign language level;
- 5.5.11. other documents and materials specified by the syllabus of the admissions test held in the form of competitive selection of documents (portfolio).
- 5.6. The applicants must submit their documents for the admissions test held in the form of the competitive selection of documents (portfolio) in the form of an electronic image in PDF (PortableDocumentFiles) interplatform open format for electronic documents. The electronic image of a document must ensure the visual identity to its hard copy original on a 1:1 scale. The quality of the submitted copy of a document must ensure the entire legibility of the text of the document and identification of its details. Should the hard copy of a document contain two or more pages, the electronic

images of such documents shall constitute one file. It is strongly recommended to make a scanned copy of documents with a resolution of at least 300 dpi. The applicants must not submit illegible scanned copies and images with sufficient parts missing (text segments, signatures, seals or stamps, etc.).

- 5.7. Depending on the syllabus of the respective admissions test, points may be assigned for the average diploma score and the grade for the graduation project when participating in the competitive selection to degree programmes with admissions test held in the form of competitive selection of documents (portfolio). In this case, for the point assignment, only one education credential shall be considered when applicants participate in the competitive selection for one field of study. The diploma score average shall be calculated based on the standard education credential supplement. Should a supplement to the standard education credential be missing, the average score for the particular applicant shall be zero (0) points.
- 5.8. Should they submit forged documents, documents containing deliberately falsified information, and provide papers with illegal borrowings (plagiarism) or papers performed by a third party, the applicant shall be liable in compliance with the legislation of the Russian Federation. Thus, if such facts have been established, the applicant shall be assigned the lowest point for their admissions test zero (0) points.
- 5.9. Papers specified in Clauses 5.5.1 and 5.5.2 of the Regulations for Admissions Tests (an essay and a motivation letter) shall be graded by the relevant examination board members in the electronic system of portfolio assessment as a single set of papers.
- 5.10. The essay and the motivation letter must not contain information that may help identify their authorship, except for the cases explicitly specified in the syllabus of the admissions test. Should any of the documents (the essay or the motivation latter) contain information disclosing the authorship, including those of them submitted in handwriting, the applicant shall be assigned the lowest point (zero (0) points) for both of the documents (the essay and the motivation letter)

Appendix 2 to the Regulations for Admission to Degree Programmes of Higher Education (Bachelor's Degree Programmes, Specialist's Degree Programmes, and Master's Degree Programmes) at St Petersburg University in 2022

Regulations for Review of Written Works, Appeal Submission, and Appeal Proceedings

- 1. Should an applicant disagree with the result of the respective admissions test after it has been announced, they can review their written work, written part of the oral and written admissions test, documents submitted within the portfolio, papers submitted within the practical, creative competition (hereinafter referred to as 'written work') in the manner prescribed by the Regulations for Review of Written Works, Appeal Submission, and Appeal Proceedings (hereinafter referred to as the Regulations for Review) and file in a reasoned letter of appeal (hereinafter referred to as the appeal). Such an appeal can also be submitted if the applicant believes that the Regulations for Admissions Tests have been violated.
- 2. The date of the announcement of admissions test results shall be the day the admissions test results are made available in the manner prescribed by Clause 1.6 of the Regulations for Admission.
- 3. The applicants shall be informed of the result announcement date for each admissions test, the period specified for submitting applications concerning reviewing the written papers accomplished during the admissions test, the period specified for reviewing written papers, the period of time specified for submitting the appeals, the period of time of the appeal consideration for each admissions test in the manner prescribed by Clause 1.6 of the Regulations for Admission. The above-mentioned information shall also be included in the admissions test schedule.
- 4. The information about the assessment of the documents within the portfolio shall be available for the applicant through their Personal Account with the detailed description of the assessment procedure and indication of the score of points assigned for each submitted document based on assessment criteria of the admissions test specified in the syllabus of the admissions test. Should an admissions test be held in different from a competitive selection of documents (portfolio) form, persons that have submitted their applications for reviewing, their other written works shall have access to the awarded grades for their examination paper through the St Petersburg University system of distance learning in accordance with the approved schedule of admissions test sessions.
- 5. The applicants shall file in by their letters of appeal in person addressing them to the chair of the Appeals Panel either on the same or the following working day after the result announcement day for the respective examinations test. The applicant can also file an appeal claiming a violation of the Regulations for Admissions Tests during the day a respective admissions test is being held. The applicant shall file an appeal via email to the address of the respective information centre of the Admission Committee in the manner prescribed by Clause 1.6 of the Regulations for Admission and in the form established at St Petersburg University. Thus, such an appeal may not observe standards set in Clauses 2.12, 2.13, and 8.7 of the Regulations for Admission.
- 6. The appeal must contain either specific reasons why the applicant disagrees with the grade or certain evidence of the Regulations for Admissions Tests violation.
- 7. The admissions test schedule must specify the period of considering an appeal for the particular admissions test.
- 8. An authorised St Petersburg University official shall approve the composition of the Appeals Panel.
 - 9. At least three (3) members of the Appeals Panel shall consider appeals.
- 10. The applicant shall be eligible to be present at the consideration of their appeal using a system of audio and video conferencing

- 11. The appeal consideration shall not be retaking the admissions test. The appeal proceedings shall only consider the degree of compliance with the prescribed provisions of the Regulations for Admissions Tests and/or criteria of admissions test assessment.
- 12. Upon the result of the appeal proceeding, the Appeals Panel may make the grade higher, lower, or left unchanged.
- 13. Upon the result of the appeal proceeding, the Appeals Panel shall make a decision concerning the admissions test grade. The Appeals Panel shall adopt its decisions by majority vote.
- 14. The Appeals Panel decision duly documented in the minutes and approved by the executive secretary of the Admissions Committee shall be final. The Appeals Panel decision duly documented in the minutes shall be made available for the applicant by sending it to the email address at least within the next working day after the day the appeal has been considered in accordance with the admissions tests schedule.
- 15. The Appeals Panel decisions on changing the grade for the admissions test (making it either higher or lower) shall influence the admissions test results that shall be published in the manner prescribed by Clause 1.6 of the Regulation for Admission.
- 16. The minutes of the Appeals Panel decisions shall be kept as long as prescribed by the File Register.

Appendix 3 to the Regulations for Admission to Degree Programmes of Higher Education (Bachelor's Degree Programmes, Specialist's Degree Programmes, and Master's Degree Programmes) at St Petersburg University in 2022

Consideration of the Applicants' Individual Achievements when applying for admission to degree programmes

S AND MECHANICS Advanced Mathematics	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Mathematics Fundamental Mechanics Computational Technologies (Fundamental Informatics) and Information Technologies)	Winners - 100 points Awardees - 50 points
Advanced Mathematics	University olympiad for students and young professionals	Fundamental Mechanics Computational Technologies (Fundamental Informatics)	1
	(T = 0 1 11 11 T 1 1 1 1 1 1 1 1 1 1 1 1 1 1		
	'I am a Professional' All-Russia olympiad for students	Mathematics	Students medal-holders - 100 points Winners - 25 points Awardees - 10 points
Software and Data Science	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Mathematics Computational Technologies (Fundamental Informatics) and Information Technologies)	Winners - 100 points Awardees - 50 points
All degree programmes in the field of study except 'Software and Data Science' programme	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Mathematics Fundamental Mechanics Computer Technology (Fundamental Computer Science and Information Technology)	Winners - 100 points Awardees - 50 points
All n t	degree programmes he field of study ept 'Software and ta Science' programme	degree programmes he field of study ept 'Software and 'Software and 'Days and young professionals University olympiad for students and young University olympiad for students and young professionals	University olympiad for students and young professionals Detropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals Detropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals Detropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals Detropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals Computer Technology (Fundamental Computer Science and Information Technology)

		' I am a Professional ' All-Russia olympiad for students	Information and Cyber Security Artificial Intelligence Mathematics Mathematical Modelling Robotics Programming and Information Technology Control in Engineering Systems	Students medal-holders - 100 points Winners - 50 points Awardees - 25 points
		'Lomonosov' universiade	Applied Mathematics and Informatics	Winners - 10 points Awardees - 5 points
		Vysshaia Liga olympiade for students and alumni	Mathematics Computer Networks and Systems Applied Mathematics Applied Mathematics and Informatics Game theory	Students medal-holders - 15 points Winners - 10 points Awardees - 5 points
	Mechanics and Mathematical Modelling	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Mathematics Fundamental Mechanics Computational Technologies (Fundamental Informatics and Information Technologies)	Winners - 100 points Awardees - 50 points
		'I am a Professional' All-Russia olympiad for students	Mathematical Modelling	Students medal-holders - 100 points Winners - 50 points Awardees - 25 points
		Vysshaia Liga olympiade for students and alumni	Mathematics Applied Mathematics	Students medal-holders - 15 points Winners - 10 points Awardees - 5 points
02.00.00 COMPUTER	AND INFORMATION SC	IENCES		
02/04/2002 Fundamental Informatics and Information Technology	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Mathematics Fundamental Mechanics Computer Technology (Fundamental Computer Science and Information Technology)	Winners - 100 points Awardee- 50 points

		'I am a Professional' All-Russia olympiad for students	Information and Cyber Security Artificial Intelligence Mathematical Modelling Programming and Information Technology Robotics Control in engineering systems	Students medal-holders 100 points Winners - 50 points Awardee- 25 points
		'Digital Breakthrough' All-Russia contest Vysshaia Liga olympiade for students and alumni 'Abilympics' National championship in professional competency among individuals with a disability or additional support needs		Winners - 10 points
			Mathematics Computer Networks and Systems Applied Mathematics Applied Mathematics and Informatics	Students medal-holders 15 points Winners - 10 points Awardees - 5 points
			Database administration Web development (Programming) Information Security Virtual and augmented reality development Robotics Network and System Administration	Winners - 10 points
		WorldSkills Russia all-Russia championships in professional competence	DigitalSkills championship in a high-tech profession	Winners - 10 points
Administration of A	Software and Administration of Information Systems	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Mathematics Fundamental Mechanics Computer Technology (Fundamental Computer Science and Information Technology)	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Information and Cyber Security Artificial Intelligence Mathematical Modelling Programming and Information Technology Robotics Control in engineering systems	Students medal-holders 100 points Winners - 50 points Awardee- 25 points
		'Digital Breakthrough' All-Russia contest		Winners - 10 points
		Vysshaia Liga olympiade for students and alumni	Mathematics Computer Networks and Systems Applied Mathematics	Students medal-holders 15 points Winners - 10 points

			Applied Mathematics and Informatics	Awardees - 5 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Database administration Web development (Programming) Information Security Virtual and augmented reality development Robotics Network and System Administration	Winners - 10 points
		WorldSkills Russia all-Russia championships in professional competence	DigitalSkills championship in high-tech professions	Winners - 10 points
03.00.00 PHYSICS AN	D ASTRONOMY	<u> </u>	<u> </u>	
03.04.01 Applied Mathematics and Physics	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Physics Mathematics Fundamental Mechanics Computer Technology (Fundamental Computer Science and Information Technology)	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Physics and Technologies of the unique MEGA-Science facilities Control in Engineering systems Programming and Information Technology	Students medal-holders - 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Physics	Students medal-holders - 15 points Winners - 10 points Awardees - 5 points
		NRNU MEPHI International Physics and Engineering Olympiad for students		Winners - 10 points Awardees - 5 points
		CASE-IN International Engineering Championship	Main fields 'Digital Atom', 'Electroenergetics''	Winners - 10 points
03.04.02 Physics	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Physics	Winners - 100 points Awardee- 50 points

		'I am a Professional' All-Russia olympiad for students	Physics Physics and Technologies of the unique MEGA-Science facilities	Students medal-holders - 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Physics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Lomonosov' universiade	Basic and Applied Physics	Winners - 10 points Awardees - 5 points
		NRNU MEPHI International Physics and Engineering Olympiad for students		Winners - 10 points Awardees - 5 points
		CASE-IN International Engineering Championship	Fields of study 'Digital Atom', 'Electroenergetics''	Winners - 10 points
04.00.00 CHEMISTRY	Ž			
04/04/2001 Chemistry	Chemistry	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	International Tournament in Natural Sciences	Winners - 25 points Awardee- 10 points
		'I am a Professional' All-Russia olympiad for students	Chemistry Chemical Technology Petroleum Engineering	Students medal-holders - 15 points Winners - 10 points Awardee- 5 points
		CASE-IN International Engineering Championship	Field of study 'Oil Chemistry'	Winners - 10 points
05.00.00 EARTH SCIE	ENCES			
05.04.2001 Geology	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Earth Sciences	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Geology Petroleum Engineering	Students medal-holders - 100 points Winners - 50 points Awardee- 25 points
		'Lomonosov' universiade	Geology	Winners - 10 points Awardees - 5 points

		CASE-IN International Engineering Championship	Fields of study 'Mining and Mineral Engineering', Geological Prospecting', 'Oil Chemistry'	Winners - 10 points
05.04.02 Geography	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Earth Sciences	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Geography Global Ocean Ecology	Students medal-holders - 100 points Winners - 50 points Awardee- 25 points
		'Lomonosov' universiade	Geography and Tourism	Winners - 10 points Awardees - 5 points
05.04.03 Cartography and Geoinformatics	Geographical Information Mapping	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Earth Sciences	Winners - 100 points Awardee- 50 points
		'I am a Professional' All– Russia olympiad for students	Geography	Students medal-holders - 100 points Winners - 50 points Awardee- 25 points
05.04.04 Hydrometeorology	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Earth Sciences	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Geography Global Ocean Ecology	Students medal-holders - 100 points Winners - 50 points Awardee- 25 points
05/04/2006 Ecology and Nature Management	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Earth Sciences	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Biotechnologies Geography Ecology	Students medal-holders - 100 points Winners - 50 points Awardee- 25 points
		'Lomonosov' universiade	Soil Science and Ecology	Winners - 10 points Awardees - 5 points

		'Moia strana - moia Rossiia' all-Russia contest	'Ecology of my native country' nomination	Winners - 10 points
05.04.06 Ecology and	Ecology Biodiversity and	'Lomonosov' universiade	Biology	Winners - 10 points
Nature Management	Nature Protection		Soil Science and Ecology	Awardees - 5 points
06.00.00 BIOLOGICA	L SCIENCES			
06.04.01 Biology	All degree programmes within this field of study	'I am a Professional' All-Russia olympiad for students	Bioengineering and Bioinformatics Biotechnologies Genetic Technologies	Students medal-holders - 100 points Winners - 50 points Awardee- 25 points
		'Lomonosov' universiade	Biology	Winners - 10 points Awardees - 5 points
06.04.2002 Soil Science	Soil Science	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Earth Sciences	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Ecology Biotechnologies	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Lomonosov' universiade	Soil Science and Ecology	Winners - 10 points Awardees - 5 points
09.00.00 INFORMAT	 ICS AND COMPUTER SCI	 ENCE		
09.04.03 Applied	Artificial Intelligence and	Petropolitan Science (Re)Search, St Petersburg	Mathematics	Winners - 100 points
Informatics	Data Science	University olympiad for students and young professionals	Fundamental Mechanics Computer Technology (Fundamental Computer Science and Information Technology)	Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Information and Cyber Security Artificial Intelligence Mathematical Modelling Programming and Information	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
			Technologies Software Engineering Robotics	

		Vysshaia Liga olympiade for students and alumni	Mathematics Computer Networks and Systems Applied Mathematics and Informatics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Database administration Web development (Programming) Information Security Virtual and augmented reality development Robotics Network and System Administration	Winners - 10 points
09.04.04 Software Soft Engineering	Software Engineering	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Mathematics Fundamental Mechanics Computer Technology (Fundamental Computer Science and Information Technology)	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Information and Cyber Security Artificial Intelligence Mathematical Modelling Programming and Information Technologies Software Engineering Robotics	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Digital Breakthrough' All-Russia contest		Winners - 10 points
		Vysshaia Liga olympiade for students and alumni	Mathematics Computer Networks and Systems Applied Mathematics Applied Mathematics and Informatics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Database administration Web development (Programming) Information Security Virtual and augmented reality development Robotics Network and System Administration	Winners - 10 points
21.00.00 APPLIED GF	 EOLOGY, MINING, PETI	ROLEUM ENGINEERING AND GEODESY		

	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Earth Sciences	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Geology Mining Engineering Petroleum Engineering	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		CASE-IN International Engineering Championship	Field of study 'Petroleum Engineering'	Winners - 10 points
		'Gazprom' olympiad for students	Profile 'Petroleum Engineering'	Winners– 10 points
21.04.02 Land Use and Cadastres	Management of Real Estate Objects and Development of	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Earth Sciences	Winners - 100 points Awardee- 50 points
	Territories	'I am a Professional' All-Russia olympiad for students	Geography	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
27.00.00 CONTROL IN	N ENGINEERING SYSTE	MS		1
27/04/2003 Systems Analysis and Control	Engineering Informatics and Artificial Intelligence	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Computer Technology (Fundamental Computer Science and Information Technology)	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Artificial Intelligence Control in Engineering Systems Robotics	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Digital Breakthrough' All-Russia contest		Winners - 10 points
		Vysshaia Liga olympiade for students and alumni	Mathematics Computer Networks and Systems Applied Mathematics Applied Mathematics and Informatics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points

		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Database administration Web development (Programming) Information Security Virtual and augmented reality development Robotics Network and System Administration	Winners - 10 points
28.00.00 NANOTECH	NOLOGY AND NANOMA	TERIALS		
28.04.04 Nanosystems and Nanomaterials	Fundamental and Applied Aspects of Nanomaterials and Nanotechnology	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	International Tournament in Natural Sciences	Winners - 25 points Awardee- 10 points
		'I am a Professional' All-Russia olympiad for students	Physics and Technologies of the unique MEGA-Science facilities Robotics	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		Vysshaia Liga olympiade for students and alumni	Physics	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		'Lomonosov' universiade	Chemistry, Physics and Material Mechanics	Winners - 10 points Awardee- 5 points
37.00.00 PSYCHOLO	GICAL SCIENCES			
37.04.01 Psychology	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Psychology	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Psychology	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Psychology	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points

		'Lomonosov' universiade	Psychology	Winners - 10 points Awardees - 5 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Psychology	Winners - 10 points
37.04.01 Psychology	Educational Psychology	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Psychology	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Psychology	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
			Preschool Pedagogy Basic Pedagogy Special (defectological) education	Students-medal holders - 50 points Winners - 25 points Awardee- 10 points
		Vysshaia Liga olympiade for students and alumni	Psychology Education and Pedagogics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Lomonosov' universiade	Psychology	Winners - 10 points Awardees - 5 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Psychology	Winners - 10 points
		'Flagmany Obrazovaniia' all-Russia professional contest		Winners - 10 points
37.04.02 Conflictology	Conflict Management	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Psychology	Winners - 100 points Awardee- 50 points

		'I am a Professional' All-Russia olympiad for students	Psychology	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Psychology	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Lomonosov' universiade	Psychology	Winners - 10 points Awardees - 5 points
		'Upravlay' all-Russia youth cup in management		Winners - 10 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Psychology	Winners - 10 points
38.00.00 ECONOMIC	S AND MANAGEMENT			
38.04.01 Economics	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Economics and Finance	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Economics	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Economics	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Economics and Accounting Entrepreneurship	Winners - 10 points

	Management of Organisational Development Accounting, Analysis and	'Upravlay' all-Russia youth cup in management	Economics and Accounting Entrepreneurship	Winners - 10 points
	Audit Economics and Enterprise Management Company Economics and Enterprise Management	'Digital Breakthrough' All-Russia contest		Winners - 10 points
	Mathematical Approach to Economics	'Lomonosov' universiade	Mathematical Approach to Economics	Winners - 10 points Awardee- 5 points
38.04.02 Management	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Urban Development Management(Management)	Winners - 100 points Awardees - 50 points
		'I am a Professional' All-Russia olympiad for students	Management	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Management	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		'Upravlay' all-Russia youth cup in management		Winners and awardees (10 points)
		'Digital Breakthrough' All-Russia contest		Winners - 10 points
		'My pilot business project' international contest		Winners - 10 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Entrepreneurship	Winners - 10 points
		'Moia strana - moia Rossiia' all-Russia contest	'My entrepreneur initiative. Creative industries for regional development' nomination	Winners - 10 points

38.04.03 Personnel Management	Human Resource Management	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Economics and Finance	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Personnel Management	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Economics and Accounting Entrepreneurship	Winners - 10 points
		'Upravlay' all-Russia youth cup in management		Winners and awardees (10 points)
38.04.04 Public and Municipal Administration	Smart City Management (Master in Smart City Management - MSC)	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Urban Development Management(Management)	Winners - 100 points Awardees - 50 points
		'I am a Professional' All-Russia olympiad for students	Public Administration	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Public Administration Urban Spatial Development Management	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		'Lomonosov' universiade	Public Administration	Winners - 10 points Awardee- 5 points
		'Upravlay' all-Russia youth cup in management		Winners - 10 points
41.04.04 Political Science	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Political Science	Winners - 100 points Awardee- 50 points

		'History of my local governance' all-Russia contest	Nominations: 'My Community (district, street) Self-Governing Committee working experience'; 'History of my local governance' Research'; 'History of my local governance' videoclip	Winners - 5 points
38.04.05 Business Informatics	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Economics and Finance	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Business Informatics	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Business Informatics	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		'Digital Breakthrough' All-Russia contest		Winners - 10 points
38.04.08 Finance and Credit	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Economics and Finance	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Finance and Credit	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Finances and Investing	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Economics and Accounting Entrepreneurship	Winners - 10 points

39.04.01 Sociology	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Social Science (Sociology, Social Work)	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Sociology and Social Work	Students-medal holders 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Social Studies	Students-medal holders 15 points Winners - 10 points Awardee- 5 points
		'Lomonosov' universiade	Social Studies	Winners - 10 points Awardees - 5 points
		'History of my local governance' all-Russia contest	Nominations: 'My Community (district, street) Self-Governing Committee working experience,' 'History of my local governance' Research'; 'History of my local governance' videoclip	Winners - 5 points
9.04.02 Social Work	Social Work	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Social Science (Sociology, Social Work)	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Sociology and Social Work	Students-medal holders 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Social Studies	Students-medal holders 15 points Winners - 10 points Awardee- 5 points
		'Lomonosov' universiade	Social Studies	Winners - 10 points Awardees - 5 points
		'#MYVMESTE' international award	Volunteers and NPOs Treck	Winners - 10 points
		'The Best of the Projects in Social Work' contest		Winners - 10 points

		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Medical and Social Care Social Work	Winners - 10 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives	Field of study 'Volunteering', 'Social upward mobility development'	Winners - 10 points
40.00.00 LAW		<u> </u>	1	
40.04.01 Law	Bankruptcy Law Civil Procedure, Arbitration Procedure Civil Law, Family Law	'I am a Professional' All-Russia olympiad for students	Law (Civil Law profile)	Medal-holders: Gold - 100 points Silver - 80 points Bronze - 60 points Winners - 10 pointsinners - 10 points
	International Private Law Real Estate Lawyer Business Law Corporate Lawyer Legal support of business	'Moia strana - moia Rossiia' all-Russia contest	'Intellectual property of my native country' nomination	Winners - 10 points
	Legal Protection of Economic Competition Transnational Legal Practice (in English) Legal support of international business (In English) Lawyer Financial Markets (Finance Lawyer)	Vysshaia Liga olympiade for students and alumni	Civil Law	Students-medal holders - 10 points

40.04.01 Law	Tax Law Criminal Law Ecological and legal support of economic activity Lawyer in the field of normative activity Sports Lawyer Lawyer in the Field of Criminal Justice Transnational Legal Practice (in English) Legal support of international business In English	'I am a Professional' All-Russia olympiad for students	Law (Civil Law profile)	Medal-holders: Gold - 100 points Silver - 80 points Bronze - 60 points Winners - 10 pointsinners - 10 points
40.04.01 Law	Medical Law, Pharmaceutical Law	'I am a Professional' All-Russia olympiad for students	Law (Civil Law profile) Law (Civil Law profile) General Medicine and Paediatrics Medical and Preventive Care Dentistry	Medal-holders: Gold - 100 points Silver - 80 points Bronze - 60 points Winners - 10 points
40.04.01 Law	Bankruptcy Law International Public Law Ecological and legal support of economic activity	'I am a Professional' All-Russia olympiad for students	Law (Civil Law profile)	Medal-holders: Gold - 100 points Silver - 80 points Bronze - 60 points Winners - 10 points

	T	T		
	Lawyer in the field of normative activity			
	Sports Lawyer			
	Tax Law			
	Transnational Legal Practice (in English) Legal support of international business In English			
40.04.01 Law	International Public Law	International Public Law essay contest at Saint- Petersburg State University		Winners - 30 points
	International Public Law	Vysshaia Liga olympiade for students and alumni	International Law	Students-medal holders - 10 points
	Lawyer Financial Markets (Finance Lawyer)	Vysshaia Liga olympiade for students and alumni	Public Law and Public Finances	Students-medal holders - 10 points
40.04.01 Law	Civil Procedure, Arbitration Procedure	Vysshaia Liga olympiade for students and alumni	Judicial Power and Procedural Law	Students-medal holders - 10 points
	Lawyer in the Field of Criminal Justice			
	Legal Protection of Economic Competition	Vysshaia Liga olympiade for students and alumni	Public Law and Public Finances (Track Public Law), Economics (track Economics), Civil Law (track Civil Law and Civil Procedure)	Students-medal holders - 10 points
	Legal Protection of Economic Competition	'Tochka rosta' all-Russia essay contest for undergraduate and graduate students		Winners - 20 points Awardees– 10 points
	Lawyer in the field of normative activity	'History of my local governance' all-Russia contest	Nominations: 'My Community (district, street) Self-Governing Committee working experience,'	Winners - 5 points

40.04.01 Law	All degree programmes	All-Russian Student Law Olympiad	'History of my local governance' Research'; 'History of my local governance' video- clip	Winners - 5 points
	within this field of study	Vysshaia Liga olympiade for students and alumni	Political Science	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		'Lomonosov' universiade	Political Science	Winners - 10 points Awardee- 5 points
		'Moia strana - moia Rossiia' all-Russia contest	'My entrepreneur initiative. Creative industries for regional development' nomination	Winners - 10 points
		'Leaders of Russia. Politics' all-Russia contest		Winners - 10 points
		'Contest for Polyglots' all-Russia project		Winners - 10 points
		'History of my local governance' all-Russia contest	Nominations: 'My Community (district, street) Self-Governing Committee working experience,' 'History of my local governance' Research'; 'History of my local governance' videoclip	Winners - 5 points
	Political Governance and Public Policy (in English) Russia and China in the global politics (in English)	'Upravlay' all-Russia youth cup in management		Winners - 10 points

41.04.05 International Relations	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	International Relations	Winners - 100 points Awardee- 50 points
		Vysshaia Liga olympiade for students and alumni	International Relations	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		'Lomonosov' universiade	International Relations International links and Global Research	Winners - 10 points Awardee- 5 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives	Sub-areas 'Promoting the strengthening of the youth inter-confessional and international cooperation', 'Prevention and counteraction to extremism'	Winners - 10 points
	Artificial Intelligence and international security	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	International Relations	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Artificial Intelligence	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	International Relations	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		'Lomonosov' universiade	International Relations International links and Global Research	Winners - 10 points Awardee- 5 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives	Sub-areas 'Promoting the strengthening of the youth inter-confessional and international cooperation', 'Prevention and counteraction to extremism'	Winners - 10 points
	A AND INFORMATION L			
42.04.01 Advertising and Public Relations	All degree programmes within this field of study	'I am a Professional' All-Russia olympiad for students	Advertising and Public Relations in Politics Organising cultural and exhibition events	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points

		Vysshaia Liga olympiade for students and alumni	Media Communications	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives	Field of study 'Youth Media'	Winners - 10 points
		'Leader in the Internet Communication' all-Russia project		Winners - 10 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Publishing Multimedia Journalism Text Processing Photographer-Reporter	Winners - 10 points
		'TopBLOG' all-Russia game project-contest		Winners - 10 points
	Managing Communication projects in Politics and Business	'I am a Professional' All-Russia olympiad for students	Advertising and Public Relations in Politics	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Media Communications	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives	Field of study 'Youth Media'	Winners - 10 points
		'Leader in the Internet Communication' all-Russia project		Winners - 10 points
		'Upravlay' all-Russia youth cup in management		Winners - 10 points
42.04.02 Journalism	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Interdisciplinary research (Arts and Humanitarian Sciences)	Winners - 100 points Awardee- 50 points

	'I am a Professional' All-Russia olympiad for students	Journalism Linguistics Philology Linguistics and Literary Studies	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
	'Lomonosov' universiade	Journalism	Winners - 10 points Awardee- 5 points
	The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives	Field of study 'Youth Media'	Winners - 10 points
	'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Publishing Multimedia Journalism Text Processing Photographer-Reporter	Winners - 10 points
	'TopBLOG' all-Russia game project-contest		Winners - 10 points
International Journalism	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	International Relations interdisciplinary research (Arts and Humanitarian Sciences)	Winners - 100 points Awardee- 50 points
	'I am a Professional' All-Russia olympiad for students	Linguistics Philology Linguistics and Literary Studies Journalism	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
	'Lomonosov' universiade	Journalism	Winners - 10 points Awardee- 5 points
	The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives	Field of study 'Youth Media'	Winners - 10 points
	'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Publishing Multimedia Journalism Text Processing Photographer-Reporter	Winners - 10 points
43.00.00 SERVICES AND TOURISM			

43.04.02 Tourism	Tourist Destination Management	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Earth Sciences	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Ecology Tourism	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Lomonosov' universiade	Geography and Tourism	Winners - 10 points Awardee- 5 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives	Sub-area 'Development of Youth Tourism and Regional Studies'	Winners - 10 points
		'Masters of Hospitality' all-Russia contest		Winners - 10 points
		'Moia strana - moia Rossiia' all-Russia contest	'My Hospitable Russia' nomination	Winners - 10 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Tourism	Winners - 10 points
44.00.00 EDUCATION	<u> </u> N AND PEDAGOGICAL SC	CIENCES		<u> </u>
44.04.01 Pedagogical Education	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	All fields of study	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Class supervisor Pedagogical education (preschool) Pedagogical education (basic) Special (defectological) education	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Education and Pedagogics	Students-medal holders - 15 points Winners - 10 points

				Awardees - 5 points
		'Flagmany Obrazovaniia' all-Russia professional contest 'Uchitel Buduschego' all-Russia contest		Winners - 30 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Elementary school teachers	Winners - 10 points
		'Moia strana - moia Rossiia' all-Russia contest	'Moya Pedagogicheskaia Initsiativa (National project 'Obrazovaniie' nomination)	Winners - 10 points
45.00.00 LINGUISTI	CS AND LITERARY STUD	DIES	1	
45.04.01 Philology	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Philology Linguistics	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Linguistics Philology Linguistics and Literary Studies	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		Vysshaia Liga olympiade for students and alumni	Philology	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Lomonosov' universiade	Philology	Winners - 10 points Awardees - 5 points
		'Moia strana - moia Rossiia' all-Russia contest	''My native tongue' nomination	Winners - 10 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives	Sub-area 'Literature and History'	Winners - 10 points
			Sub-area 'Literature and History'	Winners - 10 j

45.04.01 Philology	Literary Art	'Vysshaia Liga' olympiade for students and alumni	Philology Literary Mastership	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
45.04.02 Linguistics	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Philology Linguistics	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia olympiad for students	Linguistics Philology Linguistics and Literary Studies	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Vysshaia Liga' olympiade for students and alumni	Linguistics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Translator	Winners - 10 points
	Translation(Interpreting) and Business Communication in Atomic Energy	'I am a Professional' All-Russia olympiad for students	Linguistics Philology Linguistics and Literary Studies Thermal power engineering and thermal engineering Electrical Engineering	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
	Artificial Intelligence and Speech Activity Modelling	'I am a Professional' All-Russia olympiad for students	Linguistics Philology Linguistics and Literary Studies Artificial Intelligence	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
46.00.00 HISTORY A	ND ARCHAEOLOGY			-
46.04.01 History	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	History of Arts	Winners - 100 points Awardee- 50 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives	Field of study 'Patriotic Education' Sub-area 'Literature and History'	Winners - 10 points
			·	

		'Vysshaia Liga' olympiade for students and alumni	History History of Arts	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Lomonosov' universiade	Theory and History of Art	Winners - 10 points Awardee- 5 points
46.04.03 Anthropology and Ethnology	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	History of Arts	Winners - 100 points Awardee- 50 points
		'Vysshaia Liga' olympiade for students and alumni	History History of Arts	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Lomonosov' universiade	Theory and History of Art	Winners - 10 points Awardee- 5 points
	Y, ETHICS, AND RELIGI		T =	
47.04.01 Philosophy	All degree programmes within this field of study	'Vysshaia Liga' olympiade for students and alumni	Philosophy	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
50.00.00 ART STUDIE	S			
50.04.01 Arts and Humanities	Art Criticism Music Criticism	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Interdisciplinary research (Arts and Humanitarian Sciences)	Winners - 100 points Awardee- 50 points
	Curatorial Research	The Federal Agency for Youth Affairs (among	Fields of study 'Theatre and	Winners - 10 points
	Historical Performance Using Keyboard Musical Instruments	individuals) grant contest of the youth initiatives	Cinematography' 'Music and choreography', 'Artistic work'	
50.04.01 Arts and Humanities	Intercultural Education	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Interdisciplinary research (Arts and Humanities) Linguistics Pedagogy	Winners - 100 points Awardee- 50 points

			Psychology	
		'I am a Professional' All-Russia olympiad for students	Class supervisor Pedagogical education (preschool) Pedagogical education (basic) Special (defectological) education	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Flagmany Obrazovaniia' all-Russia professional contest 'Uchitel Buduschego' all-Russia contest		Winners - 30 points
50.04.01 Arts and Humanities	Cognitive Studies	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	Linguistics Interdisciplinary research (Arts and Humanities) Psychology	Winners - 100 points Awardee- 50 points
50.04.03 History of Arts	All degree programmes within this field of study	Petropolitan Science (Re)Search, St Petersburg University olympiad for students and young professionals	History of Arts	Winners - 100 points Awardee- 50 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives	Fields of study 'Theatre and Cinematography' 'Music and choreography', 'Artistic work'	Winners - 10 points
51.00.00 CULTURAI	L STUDIES AND SOCIOCU	LTURAL PROJECTS	1	
51.04.01 Cultural Studies	All degree programmes within this field of study	'Vysshaia Liga' olympiade for students and alumni	Cultural Studies	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
54.00.00 FINE AND A	APPLIED ARTS			
54.04.01 Design	All degree programmes within this field of study	'I am a Professional' All-Russia olympiad for students	Design	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Otlichno!' all-Russia youth festival of design	Graphic Design	Winners - 10 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives	Sub-areas 'Architecture, design, urbanism', 'Artistic work'	Winners - 10 points

		'Digital Breakthrough' All-Russia contest		Winners - 10 points
		'Abilympics' National championship in professional competency among individuals with a disability or additional support needs	Character design/Animation Poster design Landscape design Artistic design	Winners - 10 points
54.04.04 Restoration	Restoration of Fine and Applied Art Objects	'Restavrator goda' contest of the professional mastership	Restoration	Winners - 25 points Awardee- 10 points
58.04.01 Asian and African Studies	All degree programmes within this field of study	'I am a Professional' All-Russia olympiad for students	Asian Studies Chinese Studies, Korean Studies, Japanese Studies	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
58.04.01 Asian and African Studies	All degree programmes within this field of study	'Vysshaia Liga' olympiade for students and alumni	Asian Studies Contemporary Asia	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
All fields of study	All degree programmes	'Leaders of Russia' all-Russia contest for managers		Winners - 25 points
All fields of study	All degree programmes	'Student of the year' Russian national award		Winners– 10 points
All fields of study	All degree programmes	'Tvoi Khod' all-Russia contest		Winners– 10 points
All fields of study	All degree programmes	'Zolotaia stazhirovka' all-Russia contest		Winners– 10 points
All fields of study	All degree programmes	All-Russia Population Census		Censor – 5 points
All fields of study	All degree programmes	'Moia strana - moia Rossiia' all-Russia contest	'Commonwealth of our countries' nomination. (The nomination for foreigners and compatriots residing abroad)	Winners– 5 points

- 1. The consideration of the individual achievements of persons applying for master's degree programmes shall be carried out by assigning and adding points to the points awarded for the admissions test (admissions tests).
- 2. Points for individual achievements shall be assigned and added only if the applicant obtains at least 65 points for the admissions tests according to the admissions tests priority set forth by the Admissions Committee.
- 3. Should there be several types of individual achievements specified for one field of study/degree programme, the points shall be assigned only for one type of the specified individual achievements (upon the applicant's choice). In case there are several individual achievements of a similar type (several olympiad diplomas of a winner/awardee, several certificates of online course completion), the points shall be assigned only for one individual achievement.

- 4. In their admission application, the applicant must specify the particular individual achievement for which they may be assigned additional points (if the applicant has several individual achievements when being admitted to one field of study/degree programme, they must specify only one of them) and specify the details of the document confirming this particular individual achievement.
 - 5. The applicant must attach documents confirming their individual achievements to their personal application form.
- 6. Should the admissions test be held in the form of a competitive selection of documents (portfolio) and points be assigned to the applicant for the submitted particular documents (portfolio), points for other individual achievements based on these particular documents (portfolio) considered earlier shall not be assigned.
- 7. The results of accomplishing open online courses at St Petersburg University shall be considered as individual achievements during the admissions procedure to master's degree programmes observing the correspondence between open online courses at St Petersburg University and degree programmes (enlarged groups of fields of study) to which the admission has been announced at St Petersburg University. The applicant shall be assigned five (5) points if they provide both a certificate of the successful completion of any open online course at St Petersburg University and meet requirements outlined in paragraph 2 of the Considerations of the Applicant's Individual Achievements when applying to master's degree programmes.

The correspondence between the open online courses at St Petersburg University and the degree programmes (enlarged groups of fields of study) shall be established as follows:

THE COURSE TITLE	FIELD OF STUDY	
PHYSICAL AND MATHEMATICAL STUDIES		
Competitive Programmer's Core Skills	02.00.00 Computer and Information Sciences	
Computational Geometry	02.00.00 Computer and Information Sciences	
Higher Mathematics Algebra: Introduction to Group Theory	09.00.00 Informatics and Computer Engineering	
Algorithms on Graphs	02.00.00 Computer and Information Sciences	
WordPress web-design	02.00.00 Computer and Information Sciences	
Introduction to Web Data Analysis	02.00.00 Computer and Information Sciences	
'Functional Programming in Haskell and everywhere' (Any of the three courses)	02.00.00 Computer and Information Sciences	

'History of Mathematics: Methodology'	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences
'History of Mathematics: Chronology'	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences
Databases	01.00.00 Mathematics and Mechanics
Introduction to Data Science	02.00.00 Computer and Information Sciences
Computer Architecture	01.00.00 Mathematics and Mechanics
Mathematical Approach to Psychology Fundamentals of Application'	01.00.00 Mathematics and Mechanics
Introduction to Deformable Body Mechanics	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences
Introduction to Theory of Cybernetic Systems	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences 01.04.2002 Applied Mathematics and Informatics
Quantum Computing	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences 01.04.2002 Applied Mathematics and Informatics
Neutron scattering methods: diffraction, spectroscopy, SANS, reflectometry	03.00.00 Physics and Astronomy

Quantum Computing. Fewer Formulas - More Understanding	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences 01.04.2002 Applied Mathematics and Informatics
The Introduction to Quantum Computing	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences 01.04.2002 Applied Mathematics and Informatics
Physical Principles of Quantum Computing	01.00.00 Mathematics and Mechanics 03.00.00 Physics and Astronomy
Physical Principles of Quantum Computing	01.00.00 Mathematics and Mechanics 03.00.00 Physics and Astronomy
Database and Work with Data	01.00.00 Mathematics and Mechanics
Fundamentals of Electron and Ion Microscopy (Fundamentals of electron and ion microscopy)	03.00.00 Physics and Astronomy
Phase Transitions in Bulk Materials and Low-Dimensional Systems	03.00.00 Physics and Astronomy
Introduction to Data Analysis	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences
One-way Quantum Computation	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences
Nuclear Physics	03.00.00 Physics and Astronomy

Statistics with R made simple	04.00.00 Chemistry
(Any course from the specialisation)	05.00.00 Earth Sciences
	06.00.00 Biological Sciences
	27.00.00 Control in Engineering Systems
	32.00.00 Health Sciences and Preventive Medicine
	37.00.00 Psychological Sciences (Degree programme 'General and Cognitive
	Psychology')
	38.00.00 Economics and Management
	39.00.00 Sociology and Social Work
	50.04.01 Arts and Humanities (degree programme 'Cognitive Studies')
Mathematical Game Theory (in any language)	01.00.00 Mathematics and Mechanics
	02.00.00 Computer and Information Sciences
	01.04.2002 Applied Mathematics and Informatics
Creation of three-dimensional models of objects based on	02.00.00 Computer and Information Sciences
digital images	01.04.2002 Applied Mathematics and Informatics
University lessons on Physics	03.00.00 Physics and Astronomy
Web Analytics Elementary level	02.00.00 Computer and Information Sciences
Web Analytics Advanced level	02.00.00 Computer and Information Sciences
Amusing blockchain	01.00.00 Mathematics and Mechanics
	02.00.00 Computer and Information Sciences
BlockChain and Finances: Concepts, Technologies,	01.00.00 Mathematics and Mechanics
and Control	02.00.00 Computer and Information Sciences
Algorithms with Numbers	01.00.00 Mathematics and Mechanics

	02.00.00 Computer and Information Sciences
Law Issues in BlockChain	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences 40.00.00 Law
T .	NATURAL SCIENCES
Introduction to Bioinformatics: Metagenomics	06.00.00 Biological Sciences 09.00.00 Informatics and Computer Engineering
Inorganic Chemistry: Introduction to the Chemistry of Elements	04.00.00 Chemistry
Substance Structure: From Nuclei and Molecules to Matters and Nanoparticles	04.00.00 Chemistry
Introduction to Physiology	06.00.00 Biological Sciences
Economy of Mineral Raw Materials	05.00.00 Earth Sciences
Ecology and Environment	05.00.00 Earth Sciences
Photometric Processing of UAV Aerial Photographs	05.00.00 Earth Sciences 21.00.00 Applied Geology, Mining, Petroleum Engineering, and Geodesy
UAV Aerial Photography Technology	05.00.00 Earth Sciences 21.00.00 Applied Geology, Mining, Petroleum Engineering, and Geodesy
Coronaviruses SARS-Co-2 and Pathogens of Acute Respiratory Viral Infections	06.00.00 Biological Sciences

Symbiosis genetics (Principles of Formation of Plant-Microbe Genetic Systems)	06.00.00 Biological Sciences	
Fundamentals of Rational Agro-landscape Management (Principles of Agriculture)	06.00.00 Biological Sciences	
Transcription and the RNA World	06.00.00 Biological Sciences	
Developmental Genetics	06.00.00 Biological Sciences	
Life Safety	44.00.00 Education and Pedagogical Sciences	
Cutting-Edge Pedagogics. Its ways.	44.00.00 Education and Pedagogical Sciences	
Modern educational technologies	44.00.00 Education and Pedagogical Sciences	
Modern Issues of Continuous Education	44.00.00 Education and Pedagogical Sciences	
Graphic Visualisation of Research and Projects	44.00.00 Education and Pedagogical Sciences	
Creating Science, Technology, and Creativity Studio at School	44.00.00 Education and Pedagogical Sciences	
Introduction to Geology of Minerals	05.00.00 Earth Sciences	
Protists: Evolution and Ecology of Microbial Eukaryotes	06.00.00 Biological Sciences	
Introduction to Parasitology	06.00.00 Biological Sciences	
SOCIAL SCIENCES		

Psychology of Consciousness	37.00.00 Psychological Sciences (degree programmes 'Psychology of Education', 'Psychology of Physical Culture and Sports', 'Organisational Psychology and Psychology of Management', 'Personality Psychology') 44.00.00 Education and Pedagogical Sciences 50.04.01 Arts and Humanities (degree programme 'Cognitive Studies')
Autistic Spectrum Disorders	37.00.00 Psychological Sciences (degree programmes 'Psychology of Education', 'Social Psychology and Political Psychology', General and Cognitive Psychology', 'Psychology of Physical Culture and Sports') 44.00.00 Education and Pedagogical Sciences
Psycholinguistics	45.00.00 Linguistics and Literary Studies 44.00.00 Education and Pedagogical Sciences 50.04.01 Arts and Humanities (degree programme 'Cognitive Studies')
Neurolinguistics	45.00.00 Linguistics and Literary Studies 44.00.00 Education and Pedagogical Sciences 50.04.01 Arts and Humanities (degree programme 'Cognitive Studies')
Visual Perception and Visual Deception	37.00.00 Psychological Sciences (degree programmes 'Psychology of Physical Culture and Sports', 'Psychology of Education', 'Social Psychology and Political Psychology') 50.04.01 Arts and Humanities (degree programme 'Cognitive Studies')
Political Governance and Public Policy in Russia	41.00.00 Political Sciences and Regional Studies
Sociology of Labour	39.00.00 Sociology and Social Work
Sociology	39.00.00 Sociology and Social Work

Digitalisation of International Relations	41.00.00 Political Sciences and Regional Studies
USA Public Diplomacy	41.00.00 Political Sciences and Regional Studies
Russia and Nuclear Arms Control	41.00.00 Political Sciences and Regional Studies
Legal Regulation of Internet Relations. Russia's Perspective	40.00.00 Law
Constitutional Reforms in Russia	40.00.00 Law
Methodological Issues of Psychology	37.00.00 Psychological Sciences (except degree programme 'Psychology of Development') 44.00.00 Education and Pedagogical Sciences
Mathematical Approach to Psychology Fundamentals of Application'	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences 37.00.00 Psychological Sciences
Introduction to Gender Psychology	39.00.00 Sociology and Social Work
Fundamentals of International Relations Theory	41.00.00 Political Sciences and Regional Studies
Efficient Manager (any course of the specialisation	37.00.00 Psychological Sciences (degree programmes 'Psychology of Physical Culture and Sports', 'Psychology of Education', 'Social Psychology and Political Psychology') 38.00.00 Economics and Management
The Art of Doing Business	38.00.00 Economics and Management
Economy Data Analysis Practices from Simple to Complex	38.00.00 Economics and Management

Digital Transformation start-up versus (any course of the specialisation)	38.00.00 Economics and Management
Digitalisation and Human Resource Management	38.00.00 Economics and Management
Breakthrough Digital Technologies: Cases of Companies	38.00.00 Economics and Management
Features of Doing Business in Arctic Zone of the Russian Federation	38.00.00 Economics and Management
Doing business in Russia	38.00.00 Economics and Management
The Art of Sales	38.00.00 Economics and Management
Principles of Business	38.00.00 Economics and Management
Accounting and Audit Advanced level	38.00.00 Economics and Management
Conflict and Stress Management in an Organisation	38.00.00 Economics and Management
Insurance Business	38.00.00 Economics and Management
Modern financial technologies	38.00.00 Economics and Management
Brand Management in Digital Economy	38.00.00 Economics and Management
Logistics and Procurement Chain Management Today's Challenges	38.00.00 Economics and Management
Start-up: Growing into Successful Business	38.00.00 Economics and Management
Basics of Efficient Business Communication	38.00.00 Economics and Management
The Art of Tax Planning	38.00.00 Economics and Management

Legal Support of Business in Russia (in English)	40.00.00 Law
Computer Games Industry: Principal legal aspects	40.00.00 Law
Big Data in Law: Introduction to the subject matter	40.00.00 Law
Legal Regulation of Corporate Taxation	40.00.00 Law
Tax Law	40.00.00 Law
Criminology	40.00.00 Law
Criminal Procedure	40.00.00 Law
Municipal Law	40.00.00 Law
Theory of Law and State	40.00.00 Law
Foundations of Constitutional Law	40.00.00 Law
Foundations of Innovation and Intellectual Property Management	40.00.00 Law
Intellectual Property Rights in Digital Era	40.00.00 Law
International Labour Law	40.00.00 Law
Modern Legal Regulation of Natural Resource Use	40.00.00 Law
Conflictology	38.00.00 Economics and Management 39.00.00 Sociology and Social Work 41.00.00 Political Sciences and Regional Studies

Segment Analysis	38.00.00 Economics and Management
Econometrics. Advanced Level and Business Analysis	38.00.00 Economics and Management
Econometrics: Time Series Analysis	38.00.00 Economics and Management
Psychology	37.00.00 Psychological Sciences (except degree programmes 'Children's Mental Health and Early Intervention', 'Organisational Psychology and Psychology of Management')
Fundamentals of Project Activities	38.00.00 Economics and Management
Project Development and Implementation	38.00.00 Economics and Management
Countering Financial Fraud and Managing Individual Risks	38.00.00 Economics and Management
Retail Sales Management	38.00.00 Economics and Management
Current Issues in Civil Law	40.00.00 Law
Standing Out in Crowded Market	38.00.00 Economics and Management
On-Demand Supply Chain	38.00.00 Economics and Management
Digital Marketing for Electronic Commerce	38.00.00 Economics and Management
Culture in Dispersed Teams	38.00.00 Economics and Management
Primary care	44.00.00 Education and Pedagogical Sciences
Financial Literacy	38.00.00 Economics and Management
Marketing in VUCA-world	38.00.00 Economics and Management

Data Visualisation	38.00.00 Economics and Management				
Modern Technologies in Insurance	38.00.00 Economics and Management				
	HUMANITIES				
Chinese for beginners	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects				
Chinese for beginners(2)	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects				
Russian as a Foreign Language (any course from the specialisation)	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects				
Living Aspects of Russian Colloquial Speech	42.00.00 Mass Media and Information Librarianship 45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects				
History of Russian Texts	42.00.00 Mass Media and Information Librarianship 45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects				
Modern Language of Journalism	39.00.00 Sociology and Social Work 41.00.00 Political Sciences and Regional Studies 42.00.00 Mass Media and Information Librarianship 45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects				
English Preparing for candidacy exam	45.00.00 Linguistics and Literary Studies				

	51.00.00 Cultural Studies and Sociocultural Projects				
Intercultural Communication and Translations	45.00.00 Linguistics and Literary Studies				
Introduction to Group Theory	51.00.00 Cultural Studies and Sociocultural Projects				
History of English Literature	45.04.01 Philology				
English Lexicology	45.00.00 Linguistics and Literary Studies				
Theoretical Issues in English Grammar	51.00.00 Cultural Studies and Sociocultural Projects				
Cultural Linguistics of Great Britain and the USA	45.04.01 Philology				
Russian Stylistics	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects				
Academic Writing in Russian	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects				
Tasks and Techniques of Linguistic Expert Evaluation	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects				
Official Language in Russia and Abroad	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects				
Language of Modern Advertising and Mass Media	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 42.00.00 Mass Media and Information Librarianship				

Telecommunication Computer Technologies in Advertising and Public Relations	42.00.00 Mass Media and Information Librarianship				
Web Site Promotion	42.00.00 Mass Media and Information Librarianship				
Philosophy	47.00.00 Philosophy, Ethics, and Religious Studies				
Latin Beginner Level	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects				
Islam: History, Culture, and Practices	41.00.00 Political Sciences and Regional Studies 46.00.00 History and Archaeology; 47.00.00 Philosophy, Ethics, and Religious Studies 58.04.00 Asian and African Studies				
Arabic Introductory Course	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.04.00 Asian and African Studies				
Arabic (2)	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.04.00 Asian and African Studies				
Persian for beginners	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.04.00 Asian and African Studies				
Persian for beginners (2)	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.04.00 Asian and African Studies				

Turkish for beginners	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.04.00 Asian and African Studies
Japanese	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.04.00 Asian and African Studies
History and Culture of St Petersburg and Leningrad Oblast	46.00.00 History and Archaeology
History of Russia	46.00.00 History and Archaeology
General History (1)	46.00.00 History and Archaeology
General History (2)	46.00.00 History and Archaeology
Hagiography and Other Sources on the History of the Russian Orthodox Church	46.00.00 History and Archaeology
Introduction to Literary Studies	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects
History of Russian Literature: Russian Literature: Golden Age	45.00.00 Linguistics and Literary Studies
History of Russian Literature: Russian Literature: Silver Age	45.00.00 Linguistics and Literary Studies
History of Russian Literature: 18th Century: Development and Cultural and Historical Context	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects
History of Soviet Literature	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects

Modern Russian Literature	45.00.00 Linguistics and Literary Studies
Russian Émigré Environment	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects
Introduction to Chinese Characters	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.04.00 Asian and African Studies
Korean for Beginners	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.04.00 Asian and African Studies
Logical Aspects of Cognitive Processes	47.00.00 Philosophy, Ethics, and Religious Studies
Cultural Studies	47.00.00 Philosophy, Ethics, and Religious Studies 51.00.00 Cultural Studies and Sociocultural Projects
Chinese for beginners: Phonetic Aspects	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.04.00 Asian and African Studies
Business English	45.00.00 Linguistics and Literary Studies
Business Russian Communication (Any course from the specialisation)	45.00.00 Linguistics and Literary Studies
Ethics	47.00.00 Philosophy, Ethics, and Religious Studies
Religious Studies	47.00.00 Philosophy, Ethics, and Religious Studies

Spanish for beginners (any course from the specialisation)	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects			
Chinese: Five Steps to Success (any course from the specialisation)	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.04.00 Asian and African Studies			
	ART AND CULTURE			
Design Methods: Inspiration Control	54.00.00 Fine and Applied Arts			
Creative Communication: Methods and Tools	54.00.00 Fine and Applied Arts			
Artistic Web Design	54.00.00 Fine and Applied Arts			
Adobe Illustrator for beginners	54.00.00 Fine and Applied Arts			
Interface Design (any course from the specialisation)	54.00.00 Fine and Applied Arts			
Six Steps to Digital Portfolio	54.00.00 Fine and Applied Arts			
Portfolio Design in InDesign	54.00.00 Fine and Applied Arts			
Portfolio Design in Figma and Tilda	54.00.00 Fine and Applied Arts			

Appendix 4

to the Regulations for Admission to Degree Programmes of Higher Education (Bachelor's Degree Programmes, Specialist's Degree Programmes, Master's Degree Programmes) at St Petersburg University in 2022

Correspondence between the disciplines of the All-Russian School Olympiad and the disciplines of admissions tests

Disciplines of the All-Russian School Olympiad	The discipline of admissions tests				
English	Foreign Language				
Astronomy	Physics				
Biology	Biology				
Geography	Geography				
Informatics and ICT	Informatics and ICT				
Art (World Art)	History				
Spanish	Foreign Language				
History	History				
Italian	Foreign Language				
Chinese	Foreign Language				
Literature	Literature				
Mathematics	Mathematics				
German	Foreign Language				
Social Science	Social Science				
Life Safety	_				
Law	Social Science				
Russian	Russian				
Technology	_				
Physics	Physics				
Physical Education	Physical Education				
French	Foreign Language				
Chemistry	Chemistry				
Ecology	Biology, Geography				
Economics	Social Science				

Appendix 5 to the Regulations for Admission to Degree Programmes of Higher Education (Bachelor's Degree Programmes, Specialist's Degree Programmes, Master's Degree Programmes) at St Petersburg University in 2022

Procedures for distributing within (enrolment to) the bachelor's degree programmes admission to which is organised according to the totality

of the degree programmes in the framework of the field of study (a single competitive selection for a group of degree programmes)

Field of study title	Programme title	Type of study	Minimum quantity (state-funded, fee- based)	Maximum quantity (state- funded,)	Minimum quantity (state-funded, fee- based)
1	2	3	4	5	6
03.00.00 PHYSICS AND ASTI	RONOMY — BACHELOR'S DEGREE PROGRAMMES				
03.03.02 Physics	Physics	Full-time	10	65	70
03.03.03 Radiophysics	Electromagnetic and Acoustic Processes	run-unie	10	10	15
40.00.00 LAW — BACHELO	R'S DEGREE PROGRAMMES				
40.03.01 Law	Law (with Advanced Studies in Chinese Language and Law)	- Full-time	10	20	90
40.03.01 Law	Law (with Advanced Studies in Japanese Language and Law)	- Full-time	10	20	30
43.00.00 SERVICES AND TO	URISM — BACHELOR'S DEGREE PROGRAMMES	•	-		
43.03.02 Tourism	The organisation of Tourist Activities (with Advanced Chinese Language Training)	Full-time	10	12	22
43.03.02 Tourism	Tourism		10	5	15
58.00.00 ASIAN AND AFRICA	AN STUDIES — BACHELOR'S DEGREE PROGRAMME	ES .			•
58.03.01 Asian and African Studies	Arabic Philology		5	5	12
58.03.01 Asian and African Studies	Hebrew Studies		5	5	12
58.03.01 Asian and African Studies	Indo-Aryan (Bengali) Philology		6	6	12
58.03.01 Asian and African Studies	Iranian Philology		6	6	12
58.03.01 Asian and African Studies	History of Caucasus (Armenia)		5	5	12

Field of study title	Programme title	Type of study	Minimum quantity (state-funded, fee- based)	Maximum quantity (state- funded,)	Minimum quantity (state-funded, fee- based)
58.03.01 Asian and African Studies	History of China	Full-time	5	5	24
58.03.01 Asian and African Studies	History of Korea		5	5	12
58.03.01 Asian and African Studies	History of Japan		5	5	24
58.03.01 Asian and African Studies	Mongolian and Tibetan Philology		6	6	12
58.03.01 Asian and African Studies	Turkic Philology		5	5	12
58.03.01 Asian and African Studies	Languages and Cultures of East Africa (Hausa Language)		6	6	12
58.03.01 Asian and African Studies	Languages, History and Culture of Ancient Egypt (Egyptology)		6	6	12

- 1. Distribution within degree programmes takes place at the point of the enrolment (by order on enrolment) based on the admission application according to the ranked list of the applicants and taking into account the priority of the degree programmes specified in the admission application complying with the specified limits for the minimum and maximum quantity of students mastering the degree programme as outlined in these Procedures.
- 2. Should a single competitive selection with subsequent distribution within the degree programmes be organised, the admission application must specify the priority of all offered degree programmes (hereinafter referred to as internal priorities) within the single competitive selection for a group of degree programmes. The applicants must rank degree programmes in descending order of priority within the single competitive selection for a group of degree programmes in the annex to their application.
- 3. Should the number of students applying for the enrolment to a particular degree programme (taking into account all applicants, including foreign nationals and stateless citizens, both on government-funded and fee-paid places paid for by individuals and/or legal entities) be less than the minimum quantity of students for the degree programme specified in these Procedures (column 4 'Minimum quantity (state-funded, fee-based)', this degree programme shall not be launched (open). The enrollees shall be distributed within other degree programmes within the corresponding group of the degree programmes complying with the priority of degree programmes specified in the admission application. In this case, the enrollees shall be distributed within the opening degree programmes (degree programmes with the largest quantity of the enrollees applied for the enrolleent) according to the degree programmes priority specified in the admission application. Should there be a demand to increase the maximum quantity of students as specified in column 5 'Maximum quantity (state-funded)' upon the admissions test results, the enrolment shall be carried out regardless of this limit and considering the actual quantity of the enrollees applied for the corresponding degree programmes within the admission quotas set forth by the St Petersburg University Admissions Committee.

4. The limits for the maximum quantity of students in any degree programme (column 6 'Maximum quantity (government-funded, fee-based)' shall be set forth taking into account the admissions of all types of students, including foreign nationals and stateless citizens, and correspond to organisational and technical availabilities of the particular degree programme. At the same time, the enrolment shall be carried out strictly in accordance with the number of places allocated both for admission quotas and for fee-based ones set forth by the decision of the St Petersburg University Admissions Committee.

Appendix 6 to the Regulations for Admission to Degree Programmes of Higher Education (Bachelor's degree programmes, Specialist's degree programmes, Master's degree programmes) at St Petersburg University in 2022

Procedures for distribution within (enrolment to) the Master's degree programmes admission to which is organised according to the totality of the degree programmes in the framework of the field of study (a single competitive selection for a group of degree programmes)

Field of study title	Programme title	Type of study	Minimum quantity (state-funded, fee- based)	Maximum quantity (state-funded)	Minimum quantity (state-funded, fee- based)
01.00.00 MATHEMATICS AND	MECHANICS				
	Mathematical Modelling, Programming, and Artificial Intelligence		10	25	45
	Methods of Applied Mathematics and Computer Science in Control Problems		10	30	45
	Mathematical and Information Support of Economic Activity		10	30	45
01.04.02 Applied Mathematics	Mathematical Approach to Economic Analysis	F. 11 d'ann	10	30	45
and Informatics	Operations Research and Systems Analysis	Full-time	10	30	45
	Applied Mathematics and Computer Science in Problems of Medical Diagnosis		10	30	45
	Applied Information Technology Information Expert Systems		10	30	45
	Mathematical Robotics and Artificial Intelligence		10	40	50
02.00.00 COMPUTER AND INF	ORMATION SCIENCES				
02.04.02 Fundamental	Artificial Intelligence Technologies and Big Data		10	45	50
Informatics and Information Technology	Blockchain Computational Technologies	Full-time	10	15	20
03.00.00 PHYSICS AND ASTRO	NOMY	1	1	1	ı
03/04/2001 Applied Mathematics	Applied Physics and Mathematics	Full-time	10	15	20
and Physics	Engineering Electrophysics	run-ume	10	15	20

	Mathematical and Information Technologies		10	15	20
	Applied Informatics		10	15	20
	Physics		10	65	100
03/04/2002 Physics	Condensed Matter Physics on the Mega-Class Research Facilities	Full-time	5	10	20
05.00.00 EARTH SCIENCES		<u> </u>	-		
	Geology		10	30	35
05 04 2001 Coology	Hydrogeology and Engineering Geology	Full-time	10	10	12
05.04.2001 Geology	Geophysics	run-ume	10	15	18
	Structural Mineralogy and Material Science]	5	5	7
	Urban Geography		10	12	17
05.04.02 Geography	Economic Geography and Digital Spatial Analitics	Full-time	10	13	18
	Hydrosphere and Atmosphere: Modelling and Forecasting		10	15	17
05.04.04 Hydrometeorology	Physical Oceanography and Bioproductivity of Ocean and Seas (PHOBOS)	Full-time	10	10	12
06.00.00 BIOLOGICAL SCIE	NCES				
	Biology		10	80	90
	Molecular and Genetic Fundamentals of Agrobiotechnology		10	12	30
06.04/.001 Biology	Bioinformatics	Full-time	10	12	30
	Cellular and Genetic Technologies		10	12	30
38.00.00 ECONOMICS AND I	MANAGEMENT				
38.04.01 Economics	Business of Russia in the Global Economy and Eurasian Integration	Full-time	10	10	30
38.04.01 Economics	Applied Macroeconomics, Economic Policy and State Regulation	run-ume	10	10	30
	Management of Organisational Development		10	15	30
38.04.01 Economics	Accounting, Analysis and Audit	Full-time	10	15	30
	Corporate Economics and Innovative Activity Economics	1 -	10	15	30
	Financial Economics		10	10	30
38.04.08 Finance and Credit	Risk Management and Insurance	Full-time	10	10	30
	Financial Markets and Banks	†	10	10	30
39.00.00 SOCIOLOGY AND S	SOCIAL WORK	<u> </u>			

	Sociology		10	25	50
39.04.01 Sociology	Sociology in Russia and China	Full-time	10	10	50
0.00.00 LAW 0.04.01 Law 1.00.00 POLITICAL SCIENCE 1.04.04 Political Science	Digital Sociology		10	10	20
40.00.00 LAW					
			10	10	30
40.04.01 Law	International Private Law	Full time	10	10	30
41.00.00 POLITICAL SCIENCE	Real Estate Lawyer	Tun-time	run-time 10	10	30
	Corporate Lawyer		10	10	30
41.00.00 POLITICAL SCIENCE	AND REGIONAL STUDIES				
	Applied Political Science and Ethno-Political Processes in Present-Day World		10	10	15
41.04.04 Political Science	Organisation of Government Relations	Full-time	10	10	15
	Public policy in present-day Russia		10	10	15
1.04.04 Political Science	Political Globalism		10	10	15
41.04.04 Political Science	Political Governance and Public Policy (in English)	Full time	10	4	15
41.04.04 I Officer Science	Business Law International Private Law Real Estate Lawyer 10 10 10 Real Estate Lawyer 10 10 10 TIICAL SCIENCE AND REGIONAL STUDIES Applied Political Science and Ethno-Political Processes in Present-Day World 10 Organisation of Government Relations 10 10 Public policy in present-day Russia 10 10 Political Globalism 10 10 Political Governance and Public Policy (in English) 10 Russia and China in the global politics (in English) 10 Russia and China in the global politics (in English) 10 American Studies 10 10 Baltic and Nordic Studies 10 10 International Relations 10 10 Diplomacy of the Russian Federation and Foreign States 10 Theory of International Relations and Foreign Policy Analysis International Humanitarian Cooperation Full-time 10 Public Relations as Part of International Relations 10	10	4	15	
				9	20
	BRICS Studies		10	10	20
41 04 05 Intermedianal Paletiana	European Studies	Fig.11 4ima	10 10 10 10 10 10 10 10 10 10 10 10 10 1	10	20
41.04.03 International Relations	International Relations in the Post-Soviet Countries	run-ume	10	9	20
41.04.05 International Relations	Baltic and Nordic Studies		10	10	20
	International Private Law Real Estate Lawyer Corporate Lawyer 10 10 10 10 10 10 10 1	9	20		
	Diplomacy of the Russian Federation and Foreign States		10	10	20
			10	10	20
41 04 05 International Relations	International Humanitarian Cooperation	Full-time	10	10	20
	Public Relations as Part of International Relations	Tun-unic	10	10	20
	World Politics		10	10	20
	Artificial Intelligence and international security		10	10	20

41.04.05 International Relations	Strategic Research	Full-time	10	8	20
41.04.03 International Relations	International Relations (in English).	ruii-time	10	8	20
42.00.00 MASS MEDIA AND II	NFORMATION LIBRARIANSHIP				
	Corporate Communications and Advertising		10	10	40
42.04.01 Advertising and Public	Media Communications	Full-time	10	10	40
Relations	Managing Communication projects in Politics and Business		10	8	40
	International Journalism		10	10	40
	Authorial Audio Visual Project		10	10	40
	Research Journalism		10	10	40
42.04.02 Journalism	Popular Science Journalism	Full-time	10	10	40
	Media Texts in Mass Media		10	10	40
	Visual Communications in Journalism and Advertising		10	10	40
	Journalism and Culture of Society		10	10	40
45.00.00 LINGUISTICS AND L	ITERARY STUDIES	1			
	Byzantine and Modern Greek Philology		5	10	12
45.04.01 Philology	Classical Philology and Classical Antiquity Tradition in World Culture	Full-time	5	10	12
45.04.04.7N.11.1	Foreign Literatures and Cultures	77.11.1	5	20	25
45.04.01 Philology	Slavonic Languages and Literatures	Full-time	5	10	14
	Intercultural Communication: German-Russian Dialogue / Interkulturelle Kommunikation als DACH-RU-Kulturdialog		5	7	12
45.04.02 Linguistics	Deutsch als Fremdsprache als Kulturdialog / German as Foreign Language and Intercultural Dialogue	Full-time	5	6	12
	Translation Theory and Cross-Linguistic Communication (German)		5	6	12
45.04.00 Time 1.41m	Linguistics of English Socio-Political Discourse	E 11 C	5	15	20
45.04.02 Linguistics	Cultural Linguistics of Great Britain and the USA	Full-time	5	15	20
	English Discourse and Variability		5	10 10 10 10 10 10 10 10 10 10 10 10 10 1	20
	Literary Translation		5	15	20
45.04.02 Linguistics	Legal Translation	Full-time	5		20
	Simultaneous Translation (English)		5		20
	Russian		5	15	20
45.04.02 Linguistics	Russian Language and Russian Culture in the Aspect of Teaching Russian as Foreign Language	Full-time	5	20	25

	Legal Linguistics		5	15	20
	Theory and History of Language and European Languages		5	20	25
45.04.02 I in a living	Romance Languages	F 11 (****	5	10	15
45.04.02 Linguistics	Linguistic Issues in Scandinavian and Dutch Studies	- Full-time	5	10	15
	Foreign Languages and Intercultural Communication in Business and Management		5	10	20
	Foreign Languages in International Relations		5	10	20
45.04.02 Linguistics	Intercultural Communication and Translations in Petroleum Business	Full-time	5	10	20
	Foreign Languages and Intercultural Communication in Tourism and Sightseeing Business		5	10	20
46.00.00 HISTORY AND ARC	CHAEOLOGY				
	History and Culture of St Petersburg and Leningrad Oblast		10	13	20
	History and Theory of Nations and the Issues of Nationalism	10 10	13	20	
	The Origin of the Modern Western Civilisation		10	13	20
46.04.01 History	Historical and Cultural Sightseeing Business	Full-time	10	13	20
	History of the Russian Civilization	1	10	13	20
	Modernisation of the West: 16th-21st Centuries	1	10	13	20
	Russia in Wars and Social Conflicts]	10	13	20
51.00.00 CULTURAL STUDII	ES AND SOCIOCULTURAL PROJECTS				
51.04.01 Cultural Studies	Axiology of Russian Culture	- Full-time	12	12	20
31.04.01 Cultural Studies	Digital Society Culture	- run-ume	12	12	20
51.04.04 Museology and Conservation of Cultural and	Visual Technology in Museums	Full-time	9	9	20
Natural Heritage Sites	Supervisory Control at Museums	Tun-time	9	9	20
58.00.00 ASIAN AND AFRICA	AN STUDIES				
	Asian and African History (with Additional Asian and African Languages Training)		5	8	12
58.04.01 Asian and African Studies	Asian and African History (with Additional Asian and African Languages Training)	Full-time	5	8	12
	Asian and African Literatures (with Additional Asian and African Languages Training)		5	10 10 11 13 13 13 13 13 13 13 13 12 12 12 9 9	12

Asian and African Politics and International Relations (with Additional Asian and African Languages Training)		5	8	12
Asian and African Religions (with Additional Asian and African Languages Training)		5	8	12
Asian and African Social and Cultural Anthropology (with Additional Asian and African Languages Training)		5	8	12
Asian and African Economics and International Economic Relations International Relations (with Additional Asian and African Languages Training)		5	8	12
Asian and African Languages	-	5	8	12

- 1. Distribution within degree programmes shall be carried out at point of the enrolment (by order on enrolment) based on the admission application according to the ranked list of the applicants and taking into account the priority of degree programmes specified in the admission application observing the limits for the minimum and maximum quantity of students mastering the degree programme as outlined in these Procedures.
- 2. Should a single competitive selection with subsequent distribution within the degree programmes be organised, the admission application must specify the priority of all offered degree programmes (hereinafter referred to as internal priorities) within the single competitive selection for a group of degree programmes
- 3. Should the quantity of students applying for the enrolment to a particular degree programme (taking into account all applicants, including foreign nationals and stateless citizens, both on government-funded and fee-paid places paid for by individuals and/or legal entities) be less than the minimum quantity of students for the degree programme specified in these Procedures (column 4 'Minimum quantity (state-funded, fee-based)', this degree programme shall not be launched (open). The enrollees shall be distributed within other degree programmes within the corresponding group of the degree programmes complying with the priority of degree programmes specified in the admission application. In this case, the enrollees shall be distributed within the opening degree programmes (degree programmes with the largest quantity of the enrollees applied for the enrollement) according to the degree programmes priority specified in the admission application. Should there be a demand to increase the maximum quantity of students as specified in column 5 'Maximum quantity (state-funded)' upon the admissions test results, the enrolment shall be carried out regardless this limit and considering the actual quantity of the enrollees applied for the corresponding degree programmes within the admission quotas set forth by the St Petersburg University Admissions Committee
- 4. The limits for maximum quantity of students in any degree programme (column 6 'Maximum quantity (state-funded, fee-based)' shall be set forth taking into account the admissions of all types of students, including foreign nationals and stateless persons, and correspond to organisational and technical availabilities of the particular degree programme. At the same time, the enrolment shall be carried out strictly in accordance with the quantity of places allocated both for admission quotas and for fee-based ones set forth by the decision of the St Petersburg University Admissions Committee.

Appendix 7
to the Regulations for Admission
to Degree Programmes of Higher Education
(Bachelor's degree's Degree Programmes, Specialist's degree's Degree Programmes,
Master's Degree Programmes) at St Petersburg University in 2022

Correspondence between the Centralised Test Disciplines and the General Education Admission Test Disciplines

Disciplines of the Centralised Test	Disciplines of the General Education Admissions test
Russian	Russian
Belarusian	Not applicable
Physics	Physics
Mathematics	Mathematics
Chemistry	Chemistry
Biology	Biology
Foreign language (English, German,	Foreign Language
French, Spanish, Chinese)	
History of Belarus	Not applicable
Social Studies	Social Science
Geography	Geography
General History	Not applicable

Appendix 8

To the Regulations for Admission
to Degree Programmes of Higher Education
(Bachelor's degree's Degree Programmes, Specialist's degree's Degree Programmes,
Master's Degree Programmes) at St Petersburg University in 2022

Correspondence between the All-Russia School Olympiad Profiles or the International Olympiad in General Education Subjects and Disciplines Within Fields of Study and Specialities Open for Admission at St Petersburg University

37.03.02 Conflictology Bachelor's degree	Disciplines of the All- Russian School Olympiad	Titles of the specialised fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled on these fields of study and specialities)	Level of education
39.03.01 Sociology 39.03.02 Social Work 39.03.02 Social Work 40.03.01 Law 40.03.01 Law 40.03.05 International Relations 43.03.02 Tourism 45.03.02 Tourism 45.03.01 Philology 45.03.02 Linguistics (degree programmes 'Albanian', 'English', Danish', 'Lithuanian', 'Interlanguage Communication and Translation (English)', 'General and Applied Phonetics', 'Applied Computer and Computational Linguistics (English)', 'Russian as a Foreign Language', 'Comparative Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theory and Practice of Intercultural Communication (English)'. 47.03.01 Philosophy 47.03.03 Religious Studies 47.03.03 Religious Studies 50.03.01 Arts and Humanities Bachelor's degree 50.03.03 History of Arts 48.02 Bachelor's degree 51.03.01 Cultural Studies Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree Bachelor's degree		37.03.02 Conflictology	
Bachelor's degree 40.03.01 Law Bachelor's degree 41.03.05 International Relations Bachelor's degree 43.03.02 Tourism Bachelor's degree 45.03.01 Philology Bachelor's degree 45.03.02 Linguistics (degree programmes 'Albanian', 'English', Danish', 'Lithuanian', 'Interlanguage Communication and Translation (English)', 'General and Applied Phonetics', 'Applied Computer and Computational Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theory and Practice of Intercultural Communication (English)'. 47.03.01 Philosophy			
39.03.02 Social Work 40.03.01 Law Bachelor's degree 41.03.05 International Relations 43.03.02 Tourism Bachelor's degree 45.03.01 Philology 45.03.02 Linguistics (degree programmes 'Albanian', 'English', Danish', 'Cithuanian', 'Interlanguage Communication and Translation (English)', 'General and Applied Phonetics', 'Applied Computer and Computational Linguistics (English)', 'Russian as a Foreign Language', 'Comparative Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theory and Practice of Intercultural Communication (English)'. 47.03.01 Philosophy 47.03.03 Religious Studies 50.03.01 Arts and Humanities Bachelor's degree 50.03.03 History of Arts 50.03.03 History of Arts Bachelor's degree Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree Bachelor's degree		39.03.01 Sociology	
degree Bachelor's degree H. (103.05 International Relations Bachelor's degree Bachelor's degree H. (103.05 International Relations Bachelor's degree H. (103.05 Inguistics (degree programmes 'Albanian', 'English', Danish', 'Lithuanian', 'Interlanguage Communication and Translation (English)', 'General and Applied Phonetics', 'Applied Computer and Computational Linguistics (English)', 'Russian as a Foreign Language', 'Comparative Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theory and Practice of Intercultural Communication (English)'. 47.03.01 Philosophy			
40.03.01 Law Bachelor's degree 41.03.05 International Relations Bachelor's degree 43.03.02 Tourism Bachelor's degree 45.03.01 Philology Bachelor's degree 45.03.02 Linguistics (degree programmes 'Albanian', 'English', Danish', degree 45.03.02 Linguistics (degree programmes 'Albanian', 'English', 'General and Applied Phonetics', 'Applied Computer and Computational Linguistics (English)', 'Russian as a Foreign Language', 'Comparative Linguistics (English)', 'Russian as a Foreign Language', 'Comparative Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theory and Practice of Intercultural Communication (English)'. 47.03.01 Philosophy Bachelor's degree 47.03.03 Religious Studies Bachelor's degree 50.03.01 Arts and Humanities Bachelor's degree 50.03.03 History of Arts Bachelor's degree 51.03.01 Cultural Studies Bachelor's degree Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree		39.03.02 Social Work	
### degree 41.03.05 International Relations Bachelor's degree 43.03.02 Tourism Bachelor's degree 45.03.01 Philology Bachelor's degree 45.03.02 Linguistics (degree programmes 'Albanian', 'English', Danish', 'Lithuanian', 'Interlanguage Communication and Translation (English)', 'General and Applied Phonetics', 'Applied Computer and Computational Linguistics (English)', 'Russian as a Foreign Language', 'Comparative Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theory and Practice of Intercultural Communication (English)'. #### 47.03.01 Philosophy Bachelor's degree 47.03.03 Religious Studies Bachelor's degree 50.03.01 Arts and Humanities Bachelor's degree 50.03.03 History of Arts Bachelor's degree 51.03.01 Cultural Studies Bachelor's degree 51.03.01 Cultural Studies Bachelor's degree 58.03.01 Asian and African Studies 59.03.01 Asian and African Studie			
### Achelor's degree		40.03.01 Law	
### According to the content of the		44.00.07	
45.03.01 Philology 45.03.02 Linguistics (degree programmes 'Albanian', 'English', Danish', 'Lithuanian', 'Interlanguage Communication and Translation (English)', 'General and Applied Phonetics', 'Applied Computer and Computational Linguistics (English)', 'Russian as a Foreign Language', 'Comparative Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theory and Practice of Intercultural Communication (English)'. 47.03.01 Philosophy 47.03.03 Religious Studies 50.03.01 Arts and Humanities 50.03.03 History of Arts 50.03.03 History of Arts 62gree 51.03.01 Cultural Studies Museology and Conservation of Cultural and Natural Heritage Sites 63egree 58.03.01 Asian and African Studies 63egree 58.03.01 Asian and African Studies		41.03.05 International Relations	
### Language		42.02.02 FL	
### A5.03.01 Philology ### A5.03.02 Linguistics (degree programmes 'Albanian', 'English', Danish', 'Cithuanian', 'Interlanguage Communication and Translation (English)', 'General and Applied Phonetics', 'Applied Computer and Computational Linguistics (English)', 'Russian as a Foreign Language', 'Comparative Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theory and Practice of Intercultural Communication (English)'. ###################################		43.03.02 Tourism	
English Eng		45 02 01 DL:1-1	
English Eng		45.05.01 Philology	
*Lithuanian', 'Interlanguage Communication and Translation (English)', 'General and Applied Phonetics', 'Applied Computer and Computational Linguistics (English)', 'Russian as a Foreign Language', 'Comparative Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theory and Practice of Intercultural Communication (English)'. 47.03.01 Philosophy 47.03.03 Religious Studies Bachelor's degree 50.03.01 Arts and Humanities Bachelor's degree 50.03.03 History of Arts Bachelor's degree 51.03.01 Cultural Studies Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree		45.02.02 Linguistics (doggo programmes 'Albanian', 'English', Danish'	
and Applied Phonetics', 'Applied Computer and Computational Linguistics (English)', 'Russian as a Foreign Language', 'Comparative Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theory and Practice of Intercultural Communication (English)'. 47.03.01 Philosophy 47.03.03 Religious Studies Bachelor's degree 50.03.01 Arts and Humanities 50.03.03 History of Arts Bachelor's degree 51.03.01 Cultural Studies Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree 58.03.01 Asian and African Studies			
English (English)', 'Russian as a Foreign Language', 'Comparative Linguistics (English)', 'Theoretical and Experimental Linguistics (English)', 'Theory and Practice of Intercultural Communication (English)'. 47.03.01 Philosophy 47.03.03 Religious Studies 47.03.03 Religious Studies 50.03.01 Arts and Humanities 50.03.01 Arts and Humanities 50.03.03 History of Arts 62gree 51.03.01 Cultural Studies 83 Chelor's degree 51.03.01 Cultural Studies 84 Museology and Conservation of Cultural and Natural Heritage Sites 65 Bachelor's degree 58.03.01 Asian and African Studies 85 Bachelor's degree 58.03.01 Asian and African Studies			degree
English)', 'Theoretical and Experimental Linguistics (English)', 'Theory and Practice of Intercultural Communication (English)'. 47.03.01 Philosophy 47.03.03 Religious Studies 47.03.03 Religious Studies 50.03.01 Arts and Humanities 50.03.01 Arts and Humanities 50.03.03 History of Arts 51.03.01 Cultural Studies Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree Museology and African Studies 58.03.01 Asian and African Studies Bachelor's degree 58.03.01 Asian and African Studies			
Practice of Intercultural Communication (English)'. 47.03.01 Philosophy 47.03.03 Religious Studies Bachelor's degree 50.03.01 Arts and Humanities Bachelor's degree 50.03.03 History of Arts Bachelor's degree 51.03.01 Cultural Studies Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree Bachelor's degree	English		
47.03.01 Philosophy Bachelor's degree 47.03.03 Religious Studies Bachelor's degree 50.03.01 Arts and Humanities Bachelor's degree 50.03.03 History of Arts Bachelor's degree 51.03.01 Cultural Studies Bachelor's degree Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree			
degree 47.03.03 Religious Studies Bachelor's degree 50.03.01 Arts and Humanities Bachelor's degree 50.03.03 History of Arts Bachelor's degree 51.03.01 Cultural Studies Bachelor's degree Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree			
47.03.03 Religious Studies Bachelor's degree 50.03.01 Arts and Humanities Bachelor's degree 50.03.03 History of Arts Bachelor's degree 51.03.01 Cultural Studies Bachelor's degree Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree		47.03.01 Philosophy	
degree 50.03.01 Arts and Humanities Bachelor's degree 50.03.03 History of Arts Bachelor's degree 51.03.01 Cultural Studies Bachelor's degree Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree			
50.03.01 Arts and Humanities Bachelor's degree 50.03.03 History of Arts Bachelor's degree 51.03.01 Cultural Studies Bachelor's degree Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree		47.03.03 Religious Studies	
degree 50.03.03 History of Arts Bachelor's degree 51.03.01 Cultural Studies Bachelor's degree Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree			
50.03.03 History of Arts Bachelor's degree 51.03.01 Cultural Studies Bachelor's degree Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree		50.03.01 Arts and Humanities	
degree 51.03.01 Cultural Studies Bachelor's degree Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree			
51.03.01 Cultural Studies Bachelor's degree Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree		50.03.03 History of Arts	
Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree		74 00 04 G 1: 1 G : 1	
Museology and Conservation of Cultural and Natural Heritage Sites Bachelor's degree 58.03.01 Asian and African Studies Bachelor's degree		51.03.01 Cultural Studies	
58.03.01 Asian and African Studies Bachelor's degree		M 1 10 10 11 1N 11 1N 11 11 11 11 11 11 11 11 11	
58.03.01 Asian and African Studies Bachelor's degree		Museology and Conservation of Cultural and Natural Heritage Sites	
degree		50 02 01 Agien and African Studies	
		JO.UJ.U1 ASIAH AHI CAH Studies	
U1.04.2001 Wathematics Bachelor's		01.04.2001 Methamatics	
degree		01.04.2001 Wathematics	
02.03.01 Mathematics and Computer Sciences (programme 'Data Sciences') Bachelor's		02.03.01 Mathematics and Computer Sciences (programma 'Data Sciences')	
Astronomy O2.05.01 Mathematics and Computer Sciences (programme Data Sciences) degree	Astronomy	02.03.01 Mathematics and Computer Sciences (programme Data Sciences)	
Applied Mathematics and Physics (programme: Applied Physics and Bachelor's		Applied Mathematics and Physics (programme: Applied Physics and	
Mathematics) Applied Mathematics and Physics (programme: Applied Physics and Bachelor's degree			

	03.03.02 Physics	Bachelor's
	03.03.03 Radiophysics	degree Bachelor's
	03.05.01 Astronomy	degree Specialist's
	06.04/.001 Biology	degree Bachelor's
	06.04.02 Soil Science	degree Bachelor's
	31.05.01 General Medicine	degree
		Specialist's degree
Biology	31.05.03 Dental Medicine	Specialist's degree
	37.03.01 Psychology	Bachelor's degree
	37.05.01 Clinical Psychology	Specialist's degree
	37.05.02 Employment Psychology	Specialist's degree
	05.04.2001 Geology	Bachelor's
	05.03.02 Geography	degree Bachelor's
	05.03.03 Cartography and Geoinformatics	degree Bachelor's
	05.03.04 Hydrometeorology	degree Bachelor's
	05.03.06 Ecology and Nature Management	degree Bachelor's
		degree Bachelor's
	21.03.01 Petroleum Engineering	degree
	21.03.02 Land Use and Cadastres	Bachelor's degree
	37.03.02 Conflictology	Bachelor's degree
Geography	41.03.02 Russian Region Studies	Bachelor's degree
	41.03.05 International Relations	Bachelor's degree
	43.03.02 Tourism	Bachelor's
	46.03.01 History	degree Bachelor's
	47.03.01 Philosophy	degree Bachelor's
Informatics	47.03.03 Religious Studies	degree Bachelor's
	51.03.01 Cultural Studies	degree Bachelor's
		degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's degree
	01.04.2001 Mathematics	Bachelor's degree
	Applied Mathematics and Informatics (programme: 'Applied Mathematics', 'Fundamental Informatics and Programming', 'Modern Programming')	Bachelor's degree
VA MARKEN	02.03.01 Mathematics and Computer Sciences (programme 'Data Sciences')	Bachelor's degree
	02.03.02 Fundamental Informatics and Information Technology	Bachelor's degree

	02.03.03 Software and Administration of Information Systems	Bachelor's degree
	05.04.2001 Geology	Bachelor's
	05.03.02 Geography	degree Bachelor's
		degree
	05.03.03 Cartography and Geoinformatics	Bachelor's
	05.03.04 Hydrometeorology	degree Bachelor's
	03.03.04 Hydrometeorology	degree
	05.03.06 Ecology and Nature Management	Bachelor's
		degree
	06.04.02 Soil Science	Bachelor's
		degree
	Applied Informatics	Bachelor's
	00 02 04 Software Engineering	degree Bachelor's
	09.03.04 Software Engineering	degree
	21.03.01 Petroleum Engineering	Bachelor's
		degree
	21.03.02 Land Use and Cadastres	Bachelor's
		degree
	27.03.03 Systems Analysis and Control	Bachelor's
	27 02 02 G - G' - 1	degree
	37.03.02 Conflictology	Bachelor's
	38.03.01 Economics	degree Bachelor's
	56.03.01 Economics	degree
	38.03.03 Personnel Management	Bachelor's
		degree
	38.03.05 Business Informatics	Bachelor's
		degree
	39.03.01 Sociology	Bachelor's
	47.03.01 Philosophy	degree Bachelor's
	47.03.01 Filliosophy	degree
	47.03.03 Religious Studies	Bachelor's
	<i>θ</i>	degree
	50.03.01 Arts and Humanities	Bachelor's
		degree
	51.03.01 Cultural Studies	Bachelor's
	Museology and Conservation of Cultural and Natural II-ita Cita-	degree Bachelor's
	Museology and Conservation of Cultural and Natural Heritage Sites	degree
	37.03.02 Conflictology	Bachelor's
		degree
	39.03.02 Social Work	Bachelor's
		degree
	41.03.02 Russian Region Studies	Bachelor's
	46.03.01 History	degree Bachelor's
	40.03.01 HISTORY	degree
Art (World Art)	47.03.01 Philosophy	Bachelor's
		degree
	47.03.03 Religious Studies	Bachelor's
		degree
	50.03.01 Arts and Humanities	Bachelor's
	50.02.02 History of Arts	degree Bachelor's
	50.03.03 History of Arts	degree
	1	uegree

	51.03.01 Cultural Studies	Bachelor's
		degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's
	Transcology and Conservation of California Indicates Transcology and Conservation of California	degree
	37.03.02 Conflictology	Bachelor's
		degree
	39.03.01 Sociology	Bachelor's
		degree
	39.03.02 Social Work	Bachelor's
		degree
	40.03.01 Law	Bachelor's
		degree
	41.03.05 International Relations	Bachelor's
		degree
	43.03.02 Tourism	Bachelor's
		degree
	45.03.01 Philology (programmes 'Biblical Philology (Languages of the Bible and	Bachelor's
	Biblical Texts' 'Classical Philology (Old Greek and Latin; Ancient Literature', '	degree
	Russian Philology (Russian Language and Literature)')	
Spanish	45.03.02 Linguistics (programmes 'Albanian', 'Danish', 'Spanish', '	Bachelor's
Spainsn	Lithuanian', 'General and Applied Phonetics', 'Russian as a Foreign Language')	
		degree
	47.03.01 Philosophy	Bachelor's
		degree
	47.03.03 Religious Studies	Bachelor's
		degree
	50.03.01 Arts and Humanities	Bachelor's
		degree
	50.03.03 History of Arts	Bachelor's
		degree
	51.03.01 Cultural Studies	Bachelor's
		degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's
		degree
	58.03.01 Asian and African Studies	Bachelor's
		degree
	37.03.02 Conflictology	Bachelor's
		degree
	39.03.01 Sociology	Bachelor's
		degree
	39.03.02 Social Work	Bachelor's
		degree
	40.03.01 Law	Bachelor's
		degree
	41.03.02 Russian Region Studies	Bachelor's
		degree
	41.03.04 Political Science	Bachelor's
History		degree
History	41.03.05 International Relations	Bachelor's
		degree
	42.03.01 Advertising and Public Relations	Bachelor's
		degree
	43.03.02 Tourism	Bachelor's
		degree
	46.03.01 History	Bachelor's
		degree
	47.03.01 Philosophy	Bachelor's
		degree
	47.03.03 Religious Studies	Bachelor's
		degree

	50.03.01 Arts and Humanities	Bachelor's
		degree
	50.03.03 History of Arts	Bachelor's
		degree
	51.03.01 Cultural Studies	Bachelor's
		degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's
	wideology and Conservation of Cultural and Ivalural Heritage Sites	
		degree
	58.03.01 Asian and African Studies	Bachelor's
		degree
	37.03.02 Conflictology	Bachelor's
		degree
	39.03.01 Sociology	Bachelor's
	bytociol bottotogy	degree
	39.03.02 Social Work	Bachelor's
	39.03.02 SOCIAI WOIK	
		degree
	40.03.01 Law	Bachelor's
		degree
	41.03.05 International Relations	Bachelor's
		degree
	43.03.02 Tourism	Bachelor's
	43.03.02 Tourism	degree
	45 02 01 PH 11 1 / (P.11' 1 PH 11 1 / C.4 P.11 1	
	45.03.01 Philology (programmes 'Biblical Philology (Languages of the Bible and	Bachelor's
	Biblical Texts' 'Classical Philology (Old Greek and Latin; Ancient Literature', '	degree
	Russian Philology (Russian Language and Literature)')	
Italian	45.03.02 Linguistics (programmes 'Albanian', 'Danish', 'Spanish', '	Bachelor's
Italiali	Lithuanian', 'General and Applied Phonetics', 'Russian as a Foreign Language')	
		degree
	47.03.01 Philosophy	Bachelor's
		degree
	47.03.03 Religious Studies	Bachelor's
	Tributou Italigious Studies	degree
	50.03.01 Arts and Humanities	Bachelor's
	50.05.01 Arts and numanities	
		degree
	50.03.03 History of Arts	Bachelor's
		degree
	51.03.01 Cultural Studies	Bachelor's
		degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's
	Trascology and Conservation of Calcular and Transaction Sites	degree
	58.03.01 Asian and African Studies	Bachelor's
	58.03.01 Asian and African Studies	
		degree
	37.03.02 Conflictology	Bachelor's
		degree
	39.03.01 Sociology	Bachelor's
		degree
	39.03.02 Social Work	Bachelor's
	57.05.02 DOCIGI WOLK	
	40.02.01 L	degree
Chinese	40.03.01 Law	Bachelor's
		degree
	41.03.05 International Relations	Bachelor's
		degree
	42.03.01 Advertising and Public Relations	Bachelor's
	3	degree
	43.03.02 Tourism	Bachelor's
	TJ.0J.04 100H5H	
	45 00 01 DU 1 1 / (DU 1 1 DU 1 1 /)	degree
	45.03.01 Philology (programmes 'Biblical Philology (Languages of the Bible and	Bachelor's
	Biblical Texts' 'Classical Philology (Old Greek and Latin; Ancient Literature', '	degree
	Russian Philology (Russian Language and Literature)')	
		Bachelor's
	47.03.01 Philosophy	Dachelor s
		degree

	47.03.03 Religious Studies	Bachelor's
	Thousand Tenglous Studies	degree
	50.03.01 Arts and Humanities	Bachelor's
	50.05.01 Arts and Hamandes	degree
	50.03.03 History of Arts	Bachelor's
	50.05.05 History of Arts	degree
	51.03.01 Cultural Studies	Bachelor's
	31.03.01 Cultural Studies	
	V 1 10 10 1 10 1 1 1 1 1 1 1 1 1 1 1 1 1	degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's
		degree
	58.03.01 Asian and African Studies	Bachelor's
		degree
	37.03.02 Conflictology	Bachelor's
		degree
	42.03.02 Journalism	Bachelor's
		degree
	45.03.01 Philology	Bachelor's
		degree
	47.03.01 Philosophy	Bachelor's
	111001011 missoping	degree
	47.03.03 Religious Studies	Bachelor's
Literature	47.03.03 Religious studies	
	50.03.01 Arts and Humanities	degree Bachelor's
	50.05.01 Arts and Humanities	
	70 00 00 XX	degree
	50.03.03 History of Arts	Bachelor's
		degree
	51.03.01 Cultural Studies	Bachelor's
		degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's
		degree
	01.04.2001 Mathematics	Bachelor's
		degree
	01/03/2002 Applied Mathematics and Informatics	Bachelor's
	11	degree
	01.03.03 Mechanics and Mathematical Modelling	Bachelor's
	o Trouted Transmitted and Transmitted Transmitted	degree
	01.05.01 Fundamental Mathematics and Mechanics	Specialist's
	01.03.01 I undamental Wathematics and Wicchames	1 1
	02.03.01 Mathematics and Computer Sciences	Bachelor's
	02.03.01 Mathematics and Computer Sciences	
	02.02.01 A . 1' . 1 M . 4 1 D1 '	degree
	03.03.01 Applied Mathematics and Physics	Bachelor's
	02 02 02 PM	degree
	03.03.02 Physics	Bachelor's
		degree
Mathematics	03.03.03 Radiophysics	Bachelor's
1VIutileinuties		degree
	04.03.02 Chemistry, Physics and Material Mechanics	Bachelor's
		degree
	05.04.2001 Geology	Bachelor's
		degree
	05.03.02 Geography	Bachelor's
		degree
	05.03.03 Cartography and Geoinformatics	Bachelor's
		degree
	05.03.04 Hydrometeorology	Bachelor's
	35.55.6 Filipatomotology	degree
	05.03.06 Ecology and Nature Management	Bachelor's
	03.03.00 Ecology and Ivalure Ivianagement	degree
	06.04.02 Soil Science	Bachelor's
	00.04.02 SOII SCIENCE	
		degree

	Applied Informatics	Bachelor's
		degree
	21.03.01 Petroleum Engineering	Bachelor's
		degree
	21/03/2002 Land Use and Cadastres	Bachelor's
		degree
	27.03.03 Systems Analysis and Control	Bachelor's
		degree
	37.03.01 Psychology	Bachelor's
	27.02.02 Canflists Is an	degree Bachelor's
	37.03.02 Conflictology	degree
	37.05.01 Clinical Psychology	Specialist's degree
	37.05.02 Employment Psychology	Specialist's
	0 · · · · · · · · · · · · · · · · · · ·	degree
	38.03.01 Economics	Bachelor's
		degree
	38.03.02 Management	Bachelor's
		degree
	38.03.03 Personnel Management	Bachelor's
		degree
	38.03.04 Public and Municipal Administration	Bachelor's
		degree
	38.03.05 Business Informatics	Bachelor's
		degree
	39.03.01 Sociology	Bachelor's
		degree
	42.03.01 Advertising and Public Relations	Bachelor's
	1-	degree
	47.03.01 Philosophy	Bachelor's
	17705.011 infosoping	degree
	47.03.03 Religious Studies	Bachelor's
	17700100 Tiongious Studies	degree
	50.03.01 Arts and Humanities	Bachelor's
		degree
	51.03.01 Cultural Studies	Bachelor's
		degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's
		degree
	37.03.02 Conflictology	Bachelor's
		degree
	39.03.01 Sociology	Bachelor's
		degree
	39.03.02 Social Work	Bachelor's
	STORING BOOM I TOM	degree
	40.03.01 Law	Bachelor's
		degree
	41.03.05 International Relations	Bachelor's
		degree
German	43.03.02 Tourism	Bachelor's
		degree
	45.03.01 Philology	Bachelor's
		degree
	45.03.02 Linguistics)programmes 'Albanian', 'Danish', 'Foreign Language (Polish or German)', 'Lithuanian', 'Interlanguage Communication and Translation (German)', 'German', 'General and Applied Phonetics', 'Russian as	Bachelor's degree
	a Foreign Language')	
	47.03.01 Philosophy	Bachelor's
		degree

	47.03.03 Religious Studies	Bachelor's
		degree
	50.03.01 Arts and Humanities	Bachelor's
		degree
	50.03.03 History of Arts	Bachelor's
	50.05.05 Thistory of This	degree
	51.03.01 Cultural Studies	Bachelor's
	51.03.01 Cultural Studies	
		degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's
		degree
	58.03.01 Asian and African Studies	Bachelor's
	50.05.01 Fisher and Firsten Studies	degree
	27.02.02.C C	Bachelor's
	37.03.02 Conflictology	
		degree
	38.03.01 Economics	Bachelor's
		degree
	38.03.03 Personnel Management	Bachelor's
	50.05.05 Telsomici Munigement	degree
	38.03.05 Business Informatics	Bachelor's
	38.03.03 Business informatics	
		degree
	39.03.01 Sociology	Bachelor's
		degree
	39.03.02 Social Work	Bachelor's
	57.05.02 Booldi Work	degree
	40.00.01 X	
	40.03.01 Law	Bachelor's
		degree
	41.03.02 Russian Region Studies	Bachelor's
		degree
	41.03.04 Political Science	Bachelor's
	41.03.04 i olideal science	degree
Social Science	41.00.051	
	41.03.05 International Relations	Bachelor's
		degree
	42.03.01 Advertising and Public Relations	Bachelor's
		degree
	43.03.02 Tourism	Bachelor's
	43.03.02 Tourism	
	45.00.04 ***	degree
	46.03.01 History	Bachelor's
		degree
	47.03.01 Philosophy	Bachelor's
		degree
	47.03.03 Religious Studies	Bachelor's
	47.03.03 Religious Studies	
		degree
	50.03.01 Arts and Humanities	Bachelor's
		degree
	51.03.01 Cultural Studies	Bachelor's
		degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's
	17105010gy and Consol varion of Cultural and Natural Heritage Sites	
T10 C C :	20.02.02.0	degree
Life Safety	39.03.02 Social Work	Bachelor's
		degree
	37.03.02 Conflictology	Bachelor's
		degree
	39.03.02 Social Work	Bachelor's
	57,051,02 DOUM IT OTA	degree
	40.02.01 I	
Law	40.03.01 Law	Bachelor's
24,11		degree
	41.03.02 Russian Region Studies	Bachelor's
		degree
	41.03.05 International Relations	Bachelor's
	11.05.05 International Actations	
		degree

	46.03.01 History	Bachelor's
	40.03.01 History	degree
	47.03.01 Philosophy	Bachelor's
	47.03.01 1 miosophy	degree
	47.03.03 Religious Studies	Bachelor's
	47.03.03 Kengious Studies	degree
	51.03.01 Cultural Studies	Bachelor's
	31.03.01 Cultural Studies	degree
	Muse along and Company tion of Cultural and Noticeal Heritage Cites	Bachelor's
	Museology and Conservation of Cultural and Natural Heritage Sites	
	27 02 01 D	degree Bachelor's
	37.03.01 Psychology	
	27.02.02.03.03.01.4.1.4	degree
	37.03.02 Conflictology	Bachelor's
	05 05 04 CU 1 1 D 1 1	degree
	37.05.01 Clinical Psychology	Specialist's
		degree
	37.05.02 Employment Psychology	Specialist's
		degree
	40.03.01 Law	Bachelor's
		degree
	41.03.04 Political Science	Bachelor's
		degree
	42.03.01 Advertising and Public Relations	Bachelor's
		degree
	42.03.02 Journalism	Bachelor's
ъ.		degree
Russian	45.03.01 Philology	Bachelor's
		degree
	45.03.02 Linguistics	Bachelor's
	Č	degree
	47.03.01 Philosophy	Bachelor's
		degree
	47.03.03 Religious Studies	Bachelor's
		degree
	50.03.01 Arts and Humanities	Bachelor's
		degree
	50.03.03 History of Arts	Bachelor's
	bolobio History of this	degree
	51.03.01 Cultural Studies	Bachelor's
	21.00.01 Caltara Studios	degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's
	ividscology and conservation of cultural and ivaluati ficilitage sites	degree
	01.04.2001 Mathematics	Bachelor's
	01.07.2001 Maniemanes	degree
	01.03.03 Applied Mathematics and Informatics (programmes 'Modern	Bachelor's
	11 u u	
	Programming')	degree Bachelor's
	01.03.03 Mechanics and Mathematical Modelling	
	01.07.01 E 1 (1M.1) (1M.1) ((E.1) (1	degree
	01.05.01 Fundamental Mathematics and Mechanics (programme 'Fundamental	Specialist's
	Mechanics')	degree
Physics	02.03.01 Mathematics and Computer Sciences (programme 'Data Sciences')	Bachelor's
V		degree
	03.03.01 Applied Mathematics and Physics	Bachelor's
		degree
	03.03.02 Physics	Bachelor's
		degree
	03.03.03 Radiophysics	Bachelor's
		degree
	03.05.01 Astronomy	Specialist's
	· ·	degree

	04.03.02 Chemistry, Physics and Material Mechanics	Bachelor's
		degree
	05.04.2001 Geology	Bachelor's
		degree
	21.03.01 Petroleum Engineering	Bachelor's
		degree
	21/03/2002 Land Use and Cadastres	Bachelor's
	21/03/2002 Land Osc and Cadastics	
		degree
	27.03.03 Systems Analysis and Control	Bachelor's degree
Physical Education	49.02.01 Physical Education	Bachelor's
I nysicai Education	49.02.011 hysical Education	degree
	27 02 02 C - C - C - L - L	Bachelor's
	37.03.02 Conflictology	
		degree
	39.03.01 Sociology	Bachelor's
		degree
	39.03.02 Social Work	Bachelor's
	57.05.02 BOOM WORK	degree
	40.03.01 Law	Bachelor's
	40.05.01 Law	
		degree
	41.03.05 International Relations	Bachelor's
		degree
	43.03.02 Tourism	Bachelor's
	13.03.02 Tourism	degree
	45 02 04 DU 1	
	45.03.01 Philology (programmes 'Biblical Philology (Languages of the Bible and	Bachelor's
	Biblical Texts' 'Classical Philology (Old Greek and Latin; Ancient Literature', '	degree
	Russian Philology (Russian Language and Literature)')	
	45.03.02 Linguistics (programmes 'Albanian', 'Danish', 'Lithuanian', '	Bachelor's
French		
	Interlanguage Communication and Translation (French)', 'General and Applied	degree
	Phonetics', 'Russian as a Foreign Language', 'French')	
	47.03.01 Philosophy	Bachelor's
	47.03.011 imosophy	
	47.02.02 P. 11.1. G. 11	degree
	47.03.03 Religious Studies	Bachelor's
		degree
	50.03.01 Arts and Humanities	Bachelor's
		degree
	50.03.03 History of Arts	Bachelor's
	50.05.05 History of the	degree
	51.03.01 Cultural Studies	Bachelor's
	51.03.01 Cultural Studies	
		degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's
		degree
	58.03.01 Asian and African Studies	Bachelor's
		degree
	04.03.01 Chemistry	Bachelor's
	07.03.01 Chemistry	
	04.00.00.01	degree
	04.03.02 Chemistry, Physics and Material Mechanics	Bachelor's
		degree
	05.04.2001 Geology	Bachelor's
		degree
Chemistry	05.03.06 Ecology and Nature Management	Bachelor's
	obiobio 25010g) und riddio managoment	degree
	06 04 02 5-11 5-1	
	06.04.02 Soil Science	Bachelor's
		degree
	31.05.01 General Medicine	Specialist's
		degree
	05.03.06 Ecology and Nature Management	Bachelor's
	octobio Deology and Patente Management	
		degree
Ecology	06.04.00.0.10.1	D 1 1 1
Ecology	06.04.02 Soil Science	Bachelor's degree

	37.03.02 Conflictology	Bachelor's
		degree
	39.03.02 Social Work	Bachelor's
		degree
	47.03.01 Philosophy	Bachelor's
		degree
	47.03.03 Religious Studies	Bachelor's
		degree
	51.03.01 Cultural Studies	Bachelor's
		degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's
		degree
	37.03.02 Conflictology	Bachelor's
		degree
	38.03.01 Economics	Bachelor's
		degree
	38.03.02 Management	Bachelor's
		degree
	38.03.03 Personnel Management	Bachelor's
		degree
	38.03.04 Public and Municipal Administration	Bachelor's
		degree
	38.03.05 Business Informatics	Bachelor's
	Colocide Business information	degree
	39.03.01 Sociology	Bachelor's
Economics	e y rocket besteregy	degree
	39.03.02 Social Work	Bachelor's
	CONTROL CONTROL	degree
	41.03.05 International Relations	Bachelor's
	THOUSE THE TAXABLE OF	degree
	47.03.01 Philosophy	Bachelor's
		degree
	47.03.03 Religious Studies	Bachelor's
	<i>5</i>	degree
	51.03.01 Cultural Studies	Bachelor's
		degree
	Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's

Appendix 9 to the Regulations for Admission to Degree Programmes of Higher Education (Bachelor's Degree Programmes, Specialist's Degree Programmes, Master's Degree Programmes) at St Petersburg University in 2022

List of School Olympiads held in accordance with the Procedures for School Olympiads upon which results in particular rights shall be granted at admissions to St Petersburg University in 2022; correspondence between fields of study or specialities and degree programmes open for admission at St Petersburg University in 2022, olympiad profiles and the procedures for granting particular rights to school olympiads winners and awardees held in accordance with the Procedures for School Olympiads

Particular rights shall be granted to the winners and awardees of school olympiads held in 2021/2022, 2020/2021, 2019/2020, 2018/2019, 2017/2018 school years in accordance with the Procedures for School Olympiads and the List of School Olympiads approved by the Ministry of Education of the Russian Federation for the corresponding school year given they were **10th or 11th** form students at the time of the olympiad.

Main field of study title	Programme title	Type of study	AT No.	Disciplines of admissions tests (AT)	Maximum points to exercise the particular right	Lists of the Olympiads	Olympiad level	Profile or/and the full title of the Olympiad	General education Disciplines corresponding to the Olympiad profile	Subject areas (specialities) corresponding to the profile of the Olympiad	Category of the contestants	Particular rights to be granted to the winners and awardees of the Olympiad
01.00.00 MATHEMATICS AN	ID MECHANICS — BACH	ELOR'S DEG	REE									
01.04.2001 Mathematics	Mathematics	Full-time	1	Mathematics	80	Olympiads included in the lists approved by the Ministry of Education as of 31 August 2021	I	Vysshaia Proba All- Russia School Olympiad	Mathematics	-	Winner	admissions tests not required
						No 804, 27 August 2020 No 1125.	I	Moscow School Olympiad	Mathematics	_	Winner	admissions tests not required
						No. 658 dd. 30/08/2019 No. 32H dd. 28/08/2018 No. 866 dd. 30/08/2017	I	St Petersburg State University School Olympiad	Mathematics	-	Winners and awardees	admissions tests not required
							I	St Petersburg School Olympiad	Mathematics	_	Winners and awardees	admissions tests not required
							I	Turnir Gorodov	Mathematics		Winners and awardees	admissions tests not required

			2	Informatics and ICT	75	I	Vysshaia Proba All- Russia School Olympiad	Informatics	_	Winner	admissions tests not require
						I	All-Siberian Open School Olympiad	Informatics	_	Winner	admissions tests not required
						I	Moscow School Olympiad	Informatics	_	Winner	admissions tests not required
						I-II	Innopolis Open Innopolis University Olympiad	Informatics	_	Winner	admissions tests not required
						I	School Olympiad in Informatics and Programming	Informatics	_	Winner	admissions tests not required
						I	TechnoKubok School Olympiad in Programming	Informatics	_	Winner	admissions tests not required
						I	St Petersburg State University School Olympiad	Informatics	_	Winner	admissions tests not required
						I	Open School Olympiad in Programming	Informatics	_	Winner	admissions tests not required
			2	Physics	_			No partic	cular rights granted		
			3	Russian	_			No partic	cular rights granted		
01.03/.02 Applied Mathematics and Informatics	Mathematical Modelling,	Full-time	1	Mathematics	76	I-II	_	Mathematics	_	Winners and awardees	admissions tests not required
	Programming, and Artificial Intelligence		2	Informatics and ICT	-			No partic	cular rights granted		
			3	Russian	-			No partic	cular rights granted		
01.03/02 Applied Mathematics and Informatics	Applied Mathematics, Fundamental	Full-time	1	Informatics and ICT	75	I-III	-	Informatics	-	Winners and awardees	admissions tests not required
	Informatics and Programming					ш.	Kruzhkovoie Dvizheniie Olympiad of the National Technological Initiative Artificial Intelligence	-	Mathematics and Mechanics	Winners and awardees	admissions tests not required
			2	Mathematics	76	I-III	-	Mathematics	-	Winners and awardees	admissions tests not required

						III.	St Petersburg State University School Olympiad Engineering systems	-	Applied Mathematics and Informatics	Winners and awardees	admissions tests not required
			3	Russian	-			No partic	cular rights granted		
01.03/02 Applied Mathematics and Informatics	Modern Programming	Full-time	1	Mathematics	80	I	Vysshaia Proba All- Russia School Olympiad	Mathematics	-	Winner	admissions tests not required
						I	Moscow School Olympiad	Mathematics	-	Winner	admissions tests not required
						I	St Petersburg State University School Olympiad	Mathematics	-	Winner	admissions tests not required
						I	St Petersburg School Olympiad	Mathematics	-	Winner	admissions tests not required
						I	Turnir Gorodov	Mathematics	-	Winner	admissions tests not required
			2	Informatics and ICT	80	I	Vysshaia Proba All- Russia School Olympiad	Informatics	-	Winner	admissions tests not required
						I	Moscow School Olympiad	Informatics	-	Winner	admissions tests not required
						I-II	Innopolis Open Innopolis University Olympiad	Informatics	-	Winner	admissions tests not required
						I	School Olympiad in Informatics and Programming	Informatics	-	Winner	admissions tests not required
						I	TechnoKubok School Olympiad in Programming	Informatics	-	Winner	admissions tests not required
						I	St Petersburg State University School Olympiad	Informatics	-	Winner	admissions tests not required
						I	Open School Olympiad in Programming	Informatics	-	Winner	admissions tests not required
			3	Russian	-		I I	No partic	cular rights granted	ı	L
01.03.03 Mechanics and Mathematical Modelling	Mechanics and Mathematical Modelling	Full-time	1	Mathematics	76	I-III	-	Mathematics	-	Winners and awardees	admissions tests not required

							II-III	'Lomonosov' School Olympiad Mechanics and Mathematical Modelling	-	-	Winners and awardees	admissions tests not required
							III.	St Petersburg State University School Olympiad Engineering systems	-	Mechanics and Mathematical Modelling	Winners and awardees	admissions tests not required
			2	Informatics and ICT	75	-	I-III	-	Informatics	-	Winners and awardees	admissions tests not required
							III.	Kruzhkovoie Dvizheniie Olympiad of the National Technological Initiative Artificial Intelligence	-	Mathematics and Mechanics	Winners and awardees	admissions tests not required
			3	Russian	-	-			No partice	llar rights granted		
01.00.00 MATHEMATICS AN	D MECHANICS — SPECI	IALIST'S DEC	GREE	J.	J							
01.05.01 Fundamental Mathematics and Mechanics	Fundamental Mathematics	Full-time	1	Mathematics	76	Olympiads included in the lists approved by	I-II	-	Mathematics	-	Winners and awardees	admissions tests not required
			2	Physics	-	the Ministry of		•	No partice	ılar rights granted		
			3	Russian	-	Education as of 31 August 2021			No partici	ılar rights granted		
01.05.01 Fundamental	Fundamental Mechanics	Full-time	1	Mathematics	76	No 804, 27 August	I-III	1 _ 1	Mathematics	1 -	Winners and	admissions tests not required
Mathematics and Mechanics	rundamentai wiechames	run-time	1	Wathematics	70	2020 No 1125.	II-III	'Lomonosov' School	Mathematics	Fundamental	awardees Winners and	admissions tests not required
						No. 658/dd. 30/08/2019 No. 32H dd. 28/08/2018	11-111	Olympiad Mechanics and Mathematical Modelling	-	Mathematics and Mechanics	awardees	aumissions tests not required
			2	Physics	76	No. 866 dd. 30/08/2017	I-II	-	Physics	-	Winners and awardees	admissions tests not required
			3	Russian	-	1		<u></u>	No partice	ılar rights granted		
02.00.00 COMPUTER AND IN			OR'S DEGRI							•		
02.03.01 Mathematics and Computer Sciences	Mathematics and Computer Sciences	Full-time	1	Mathematics	76	Olympiads included in the lists approved by	I	-	Mathematics	-	Winners and awardees	admissions tests not required
			2	Informatics and ICT	-	the Ministry of Education as of 31			•	ılar rights granted		
			3	Russian	=	August 2021 No 804, 27 August			No partice	ılar rights granted		
02.03.01 Mathematics and Computer Sciences	Data Sciences	Full-time	1	Mathematics	80	2020 No 1125. No. 658 dd. 30/08/2019	I	Vysshaia Proba All- Russia School Olympiad	Mathematics	-	Winner	admissions tests not required
						No. 32H dd. 28/08/2018 No. 866 dd. 30/08/2017	I	Moscow School Olympiad	Mathematics	-	Winner	admissions tests not required
							I	St Petersburg State University School Olympiad	Mathematics	-	Winner	admissions tests not required

						I	St Petersburg School Olympiad	Mathematics	-	Winners and awardees	admissions tests not required
						I	Turnir Gorodov	Mathematics	-	Winners and awardees	admissions tests not required
			2	Informatics and ICT	80	I	Vysshaia Proba All- Russia School Olympiad	Informatics	-	Winner	admissions tests not required
						І-П	All-Siberian Open School Olympiad	Informatics	-	Winner	admissions tests not required
						I	Moscow School Olympiad	Informatics	-	Winner	admissions tests not required
						1-11	Innopolis Open Innopolis University Olympiad	Informatics	-	Winner	admissions tests not required
						I	School Olympiad in Informatics and Programming	Informatics	-	Winner	admissions tests not required
						I	TechnoKubok School Olympiad in Programming	Informatics	-	Winner	admissions tests not required
						I	St Petersburg State University School Olympiad	Informatics	-	Winner	admissions tests not required
						I	Open School Olympiad in Programming	Informatics	-	Winner	admissions tests not required
			3	Russian	-			No partic	cular rights granted		
02.03.02 Fundamental Informatics and Information	Big Data and Blockchain Digital	Full-time	1	Informatics and ICT	85	I-II	-	Informatics	-	Winners and awardees	admissions tests not required
Technology	Platform		2	Mathematics	86	I-II	-	Mathematics	-	Winners and awardees	100 points for ВИ informatics
			3	Russian	-			No partic	cular rights granted		•
02.03.02 Fundamental Informatics and Information	Programming and Information Technology	Full-time	1	Informatics and ICT	85	I-II	-	Informatics	-	Winners and awardees	admissions tests not required
Technology			2	Mathematics	86	I-II	-	Mathematics	-	Winners and awardees	100 points for ВИ informatics
			3	Russian	-			No partic	cular rights granted		•
	Technologies of Programming	Full-time	1	Mathematics	76	I-II	-	Mathematics	-	Winners and awardees	admissions tests not required

02.03.03 Software and Administration of Information			2	Informatics and ICT	75		I-II	-	Informatics	-	Winners and awardees	admissions tests not required
Systems			3	Russian	-				No partic	ular rights granted		
03.00.00 PHYSICS AND ASTI	RONOMY — BACHELOR	'S DEGREE										
03.03.01 Applied Mathematics and Physics	Engineering-oriented physics	Full-time	1	Physics	76	Olympiads included in the lists approved by	I-III	-	Physics	-	Winners and awardees	admissions tests not required
			2	Mathematics	76	the Ministry of Education as of 31	I-III	-	Mathematics	-	Winners and awardees	admissions tests not required
			3	Russian	76	August 2021 No 804, 27 August	I-III	-	Russian	-	Winners and awardees	100 points for ВИ in Russian
03.03.01 Applied Mathematics and Physics	Applied Physics and Mathematics	Full-time	1	Physics	76	2020 No 1125.	I-III	-	Physics	-	Winners and awardees	admissions tests not required
						No. 658 dd. 30/08/2019	I-III	Astronomy	-	Astronomy	Winners and awardees	admissions tests not required
			2	Mathematics	76	No. 32H dd. 28/08/2018	I-III	-	Mathematics	-	Winners and awardees	admissions tests not required
			3	Russian	76	No. 866 dd. 30/08/2017	I-III	-	Russian	-	Winners and awardees	100 points for ВИ in Russian
03.03.02 Physics	Physics	Full-time	1	Physics	76		I-III	-	Physics	-	Winners and awardees	admissions tests not required
03.03.03 Radiophysics	Electromagnetic and Acoustic Processes	Full-time					I-III	Astronomy	-	Astronomy	Winners and awardees	admissions tests not required
			2	Mathematics	76		I-III	-	Mathematics	-	Winners and awardees	admissions tests not required
			3	Russian	76		I-III	-	Russian	-	Winners and awardees	100 points for ВИ in Russian
03.00.00 PHYSICS AND ASTI	RONOMY — SPECIALIS	L'S DEGREE	1	1	l			1			I	
03.05.01 Astronomy	Astronomy	Full-time	1	Physics	76	Olympiads included in the lists approved by	I-II	-	Physics	-	Winners and awardees	admissions tests not required
						the Ministry of Education as of 31	I-III	Astronomy	-	Astronomy	Winners and awardees	admissions tests not required
			2	Mathematics	-	August 2021 No 804, 27 August			No partic	ular rights granted		
			3	Russian	-	2020 No 1125. No. 658 dd. 30/08/2019 No. 32n dd. 28/08/2018 No. 866 dd. 30/08/2017			No partic	ular rights granted		
04.00.00 CHEMISTRY — BA								_				T
04.03.01 Chemistry	Chemistry	Full-time	1	Chemistry	80	Olympiads included in the lists approved by	I	-	Chemistry	-	Winners and awardees	admissions tests not required
						the Ministry of Education as of 31	II.	-	Chemistry	-	Winner	admissions tests not required
						August 2021 No 804, 27 August	II.	-	Chemistry	-	Awardee	100 points for ВИ in Chemistry
			2 (Discipli	Mathematics	80	2020 No 1125.	I	-	Mathematics	-	Winners and awardees	admissions tests not required
			nes elected by the applicant			No. 658 dd. 30/08/2019 No. 32H dd. 28/08/2018	II.	-	Mathematics	-	Winners and awardees	100 points for BU informatics

			2 (Discipli	Physics	-	No. 866 dd. 30/08/2017			No part	cular rights granted		
			nes elected by the applicant									
			3	Russian	80		I-III	-	Russian	-	Winners and awardees	100 points for ВИ in Russian
04.03.02 Chemistry, Physics and Material Mechanics	Chemical Material Science	Full-time	1	Mathematics	80		I	-	Mathematics	-	Winners and awardees	admissions tests not required
							II.	-	Mathematics	-	Winners and awardees	100 points for ВИ informatics
			2 (Discipli	Chemistry	80		I-II	-	Chemistry	-	Winners and awardees	admissions tests not required
			nes elected				III.	-	Chemistry	=	Winners and awardees	100 points for ВИ in Chemistry
			by the applicant				пт.	St Petersburg State University School Olympiad Engineering systems	-	Chemistry, Physics and Material Mechanics	Winners and awardees	100 points for ВИ in Chemistry
							III.	'Lomonosov' School Olympiad Engineering Sciences	-	-	Winners and awardees	100 points for ВИ in Chemistry
							II.	NanoSystems and Nanoengineering	-	-	Winners and awardees	admissions tests not required
			(Discipli nes elected by the applicant	Physics	80		I-II	-	Physics	-	Winners and awardees	admissions tests not required
			3	Russian	80		I-III	-	Russian	-	Winners and awardees	100 points for ВИ in Russian
05.00.00 EARTH SCIENCES		Full-time	1	Mathamatica	7.6	Olympiade instruded in	1 111		Mothamatica		Winners and	admissions tests not received
05.04.2001 Geology	Geology	Full-time	1	Mathematics	76	Olympiads included in the lists approved by	I-III	-	Mathematics	-	Winners and awardees	admissions tests not required
						the Ministry of Education as of 31	I-III	Geology	-	Geology	Winners and awardees	admissions tests not required
			(Discipli nes elected by the applicant	Physics	76	August 2021 No 804, 27 August 2020 No 1125. No. 658 dd. 30/08/2019 No. 32n dd.	I-III	-	Physics	-	Winners and awardees	admissions tests not required
			2 (Discipli	Geography	78	28/08/2018 No. 866 dd.	I	-	Geography	-	Winners and awardees	admissions tests not required
			nes elected			30/08/2017	II-III	-	Geography	-	Winner	admissions tests not required
			by the applicant				II-III	-	Geography	-	Awardee	100 points for ВИ in Geography

			3	Russian	76		I-III	-	Russian	-	Winners and awardees	100 points for ВИ in Russian
05.03.02 Geography	Geography	Full-time	1	Geography	78	1	I	-	Geography	-	Winners and awardees	admissions tests not required
							II-III	-	Geography	-	Winner	admissions tests not required
							II-III	-	Geography	-	Awardee	100 points for ВИ in Geography
			2	Mathematics	76] [I-II	-	Mathematics	-	Winners and awardees	admissions tests not required
							III.	-	Mathematics	-	Winners and awardees	100 points for ВИ informatics
			3	Russian	76] [I-III	-	Russian	-	Winners and awardees	100 points for ВИ in Russian
05.03.03 Cartography and	Cartography and Geoinformatics	Full-time	1	Geography	78		I-II	-	Geography	-	Winner	admissions tests not required
Geoinformatics	Geomiormatics						I-II	-	Geography	-	Awardee	100 points for ВИ in Geography
							III.	-	Geography	-	Winners and awardees	100 points for ВИ in Geography
			2	Mathematics	76] [I-II	1	Mathematics	-	Winners and awardees	admissions tests not required
							III.	-	Mathematics	-	Winners and awardees	100 points for ВИ in Mathematics
			3	Russian	76		I-III	-	Russian	-	Winners and awardees	100 points for ВИ in Russian
05.03.04 Hydrometeorology	Hydrometeorology	Full-time	1	Geography	78		I-III	-	Geography	-	Winners and awardees	admissions tests not required
			2	Mathematics	76] [I-III	-	Mathematics	-	Winners and awardees	admissions tests not required
			3	Russian	76] [I-III	-	Russian	-	Winners and awardees	100 points for ВИ in Russian
05.03.06 Ecology and Nature Management	Ecology and Nature Management	Full-time	1	Geography	78	1 [I-II	-	Geography	-	Winner	admissions tests not required
- Ammgement	- Training errori						I-II	-	Geography	-	Awardee	100 points for ВИ in Geography
							III.	-	Geography	-	Winners and awardees	100 points for ВИ in Geography
							II-III	Ecology	-	Ecology and Nature Management	Winners and awardees	admissions tests not required
			2	Mathematics	76] [I-II	-	Mathematics	-	Winners and awardees	admissions tests not required
							III.	-	Mathematics	-	Winners and awardees	100 points for ВИ in Mathematics
			3	Russian	76		I-III	1	Russian	-	Winners and awardees	100 points for ВИ in Russian
06.00.00 BIOLOGICAL SCIE 06.04/.001 Biology	NCES — BACHELOR'S Biology	DEGREE Full-time	1	Piology	76	Olympiads included in	ī	_	Piology	<u> </u>	Winner	admissions tasts not required
00.04/.001 blology	Diology	run-time	1	Biology	/6	the lists approved by the Ministry of	I	-	Biology Biology	-	Awardee	admissions tests not required 100 points for BU in Biology
						Education as of 31	II.	-	Biology	-	Winner	100 points for BИ in Biology
			2	Chemistry	76	August 2021 No 804, 27 August 2020	I	-	Chemistry	-	Winners and	100 points for ВИ in
						2020 No 1125. No. 658 dd.	II.	-	Chemistry	-	awardees Winner	Chemistry 100 points for AT in
			3	Russian	76	30/08/2019 No. 32H dd.	I	-	Russian	-	Winners and	Chemistry 100 points for AT in Russian
						28/08/2018	II.	-	Russian	-	awardees Winner	100 points for AT in Russian
06.04.01 Biology	1	Full-time	1	Biology	76	No. 866 dd.	Ī	_	Biology	_	Winner	admissions tests not required

		-							1		1	•
							I	-	Biology	-	Awardee	100 points for AT in Biology
							II.	=	Biology	-	Winner	100 points for AT in Biology
	Biology: Bioengineering		2	Mathematics	76		I	-	Mathematics	-	Winners and awardees	100 points for AT in Mathematics
	Technologies						II.	-	Mathematics	-	Winner	100 points for AT in Mathematics
			3	Russian	76		I	-	Russian	-	Winners and awardees	100 points for AT in Russian
							II.	-	Russian	-	Winner	100 points for AT in Russian
06.04.02 Soil Science	Soil Science	Full-time	1	Biology	76		I-III	-	Biology	-	Winners and awardees	admissions tests not required
							II-III	Ecology	-	Ecology and Nature Management	Winners and awardees	admissions tests not required
			2	Mathematics	76		I-III	-	Mathematics	-	Winners and awardees	admissions tests not required
			3	Russian	76		I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
09.00.00 INFORMATICS AN	ND COMPUTER SCIENCE -	- BACHELOI	R'S DEGREI	E		•		•		•		•
Applied Informatics	Applied Informatics in Arts and Humanities	Full-time	1	Informatics and ICT	75	Olympiads included in the lists approved by	I-II	-	Informatics	-	Winners and awardees	admissions tests not required
			2	Mathematics	76	the Ministry of Education as of 31	I-II	-	Mathematics	-	Winners and awardees	admissions tests not required
			3	Russian	-	August 2021 No 804, 27 August			No partic	ular rights granted		
09.03.04 Software	Software Engineering	Full-time	1	Mathematics	-	2020			No partic	ular rights granted		
Engineering			2	Informatics and ICT	75	No 1125. No. 658 dd. 30/08/2019	I-II	-	Informatics	-	Winners and awardees	admissions tests not required
			3	Russian	-	No. 32H dd. 28/08/2018 No. 866 dd. 30/08/2017			No partic	ular rights granted		
21.00.00 APPLIED GEOLOG	GY, MINING, PETROLEUM	ENGINEERI	NG AND GE	EODESY — BACHELO	OR'S DEGREE	•						
21.03.01 Petroleum Engineering	Petroleum Engineering	Full-time	1	Mathematics	76	Olympiads included in the lists approved by the Ministry of Education as of 31	I-III	-	Mathematics	-	Winners and awardees	admissions tests not required
			2	Physics	76	August 2021 No 804, 27 August	I-III	-	Physics	-	Winners and awardees	admissions tests not required
			3	Russian	76	2020 No 1125.	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
21/03/2002 Land Use and Cadastres	Real Estate Cadastre Assessment and	Full-time	1	Mathematics	76	No. 658 dd. 30/08/2019	I-III	-	Mathematics	-	Winners and awardees	admissions tests not required
	Information Support		2	Geography	78	No. 32H dd. 28/08/2018 No. 866 dd.	I-III	-	Geography	-	Winners and awardees	admissions tests not required
			3	Russian	76	30/08/2017	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
27.00.00 CONTROL IN ENG			DEGREE	•		•						
27.03.03 Systems Analysis and Control	Applied Information Technology	Full-time	1	Mathematics	76	Olympiads included in the lists approved by	I-III	-	Mathematics	-	Winners and awardees	admissions tests not required
						the Ministry of Education as of 31 August 2021 No 804, 27 August 2020 No 1125. No. 658 dd. 30/08/2019	III.	St Petersburg State University School Olympiad Engineering systems	-	Systems Analysis and Control	Winners and awardees	admissions tests not required

						No. 32n dd. 28/08/2018 No. 866 dd. 30/08/2017	II.	Kruzhkovoie Dvizheniie Olympiad of the National Technological Initiative Automatisation of the Business Processes	-	Control in Engineering Systems	Winners and awardees	admissions tests not required
			2	Informatics and	75		I-III		Informatics		Winners and	
			2	ICT	75		1-111	=	informatics	-	awardees	admissions tests not required
31.00.00 CLINICAL MEDIC	INE CDECIALICTIC DE	CDEE	3	Russian	-				No part	cular rights granted		
31.05.01 General Medicine	General Medicine	Full-time	1	Biology	76	Olympiads included in the lists approved by	I	-	Biology	-	Winners and awardees	admissions tests not required
						the Ministry of Education as of 31	II-III	-	Biology	-	Winners and awardees	100 points for AT in Biology
						August 2021 No 804, 27 August	I	Medicine	-	General Medicine	Winners and awardees	admissions tests not required
						2020 No 1125.	II-III	Medicine	-	General Medicine	Winners and awardees	100 points for AT in Biology
			2	Chemistry	80	No. 658 dd. 30/08/2019 No. 32H dd.	I	-	Chemistry	-	Winners and awardees	admissions tests not required
						28/08/2018 No. 866 dd.	II-III	-	Chemistry	-	Winners and awardees	100 points for AT in Chemistry
			3	Russian	76	30/08/2017	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
31.05.03 Dental Medicine	Dentistry	Full-time	1	Biology	76		I	-	Biology	-	Winners and awardees	admissions tests not required
							II.	-	Biology	-	Winners and awardees	100 points for AT in Biology
							III.	-	Biology	-	Winner	100 points for AT in Biology
							I	Medicine	-	Dentistry	Winners and awardees	admissions tests not required
							II.	Medicine	-	Dentistry	Winners and awardees	100 points for AT in Biology
							III.	Medicine	-	Dentistry	Winner	100 points for AT in Biology
			2	Chemistry	80		I	-	Chemistry	-	Winners and awardees	admissions tests not required
						_	II.	-	Chemistry	-	Winner	100 points for AT in Chemistry
			3	Russian	76		I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian
37.00.00 PSYCHOLOGICAL 37.03.01 Psychology	Psychology	Full-time	1	Biology	76	Olympiads included in	Ī	-	Biology		Winners and	admissions tests not required
57.05.01 Lajenologj	1 Sychology	1 un time		Distogy	,,,	the lists approved by the Ministry of	II.	_	Biology		awardees Winner	admissions tests not required
						Education as of 31 August 2021	II.	-	Biology	-	Awardee	100 points for AT in Biology
						No 804, 27 August 2020	III.	-	Biology	-	Winners and	100 points for AT in Biology
						No 1125. No. 658 dd.	I	Psychology	-	-	awardees Winners and	admissions tests not required
						30/08/2019 No. 32н dd.	II.	Psychology	-	-	awardees Winner	admissions tests not required
			1	İ	İ	28/08/2018	II.	Psychology	=	_	Awardee	100 points for AT in Biology

						No. 866 dd. 30/08/2017	III.	Psychology	-	-	Winners and awardees	100 points for AT in Biology
							I	Medicine	-	Psychology	Winners and awardees	admissions tests not required
							II.	Medicine	-	Psychology	Winner	admissions tests not required
							II.	Medicine	-	Psychology	Awardee	100 points for AT in Biology
			2	Mathematics	76	1	I-III	-	Mathematics	-	Winners and awardees	100 points for AT in Mathematics
			3	Russian	76		I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
37.03.02 Conflictology	Conflictology	Full-time	1	Social Science	76	1	I	-	Social Science	-	Winners and awardees	admissions tests not required
							II.	-	Social Science	-	Winner	admissions tests not required
							II.	-	Social Science	-	Awardee	100 points for AT in Social Science
							I	Philosophy	Philosophy	-	Winners and awardees	admissions tests not required
							II.	Philosophy	Philosophy	-	Winner	admissions tests not required
						Ī	II.	Philosophy	Philosophy	-	Awardee	100 points for AT in Social Science
							I	Law	Law	-	Winners and awardees	admissions tests not required
							II.	Law	Law	-	Winner	admissions tests not required
							II.	Law	Law	-	Awardee	100 points for AT in Social Science
			2	History	75		I	-	History	-	Winner	admissions tests not required
			(Discipli nes elected				I	-	History	-	Awardee	100 points for AT in Social Science
			by the applicant				II.	-	History	-	Winners and awardees	100 points for AT in Social Science
			2	Foreign Language	75		I	-	Foreign Language	-	Winner	admissions tests not required
			(Discipli nes				I	-	Foreign Language	-	Awardee	100 points for AT in Russian
			elected by the applicant				II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			3	Russian	76	1	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
37.00.00 PSYCHOLOGICAL												
37.05.01 Clinical Psychology	Clinical Psychology	Full-time	1	Biology	76	Olympiads included in the lists approved by	I	-	Biology	-	Winners and awardees	admissions tests not required
						the Ministry of Education as of 31	II.	-	Biology	-	Winner	admissions tests not required
						August 2021	II.	-	Biology	-	Awardee	100 points for AT in Biology
						No 804, 27 August 2020	III.	-	Biology	-	Winners and awardees	100 points for AT in Biology
						No 1125. No. 658 dd. 30/08/2019	I	Psychology	-	-	Winners and awardees	admissions tests not required
						No. 32н dd.	II.	Psychology	-	-	Winner	admissions tests not required
						28/08/2018 No. 866 dd.	II.	Psychology	-	-	Awardee	100 points for AT in Biology
						30/08/2017	III.	Psychology	-	-	Winners and awardees	100 points for AT in Biology

							I	Medicine	-	Clinical Psychology	Winners and awardees	admissions tests not required
							II.	Medicine	-	Clinical Psychology	Winner	admissions tests not required
							II.	Medicine	=	Clinical Psychology	Awardee	100 points for AT in Biology
			2	Mathematics	76		I-III	-	Mathematics	-	Winners and awardees	100 points for AT in Mathematics
			3	Russian	76]	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
37.05.02 Employment Psychology	Employment Psychology	Full-time	1	Biology	76		I	-	Biology	-	Winners and awardees	admissions tests not required
							II.	=	Biology	=	Winner	admissions tests not required
							II.	-	Biology	-	Awardee	100 points for AT in Biology
							III.	-	Biology	-	Winners and awardees	100 points for AT in Biology
							I	Psychology	-	-	Winners and awardees	admissions tests not required
							II.	Psychology	=	=	Winner	admissions tests not required
							II.	Psychology	-	-	Awardee	100 points for AT in Biology
							III.	Psychology	-	-	Winners and awardees	100 points for AT in Biology
							Ι	Medicine	-	Employment Psychology	Winners and awardees	admissions tests not required
							II.	Medicine	-	Employment Psychology	Winner	admissions tests not required
							II.	Medicine	-	Employment Psychology	Awardee	100 points for AT in Biology
			2	Mathematics	76	1	I-III	-	Mathematics	-	Winners and awardees	100 points for AT in Mathematics
			3	Russian	76]	I-III	=	Russian	-	Winners and awardees	100 points for AT in Russian
38.00.00 ECONOMICS AND			REE	Laca	7.	T or - : : : 1 11: T		1	1 26.0		***	1 1
38.03.01 Economics	Economics	Full-time	1	Mathematics	76	Olympiads included in the lists approved by	I-II	-	Mathematics	-	Winner	admissions tests not required
						the Ministry of Education as of 31	I-II	=	Mathematics	=	Awardee	100 points for AT in Mathematics
						August 2021 No 804, 27 August	III.	-	Mathematics	-	Winners and awardees	100 points for AT in Mathematics
			2	Social Science	76	2020	I-II	-	Social Science	-	Winner	admissions tests not required
			(Discipli nes elected			No 1125. No. 658 dd.	I-II	-	Social Science	-	Awardee	100 points for AT in Social Science
			by the applicant			30/08/2019 No. 32H dd.	III.	=	Social Science	-	Winners and awardees	100 points for AT in Social Science
)			28/08/2018 No. 866 dd.	I-II	Economics	Economics	-	Winner	admissions tests not required
						30/08/2017	I-II	Economics	Economics	-	Awardee	100 points for AT in Social Science
							III.	Economics	Economics	-	Winners and awardees	100 points for AT in Social Science
			2 (Discipli	Geography	78]	I-II	-	Geography	-	Winners and awardees	100 points for ВИ in Geography
			nes elected by the									
		_1	by the	1	1	1		1	1		I.	

			applicant								
			3	Russian	76	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
38.03.01 Economics	Economics (with	Full-time	1	Mathematics	76	I-II	-	Mathematics	-	Winner	admissions tests not required
	Advanced Studies in Chinese Economy and Language)					I-II	-	Mathematics	-	Awardee	100 points for AT in Mathematics
	Language)					III.	-	Mathematics	-	Winners and awardees	100 points for AT in Mathematics
			(Discipli nes elected by the applicant	Foreign Language	75	I-III	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			(Discipli nes elected by the applicant	Geography	78	I-II	-	Geography	-	Winners and awardees	100 points for ВИ in Geography
			3	Russian	76	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
38.03.01 Economics	Economic and	Full-time	1	Mathematics	76	I-II	-	Mathematics	-	Winner	admissions tests not required
	mathematical methods					I-II	-	Mathematics	-	Awardee	100 points for AT in Mathematics
						III.	-	Mathematics	-	Winners and awardees	100 points for AT in Mathematics
			2 (Discipli	Social Science	76	I-II	=	Social Science	=	Winner	admissions tests not required
			nes elected			I-II	-	Social Science	-	Awardee	100 points for AT in Social Science
			by the applicant			III.	-	Social Science	-	Winners and awardees	100 points for AT in Social Science
)			I-II	Economics	Economics	-	Winner	admissions tests not required
						I-II	Economics	Economics	-	Awardee	100 points for AT in Social Science
						III.	Economics	Economics	-	Winners and awardees	100 points for AT in Social Science
			(Discipli nes elected by the applicant	Geography	78	I-II	-	Geography	-	Winners and awardees	100 points for ВИ in Geography
			3	Russian	76	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
38.03.02 Management	Management	Full-time	1	Mathematics	80	I	-	Mathematics		Winner	admissions tests not required
						I	-	Mathematics	-	Awardee	100 points for AT in Mathematics
			2	Foreign Language	80	I-II	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian

			3	Russian	80	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian
38.03.02 Management	International Management	Full-time	1	Mathematics	80	I	-	Mathematics	-	Winner	admissions tests not required
	Wanagement					I	-	Mathematics	-	Awardee	100 points for AT in Mathematics
			2	Foreign Language	80	I-II	-	Foreign Language	=	Winners and awardees	100 points for AT in Russian
			3	Russian	80	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian
38.03.03 Personnel Management	Personnel Management	Full-time	1	Mathematics	76	I-II	-	Mathematics	-	Winner	admissions tests not required
						I-II	-	Mathematics	-	Awardee	100 points for AT in Mathematics
						III.	-	Mathematics	-	Winners and awardees	100 points for AT in Mathematics
			2	Social Science	76	I-II	-	Social Science	-	Winner	admissions tests not required
						I-II	-	Social Science	=	Awardee	100 points for AT in Social Science
						III.	-	Social Science	-	Winners and awardees	100 points for AT in Social Science
						I-II	Economics	Economics	-	Winner	admissions tests not required
						I-II	Economics	Economics	-	Awardee	100 points for AT in Social Science
						III.	Economics	Economics	-	Winners and awardees	100 points for AT in Social Science
			3	Russian	76	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
38.03.04 Public and Municipal Administration	Public Administration	Full-time	1	Mathematics	80	Ī	-	Mathematics	-	Winner	admissions tests not required
Administration						I	-	Mathematics	-	Awardee	100 points for AT in Mathematics
			(Discipli nes elected by the applicant	Social Science	81	I-II	-	Social Science	-	Winners and awardees	100 points for AT in Social Science
			(Discipli nes elected by the applicant	Geography	83	I	-	Geography	-	Winners and awardees	100 points for BИ in Geography
			3	Russian	80	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian
38.03.05 Business Informatics	Business Informatics	Full-time	1	Mathematics	76	I-II	-	Mathematics	-	Winner	admissions tests not required
imormatics						I-II	-	Mathematics	-	Awardee	100 points for AT in Mathematics
						III.	-	Mathematics	-	Winners and awardees	100 points for AT in Mathematics
			2	Informatics and ICT	75	I-II	-	Informatics	-	Winner	admissions tests not required
				101		I-II	-	Informatics	-	Awardee	100 points for AT in Social Science
						III.	-	Informatics	=	Winners and awardees	100 points for AT in Social Science

			3	Russian	76		I-III	-	Russian	=	Winners and awardees	100 points for AT in Russian
39.00.00 SOCIOLOGY AN	D SOCIAL WORK — BACHE	LOR'S DEGI	REE									
39.03.01 Sociology	Sociology	Full-time	1	Social Science	76	Olympiads included in the lists approved by	I	-	Social Science	-	Winners and awardees	admissions tests not required
						the Ministry of Education as of 31	II.	-	Social Science	-	Winner	admissions tests not required
						August 2021 No 804, 27 August	II.	-	Social Science	-	Awardee	100 points for AT in Social Science
						2020 No 1125.	III.	-	Social Science	-	Winner	100 points for AT in Social Science
			2	Foreign Language	75	No. 658 dd. 30/08/2019	I	-	Foreign Language	-	Winner	admissions tests not required
						No. 32н dd.	I	-	Foreign Language	-	Awardee	100 points for AT in Russian
						28/08/2018 No. 866 dd.	II-III	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			3	Russian	76	30/08/2017	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
39.03.01 Sociology	Sociological Research in Digital Society	Full-time	1	Social Science	76	1	I	-	Social Science	-	Winners and awardees	admissions tests not required
							II.	-	Social Science	-	Winner	admissions tests not required
							II.	-	Social Science	-	Awardee	100 points for AT in Social Science
							III.	-	Social Science	-	Winners and awardees	100 points for AT in Social Science
			2 (Discipli	Mathematics	76	1	I	-	Mathematics	-	Winners and awardees	admissions tests not required
			nes elected				II.	-	Mathematics	-	Winner	admissions tests not required
			by the applicant				II.	=	Mathematics	-	Awardee	100 points for AT in Mathematics
)				III.	=	Mathematics	-	Winners and awardees	100 points for AT in Mathematics
			2 (Discipli	Informatics and ICT	75		I	-	Informatics	-	Winner	admissions tests not required
			nes elected	ici			I	-	Informatics	-	Awardee	100 points for AT in Social Science
			by the applicant				II-III	-	Informatics	-	Winners and awardees	100 points for AT in Social Science
			3	Russian	76	1	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
39.03.02 Social Work	Social Work	full-time	1	History	75		I-II	-	History	-	Winners and awardees	admissions tests not required
							III.	-	History	-	Winners and awardees	100 points for AT in Social Science
			2 (Discipli	Social Science	76		I-II	-	Social Science	=	Winners and awardees	admissions tests not required
			nes elected by the applicant				III.	-	Social Science	-	Winners and awardees	100 points for AT in Social Science
		1	2	Foreign Language	75	╡	I	-	Foreign Language	=	Winner	admissions tests not required
			(Discipli nes				I	-	Foreign Language	-	Awardee	100 points for AT in Russian
			by the applicant				II-III	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian

			3	Russian	76		I-III	-	Russian	-	Winners and	100 points for AT in Russian
											awardees	
40.00.00 LAW — BACHE		_										
40.03.01 Law	Law	full-time	1	Social Science	76	Olympiads included in the lists approved by	I	-	Social Science	-	Winner	admissions tests not required
						the Ministry of Education as of 31	I	=	Social Science	-	Awardee	100 points for AT in Social Science
						August 2021 No 804, 27 August	II.	=	Social Science	-	Winners and awardees	100 points for AT in Social Science
						2020	I	Law	Law	-	Winner	admissions tests not required
						No 1125. No. 658 dd.	I	Law	Law	-	Awardee	100 points for AT in Social Science
						30/08/2019 No. 32H dd.	II.	Law	Law	-	Winners and awardees	100 points for AT in Social Science
			2	Russian	76	28/08/2018 No. 866 dd. 30/08/2017	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
			3	History	75	30/08/2017	I	-	History	-	Winner	admissions tests not required
							I	-	History	-	Awardee	100 points for AT in Social Science
							II.	-	History	-	Winners and awardees	100 points for AT in Social Science
40.03.01 Law	Law	part-time	1	Social Science	76		I	-	Social Science	-	Winner	admissions tests not required
							I	-	Social Science	-	Awardee	100 points for AT in Social Science
							II.	=	Social Science	-	Winners and awardees	100 points for AT in Social Science
							I	Law	Law	-	Winner	admissions tests not required
							I	Law	Law	-	Awardee	100 points for AT in Social Science
							II.	Law	Law	-	Winners and awardees	100 points for AT in Social Science
			2	Russian	76	1	I-III	=	Russian	-	Winners and awardees	100 points for AT in Russian
			3	History	75		I	-	History	-	Winner	admissions tests not required
							I	-	History	-	Awardee	100 points for AT in Social Science
							II.	-	History	-	Winners and awardees	100 points for AT in Social Science
40.03.01 Law	Law (with Advanced Studies in Chinese Language and Law)	full-time	1	Social Science	76		I	-	Social Science	-	Winner	admissions tests not required
	Law (with Advanced Studies in Japanese	full-time					I	-	Social Science	-	Awardee	100 points for AT in Social Science
	Language and Law)						II.	=	Social Science	-	Winners and awardees	100 points for AT in Social Science
							I	Law	Law	-	Winner	admissions tests not required
							I	Law	Law	-	Awardee	100 points for AT in Social Science
							II.	Law	Law	-	Winners and awardees	100 points for AT in Social Science
			2	Russian	76	7	I-III	=	Russian	-	Winners and awardees	100 points for AT in Russian
			3	Foreign Language	75	7	I	-	Foreign Language	-	Winner	admissions tests not required
							I	=	Foreign Language	=	Awardee	100 points for AT in Russian

							II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
40.03.01 Law	Law (for postgraduates)	part-time	1	Legal Studies	-				No particul	ar rights granted		
41.00.00 POLITICAL SCIEN			HELOR'S DE									
41.03.02 Russian Region Studies	Russian Studies	full-time	1	History	75	Olympiads included in the lists approved by	I-III	-	History	-	Winners and awardees	admissions tests not required
			(Discipli nes elected by the applicant	Social Science	76	the Ministry of Education as of 31 August 2021 No 804, 27 August 2020 No 1125. No. 658 dd.	I-III	-	Social Science	-	Winners and awardees	100 points for AT in Social Science
			2 (Discipli nes elected by the applicant	Geography	75	30/08/2019 No. 32h dd. 28/08/2018 No. 866 dd. 30/08/2017	I-III	-	Geography	-	Winners and awardees	100 points for ВИ in Geography
			3	Russian	76		I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
41.03.04 Political Science	Political Science	full-time	1	History	75		I	-	History	-	Winners and awardees	admissions tests not required
							II.	-	History	-	Winner	admissions tests not required
							II.	-	History	-	Awardee	100 points for AT in Social Science
							III.	-	History	-	Winners and awardees	100 points for AT in Social Science
			2 (Discipli	Social Science	76		I	-	Social Science	-	Winners and awardees	admissions tests not required
			nes				II.	-	Social Science	-	Winner	admissions tests not required
			elected by the applicant				II.	-	Social Science	-	Awardee	100 points for AT in Social Science
)				III.	-	Social Science	-	Winners and awardees	100 points for AT in Social Science
			(Discipli nes elected by the applicant	Geography	78		I-III	-	Geography	-	Winners and awardees	100 points for ВИ in Geography
			3	Russian	76		I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
41.03.05 International	International Relations	full-time	1	History	75		I	-	History	-	Winner	admissions tests not required
Relations							I	-	History	-	Awardee	100 points for AT in Social Science
							II-III	-	History	-	Winners and awardees	100 points for AT in Social Science
			2 (Disainli	Foreign Language	84]	I	-	Foreign Language	-	Winner	admissions tests not required
			(Discipli nes elected				I	-	Foreign Language	-	Awardee	100 points for AT in Russian
			by the applicant				II-III	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian

			(Discipli nes elected by the applicant	Geography	83		I-III	-	Geography	-	Winners and awardees	100 points for ВИ in Geography
			3	Russian	76	1	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
42.00.00 MASS MEDIA AND	INFORMATION LIBRARI	IANSHIP — B	ACHELOR'S	S DEGREE		1		I			1	
42.03.01 Advertising and Public Relations	Advertising and Public Relations	full-time	1	Russian	76	Olympiads included in the lists approved by	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
			2	History	75	the Ministry of Education as of 31	I	-	History	-	Winner	admissions tests not required
						August 2021 No 804, 27 August	I	-	History	-	Awardee	100 points for AT in Social Science
						2020 No 1125.	II-III	-	History	-	Winners and awardees	100 points for AT in Social Science
			3	Social Science	76	No. 658 dd. 30/08/2019	I	=	Social Science	-	Winner	admissions tests not required
						No. 32H dd. 28/08/2018	I	-	Social Science	-	Awardee	100 points for AT in Social Science
						No. 866 dd. 30/08/2017	II-III	-	Social Science	-	Winners and awardees	100 points for AT in Social Science
42.03.02 Journalism	Journalism	full-time	1	Russian	76	30/08/2017	I-III	=	Russian	-	Winners and awardees	100 points for AT in Russian
							I	Journalism	-	Journalism	Winner	admissions tests not required
							I	Journalism	-	Journalism	Awardee	100 points for the creative admissions test
							II-III	Journalism	-	Journalism	Winners and awardees	100 points for the creative admissions test
			2	Literature	77	1	I	-	Literature	-	Winner	admissions tests not required
							I	-	Literature	-	Awardee	100 points for AT in Social Science
							II-III	-	Literature	-	Winners and awardees	100 points for AT in Social Science
			3	Practical, creative admissions test	75		I-III	Journalism	-	Journalism	Winners and awardees	100 points for the creative admissions test
42.03.02 Journalism	Journalism	part-time	1	Russian	76		I-III	=	Russian	-	Winners and awardees	100 points for AT in Russian
							I	Journalism	-	Journalism	Winners and awardees	admissions tests not required
							II-III	Journalism	-	Journalism	Winners and awardees	100 points for the creative admissions test
			2	Literature	77	1	I	-	Literature	-	Winner	admissions tests not required
							I	-	Literature	-	Awardee	100 points for AT in Social Science
							II-III	-	Literature	-	Winners and awardees	100 points for AT in Social Science
			3	Practical, creative admissions test	75		I-III	Journalism	-	Journalism	Winners and awardees	100 points for the creative admissions test
42.03.02 Journalism	International Journalism	full-time	1	Russian	76		I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
							I	Journalism	-	Journalism	Winner	admissions tests not required
							I	Journalism	-	Journalism	Awardee	100 points for the creative admissions test
							II-III	Journalism	=	Journalism	Winners and awardees	100 points for the creative admissions test

			2	Literature	77		I	-	Literature	-	Winner	admissions tests not required
							I	-	Literature	-	Awardee	100 points for AT in Social Science
							II-III	-	Literature	-	Winners and awardees	100 points for AT in Social Science
			3	Practical, creative admissions test	75		I-III	Journalism	-	Journalism	Winners and awardees	100 points for the creative admissions test
	TOURISM — BACHELOR'S			T T		T		ı		1	T	T
43.03.02 Tourism	The organisation of Tourist Activities (with Advanced Chinese Language Training)	full-time	(Discipli nes elected	Geography	78	Olympiads included in the lists approved by the Ministry of Education as of 31	I-II	-	Geography	-	Winners and awardees	admissions tests not required
	Tourism	full-time	by the applicant			August 2021 No 804, 27 August 2020	III.	-	Geography	-	Winners and awardees	100 points for ВИ in Geography
			1 (Discipli	Foreign Language	75	No 1125. No. 658 dd.	I-II	-	Foreign Language	-	Winners and awardees	admissions tests not required
			nes elected by the applicant			30/08/2019 No. 32н dd. 28/08/2018 No. 866 dd. 30/08/2017	III.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			2	History	75	30/00/2017	I-II	-	History	-	Winners and awardees	admissions tests not required
							III.	-	History	-	Winners and awardees	100 points for AT in Social Science
			3	Russian	76		I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
45.00.00 LINGUISTICS A	ND LITERARY STUDIES — I	BACHELOR'S	DECREE									
45.03.01 Philology		DACHELOK L	DEGREE									
	Biblical Philology (languages of the Bible	full-time	1	Literature	77	Olympiads included in the lists approved by	I	-	Literature	-	Winners and awardees	admissions tests not required
	Biblical Philology		1	Literature	77		II.	-	Literature	-	awardees Winner	admissions tests not required
	Biblical Philology (languages of the Bible		1	Literature	77	the lists approved by the Ministry of Education as of 31 August 2021	П.	-			awardees Winner Awardee	admissions tests not required 100 points for AT in Social Science
,	Biblical Philology (languages of the Bible		1	Literature		the lists approved by the Ministry of Education as of 31	II.	-	Literature	-	awardees Winner	admissions tests not required 100 points for AT in Social
,	Biblical Philology (languages of the Bible		2	Literature Russian	77	the lists approved by the Ministry of Education as of 31 August 2021 No 804, 27 August 2020 No 1125. No. 658 dd.	П.	-	Literature Literature	-	awardees Winner Awardee Winners and	admissions tests not required 100 points for AT in Social Science
,g	Biblical Philology (languages of the Bible		1			the lists approved by the Ministry of Education as of 31 August 2021 No 804, 27 August 2020 No 1125. No. 658 dd. 30/08/2019 No. 32n dd.	II. II. I-II	- - Philology	Literature Literature Literature	-	awardees Winner Awardee Winners and awardees	admissions tests not required 100 points for AT in Social Science admissions tests not required
,	Biblical Philology (languages of the Bible		1			the lists approved by the Ministry of Education as of 31 August 2021 No 804, 27 August 2020 No 1125. No. 658 dd. 30/08/2019 No. 32h dd. 28/08/2018 No. 866 dd.	II. II. III	- Philology	Literature Literature Literature Russian	-	awardees Winner Awardee Winners and awardees Winner	admissions tests not required 100 points for AT in Social Science admissions tests not required admissions tests not required
g,	Biblical Philology (languages of the Bible		1			the lists approved by the Ministry of Education as of 31 August 2021 No 804, 27 August 2020 No 1125. No. 658 dd. 30/08/2019 No. 32µ dd. 28/08/2018	II. II. II. III	- Philology	Literature Literature Literature Russian Russian	-	awardees Winner Awardee Winners and awardees Winner Awardee Winner Awardee	admissions tests not required 100 points for AT in Social Science admissions tests not required admissions tests not required 100 points for AT in Russian
g,	Biblical Philology (languages of the Bible		1			the lists approved by the Ministry of Education as of 31 August 2021 No 804, 27 August 2020 No 1125. No. 658 dd. 30/08/2019 No. 32h dd. 28/08/2018 No. 866 dd.	II. II. IIII IIII IIIIIIIIIIIIIIIIIIII	- Philology	Literature Literature Literature Russian Russian Russian		awardees Winner Awardee Winners and awardees Winner Awardee Winners and awardees Winners and awardees	admissions tests not required 100 points for AT in Social Science admissions tests not required admissions tests not required 100 points for AT in Russian 100 points for AT in Russian
,	Biblical Philology (languages of the Bible		2	Russian	76	the lists approved by the Ministry of Education as of 31 August 2021 No 804, 27 August 2020 No 1125. No. 658 dd. 30/08/2019 No. 32h dd. 28/08/2018 No. 866 dd.	II. II. IIII IIII IIII	Philology Philology	Literature Literature Literature Russian Russian Russian Russian Russian		awardees Winner Awardee Winners and awardees Winner Awardee Winners and awardees Winners and awardees Winners and awardees Winners and	admissions tests not required 100 points for AT in Social Science admissions tests not required admissions tests not required 100 points for AT in Russian 100 points for AT in Russian admissions tests not required
45.03.01 Philology	Biblical Philology (languages of the Bible and Biblical Texts) Hungarian and German		2	Russian	76	the lists approved by the Ministry of Education as of 31 August 2021 No 804, 27 August 2020 No 1125. No. 658 dd. 30/08/2019 No. 32h dd. 28/08/2018 No. 866 dd.	II. II. III. I III. III. III. III. III. III.	Philology Philology Philology -	Literature Literature Literature Russian Russian Russian Russian Foreign Language		awardees Winner Awardee Winners and awardees Winner Awardee Winners and awardees Winners and awardees Winners and awardees Winners and awardees Winners and awardees Winners and	admissions tests not required 100 points for AT in Social Science admissions tests not required admissions tests not required 100 points for AT in Russian 100 points for AT in Russian admissions tests not required
	Biblical Philology (languages of the Bible and Biblical Texts) Hungarian and German languages in the Austro-Hungarian	full-time	2	Russian Foreign Language	76 75	the lists approved by the Ministry of Education as of 31 August 2021 No 804, 27 August 2020 No 1125. No. 658 dd. 30/08/2019 No. 32h dd. 28/08/2018 No. 866 dd.	II. II. II. III. III. III. III. IIIIIIII	Philology Philology Philology Philology	Literature Literature Literature Russian Russian Russian Russian Foreign Language Foreign Language		awardees Winner Awardee Winners and awardees Winner Awardee Winners and awardees Winners and awardees Winners and awardees Winners and awardees Winners and awardees Winners and awardees Winners and awardees Winners and	admissions tests not required 100 points for AT in Social Science admissions tests not required admissions tests not required 100 points for AT in Russian 100 points for AT in Russian admissions tests not required 100 points for AT in Russian 100 points for AT in Russian
	Biblical Philology (languages of the Bible and Biblical Texts) Hungarian and German languages in the	full-time	2	Russian Foreign Language	76 75	the lists approved by the Ministry of Education as of 31 August 2021 No 804, 27 August 2020 No 1125. No. 658 dd. 30/08/2019 No. 32h dd. 28/08/2018 No. 866 dd.	II. II. II. I II. II. II. II. II	Philology	Literature Literature Literature Russian Russian Russian Russian Foreign Language Foreign Language Literature	- - - - - - -	awardees Winner Awardee Winners and awardees Winner Awardee Winners and awardees Winners and awardees Winners and awardees Winners and awardees Winners and awardees Winners and awardees Winners and awardees	admissions tests not required 100 points for AT in Social Science admissions tests not required admissions tests not required 100 points for AT in Russian 100 points for AT in Russian admissions tests not required 100 points for AT in Russian admissions tests not required 100 points for AT in Russian admissions tests not required

		1	2	Russian	76	I	-	Russian	-	Winner	admissions tests not required
						I	-	Russian	-	Awardee	100 points for AT in Russian
						II.	-	Russian	=	Winners and awardees	100 points for AT in Russian
						I-II	Philology	Russian	-	Winners and awardees	admissions tests not required
			3	Foreign Language	75	I-II	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
						I-II	Philology	Foreign Language	-	Winners and awardees	100 points for AT in Russian
45.03.01 Philology	Classical Philology (Old Greek and Latin;	full-time	1	Literature	77	I	-	Literature	-	Winners and awardees	admissions tests not required
	Ancient Literature)					II.	-	Literature	-	Winner	admissions tests not required
						II.	-	Literature	-	Awardee	100 points for AT in Social Science
						I-II	Philology	Literature	-	Winners and awardees	admissions tests not required
			2	Russian	76	I	-	Russian	-	Winner	admissions tests not required
						I	-	Russian	-	Awardee	100 points for AT in Russian
						II.	-	Russian	-	Winners and awardees	100 points for AT in Russian
						I-II	Philology	Russian	-	Winners and awardees	admissions tests not required
			3	Foreign Language	75	I-II	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
						I-II	Philology	Foreign Language	-	Winners and awardees	100 points for AT in Russian
45.03.01 Philology	Russian Philology (Russian Language and	full-time	1	Literature	77	I	-	Literature	-	Winners and awardees	admissions tests not required
	Literature)					II.	-	Literature	=	Winner	admissions tests not required
						II.	-	Literature	-	Awardee	100 points for AT in Social Science
						I-II	Philology	Literature	-	Winners and awardees	admissions tests not required
			2	Russian	76	I	=	Russian	=	Winner	admissions tests not required
						I	-	Russian	=	Awardee	100 points for AT in Russian
						II.	-	Russian	-	Winners and awardees	100 points for AT in Russian
						I-II	Philology	Russian	-	Winners and awardees	admissions tests not required
			3	Foreign Language	75	I-II	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
						I-II	Philology	Foreign Language	-	Winners and awardees	100 points for AT in Russian
45.03.02 Linguistics	Albanian	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not required
						I	-	Foreign Language	-	Awardee	100 points for AT in Russian
						II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			2	Russian	76	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian
			3	Literature	77	I-II	-	Literature	-	Winners and awardees	100 points for AT in Social Science
45.03.02 Linguistics	English	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not required

		7		1		I	-	Foreign Language	-	Awardee	100 points for AT in Russia
						II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russia
			2	Russian	76	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russia
			3	Literature	77	I-II	-	Literature	-	Winners and awardees	100 points for AT in Social Science
45.03.02 Linguistics	Danish	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not require
						I	-	Foreign Language	-	Awardee	100 points for AT in Russia
						II.	-	Foreign Language	=	Winners and awardees	100 points for AT in Russia
			2	Russian	76	I-II	-	Russian	Ē	Winners and awardees	100 points for AT in Russia
			3	Literature	77	I-II	-	Literature	-	Winners and awardees	100 points for AT in Social Science
45.03.02 Linguistics	Foreign Languages	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not require
	(Polish and German)					I	-	Foreign Language	-	Awardee	100 points for AT in Russia
						II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russia
			2	Russian	76	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russia
			3	Literature	77	I-II	-	Literature	-	Winners and awardees	100 points for AT in Social Science
45.03.02 Linguistics	Spanish	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not require
						I	-	Foreign Language	-	Awardee	100 points for AT in Russia
						II.	-	Foreign Language	=	Winners and awardees	100 points for AT in Russia
			2	Russian	76	I-II	-	Russian	=	Winners and awardees	100 points for AT in Russia
			3	Literature	77	I-II	-	Literature	Ē	Winners and awardees	100 points for AT in Social Science
45.03.02 Linguistics	Lithuanian	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not require
						I	-	Foreign Language	-	Awardee	100 points for AT in Russia
						II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russia
			2	Russian	76	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russia
			3	Literature	77	I-II	=	Literature	-	Winners and awardees	100 points for AT in Social Science
45.03.02 Linguistics	Interlanguage	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not require
	Communication and Translation (English)					I	-	Foreign Language	-	Awardee	100 points for AT in Russia
						II.	=	Foreign Language	=	Winners and awardees	100 points for AT in Russia
			2	Russian	76	I-II	=	Russian	-	Winners and awardees	100 points for AT in Russia
			3	Literature	77	I-II	-	Literature	-	Winners and awardees	100 points for AT in Social Science
45.03.02 Linguistics	Interlanguage	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not require
	Communication and Translation (German)					I	-	Foreign Language	-	Awardee	100 points for AT in Russia
						II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russia

			2	Russian	76	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian
			3	Literature	77	I-II	-	Literature	=	Winners and awardees	100 points for AT in Social Science
45.03.02 Linguistics	Interlanguage	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not required
	Communication and Translation (German)					I	-	Foreign Language	=	Awardee	100 points for AT in Russian
						II.	-	Foreign Language	=	Winners and awardees	100 points for AT in Russian
			2	Russian	76	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian
			3	Literature	77	I-II	-	Literature	-	Winners and awardees	100 points for AT in Social Science
45.03.02 Linguistics	German	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not required
						I	-	Foreign Language	-	Awardee	100 points for AT in Russian
						II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			2	Russian	76	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian
			3	Literature	77	I-II	-	Literature	-	Winners and awardees	100 points for AT in Social Science
45.03.02 Linguistics	General and Applied	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not required
	Phonetics					I	-	Foreign Language	-	Awardee	100 points for AT in Russian
						II.	-	Foreign Language	=	Winners and awardees	100 points for AT in Russian
			2	Russian	76	I-II	-	Russian	=	Winners and awardees	100 points for AT in Russian
			3	Literature	77	I-II	-	Literature	=	Winners and awardees	100 points for AT in Social Science
45.03.02 Linguistics	Applied, Computer and Computational	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not required
	Linguistics (English)					I	-	Foreign Language	-	Awardee	100 points for AT in Russian
						II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			2	Russian	76	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian
			3	Literature	77	I-II	-	Literature	=	Winners and awardees	100 points for AT in Social Science
45.03.02 Linguistics	Russian as a Foreign	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not required
	Language					I	-	Foreign Language	-	Awardee	100 points for AT in Russian
						II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			2	Russian	76	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian
			3	Literature	77	I-II	-	Literature	-	Winners and awardees	100 points for AT in Social Science
45.03.02 Linguistics	Comparative Linguistics (English)	full-time	1	Foreign Language	83	I	-	Foreign Language	-	Winner	admissions tests not required
	(English)					I	-	Foreign Language	-	Awardee	100 points for AT in Russian
						II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			2	Russian	76	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian
			3	Literature	77	I-II	-	Literature	-	Winners and awardees	100 points for AT in Social Science

45.03.02 Linguistics	Theoretical and Experimental	full-time	1	Foreign Language	83		I	-	Foreign Language	-	Winner	admissions tests not required	
	Linguistics (English)						I	-	Foreign Language	-	Awardee	100 points for AT in Russian	
							II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian	
			2	Russian	76	7	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian	
			3	Literature	77	7	I-II	-	Literature	-	Winners and awardees	100 points for AT in Social Science	
45.03.02 Linguistics	Theory and Practice of Intercultural	full-time	1	Foreign Language	83		I	=	Foreign Language	=	Winner	admissions tests not required	
	Communication						I	-	Foreign Language	=	Awardee	100 points for AT in Russian	
	(English)						II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian	
			2	Russian	76		, [I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian
			3	Literature	77		I-II	-	Literature	-	Winners and awardees	100 points for AT in Social Science	
45.03.02 Linguistics	French	full-time	1	Foreign Language	83		I	-	Foreign Language	=	Winner	admissions tests not required	
							I	-	Foreign Language	=	Awardee	100 points for AT in Russian	
							II.	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian	
			2	Russian	76		I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian	
			3	Literature	77		I-II	-	Literature	-	Winners and awardees	100 points for AT in Social Science	
46.00.00 HISTORY AND A	RCHAEOLOGY — BACHE	LOR'S DEGRE	E	<u>'</u>		· ·							
46.03.01 History	History	full-time	1	History	75	Olympiads included in the lists approved by	I	-	History	=	Winners and awardees	admissions tests not required	
						the Ministry of Education as of 31	II-III	-	History	=	Winners and awardees	100 points for AT in Social Science	
			2	Social Science	76	August 2021 No 804, 27 August	I-III	-	Social Science	-	Winners and awardees	100 points for AT in Social Science	
			3	Russian	76	2020 No 1125.	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian	
46.03.01 History	Archaeology	full-time	1	History	75	No. 658 dd. 30/08/2019	I	-	History	=	Winners and awardees	admissions tests not required	
						No. 32н dd. 28/08/2018	II-III	-	History	=	Winners and awardees	100 points for AT in Social Science	
			2	Social Science	76	No. 866 dd. 30/08/2017	I-III	-	Social Science	-	Winners and awardees	100 points for AT in Social Science	
			3	Russian	76		I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian	
47.00.00 PHII OSOPHY E	THICS, AND RELIGIOUS ST	FUDIES DA	CHELOD'S	DECREE									
47.03.01 Philosophy	Philosophy Philosophy	full-time	1	Social Science	76	Olympiads included in the lists approved by	I-II	-	Social Science	-	Winners and awardees	admissions tests not required	
						the Ministry of Education as of 31	I-II	Philosophy	Philosophy	-	Winners and awardees	admissions tests not required	
						August 2021	I-II	Law	Law	-	Winners and	admissions tests not required	

			(Discipli nes elected by the applicant	History	75	No 804, 27 August 2020 No 1125. No. 658 dd. 30/08/2019 No. 32H dd. 28/08/2018	I-III	-	History	-	Winners and awardees	100 points for AT in Social Science
			2 (Discipli nes elected by the applicant	Foreign Language	75	No. 866 dd. 30/08/2017	I-III	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			3	Russian	76		I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
47.03.03 Religious Studies	Religious Studies	full-time	1	History	75		I	-	History	-	Winners and awardees	admissions tests not required
							II.	=	History	-	Winner	admissions tests not required
							II.	-	History	-	Awardee	100 points for AT in Social Science
			2 (Discipli	Social Science	76		I	-	Social Science	-	Winners and awardees	admissions tests not required
			nes elected				II-III	-	Social Science	=	Winners and awardees	100 points for AT in Social Science
			by the applicant				I-II	Philosophy	Philosophy	=	Winners and awardees	100 points for AT in Social Science
)				I-II	Law	Law	=	Winners and awardees	100 points for AT in Social Science
			2 (Discipli	Foreign Language	75		I	-	Foreign Language	-	Winners and awardees	admissions tests not required
			nes elected by the applicant				II-III	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			3	Russian	76		I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
49.00.00 PHYSICAL EDUCA			GREE	L no a				I		1	1	
49.02.01 Physical Education	Physical Culture and Sports	full-time	1	Biology	76	Olympiads included in the lists approved by	I-III	-	Biology	-	Winners and awardees	100 points for AT in Biology
			2	Physical Education	-	the Ministry of Education as of 31				ılar rights granted		
			3	Russian	76	August 2021 No 804, 27 August 2020 No 1125. No. 658 dd. 30/08/2019 No. 32n dd. 28/08/2018 No. 866 dd. 30/08/2017	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
50.00.00 ART STUDIES — BA	ACHELOR'S DEGREE											

50.03.01 Arts and Humanities	Liberal Arts and	full-time	1	History	75	Olympiads included in	I	-	History	-	Winner	admissions tests not required
	Sciences					the lists approved by the Ministry of Education as of 31	I	-	History	-	Awardee	100 points for AT in Social Science
						August 2021 No 804, 27 August	II-III	-	History	-	Winners and awardees	100 points for AT in Social Science
			2	Foreign Language	75	2020 No 1125.	I-III	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			3	Russian	76	No. 658 dd. 30/08/2019	I-III	-	Russian	Ü	Winners and awardees	100 points for AT in Russian
50.03.03 History of Arts	History of Arts	full-time	1	History	75	No. 32н dd. 28/08/2018	I	-	History	Ü	Winners and awardees	admissions tests not required
						No. 866 dd. 30/08/2017	II-III	-	History	Ü	Winners and awardees	100 points for AT in Social Science
							I-II	Academic drawing, painting, composition, history of arts and culture.	-	-	Winner	admissions tests not required
							I-II	Academic drawing, painting, composition, history of arts and culture.	-	-	Awardee	100 points for AT in Social Science
			2	Foreign Language	75	_	I-III	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			3	Russian	76		I-III	-	Russian	=	Winners and awardees	100 points for AT in Russian
51.00.00 CULTURAL STUDII	ES AND SOCIO-CULTUR	RAL PROJECTS	S — BACHE	LOR'S DEGREE								
51.03.01 Cultural Studies	Cultural Studies	full-time	1	Social Science	76	Olympiads included in the lists approved by	I	-	Social Science	-	Winners and awardees	admissions tests not required
						the Ministry of	II.	-	Social Science	-	Winner	admissions tests not required
						Education as of 31 August 2021 No 804, 27 August	II.	-	Social Science	-	Awardee	100 points for AT in Social Science
						2020 No 1125.	I	Philosophy	Philosophy	=	Winners and awardees	admissions tests not required
						No. 658 dd.	II.	Philosophy	Philosophy	-	Winner	admissions tests not required
						30/08/2019 No. 32н dd. 28/08/2018	II.	Philosophy	Philosophy	=	Awardee	100 points for AT in Social Science
						No. 866 dd. 30/08/2017	I	Law	Law	-	Winners and awardees	admissions tests not required
						30/00/2017	II.	Law	Law	-	Winner	admissions tests not required
							II.	Law	Law	-	Awardee	100 points for AT in Social Science
			2 (Discipli	History	75		I-III	-	History	-	Winners and awardees	100 points for AT in Social Science
			nes elected				I-II	Academic drawing, painting,	-	-	Winners and awardees	100 points for AT in Social Science

			2 (Discipli nes elected by the applicant	Foreign Language	75		I-III	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			3	Russian	76	1	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
Museology and Conservation of Cultural and Natural	Museology and Conservation of Cultural	full-time	1	History	75	1	I	-	History	-	Winners and awardees	admissions tests not required
Heritage Sites	and Natural Heritage Sites						II.	-	History	-	Winner	admissions tests not required
	Sites						II.	-	History	-	Awardee	100 points for AT in Social Science
							I-II	Academic drawing, painting, composition, history of arts and culture.	-	-	Winner	admissions tests not required
							I-II	Academic drawing, painting, composition, history of arts and culture.	-	-	Awardee	100 points for AT in Social Science
			2	Social Science	76	-	I-III	-	Social Science	-	Winners and	100 points for AT in Social
			(Discipli nes			-	I-II	Philosophy	Philosophy	-	awardees Winners and	Science 100 points for AT in Social
			elected								awardees	Science
			by the applicant				I-II	Law	Law	-	Winners and awardees	100 points for AT in Social Science
			(Discipli nes elected by the applicant	Foreign Language	75		I-III	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian
			3	Russian	76	1	I-III	-	Russian	-	Winners and awardees	100 points for AT in Russian
52.00.00 PERFORMING ART			IALIST'S DE									
52.05.01 Acting	Stage and Film Actor	full-time	1	Literature	-	Olympiads included in the lists approved by				ılar rights granted		
			2	Russian	-	the Ministry of Education as of 31				ılar rights granted		
			3	Stage Speech	-	August 2021				ılar rights granted		
			4	Plastique and Vocals	-	No 804, 27 August 2020			No particu	ılar rights granted		
53.00.00 MINVS 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1.	ACHELODIS PECONO		5	Acting Skills	-	No 1125. No. 658 dd. 30/08/2019 No. 32H dd. 28/08/2018 No. 866 dd. 30/08/2017			No particu	ılar rights granted		
53.00.00 MUSICAL ART — B. 53.03.02 Instrumental Music	ACHELOR'S DEGREE	full-time	1	Literature	_				No parties	ılar rights granted		
55.05.02 HISHUIHEIRAI MUSIC		run-time	1	Literature	-				No particu	ma rigins granteu		

			-				
	To at many and all		2	Russian	=		No particular rights granted
	Instrumental Performance: Organ,		3	Solo playing	-		No particular rights granted
	Harpsichord and Carillon		4	Solfège	-		No particular rights granted
	Carmon		5	Harmony	-	Olympiads included in	No particular rights granted
53.03.02 Instrumental Music	Violin	full-time	1	Literature	-	the lists approved by the Ministry of	No particular rights granted
			2	Russian	-	Education as of 31 August 2021	No particular rights granted
			3	Solo playing	-	No 804, 27 August 2020	No particular rights granted
			4	Solfège	-	No 1125.	No particular rights granted
			5	Harmony	-	No. 658 dd. 30/08/2019	No particular rights granted
53.03.03 Vocal Performance	Academic Singing	full-time	1	Literature	-	No. 32н dd. 28/08/2018	No particular rights granted
			2	Russian	-	No. 866 dd.	No particular rights granted
			3	Solo playing	-	30/08/2017	No particular rights granted
			4	Basics of the Artistry	-		No particular rights granted
			5	Theory of Music	-	-	No particular rights granted
54.00.00 FINE AND APPLIED	DARTS — BACHELOR'S	DEGREE	1	and Solfège			L
54.03.01 Design	Graphic Design	full-time	1	Literature	-	Olympiads included in	No particular rights granted
			2	Russian	-	the lists approved by the Ministry of	No particular rights granted
			3	Composition	-	Education as of 31 August 2021	No particular rights granted
			4	Drawing	-	No 804, 27 August 2020	No particular rights granted
			5	Painting	-	No 1125.	No particular rights granted
54.03.01 Design	Environmental Design	full-time	1	Literature	-	No. 658 dd. 30/08/2019	No particular rights granted
			2	Russian	-	No. 32н dd. 28/08/2018	No particular rights granted
			3	Composition	-	No. 866 dd. 30/08/2017	No particular rights granted
			4	Drawing	-	30/08/2017	No particular rights granted
			5	Painting	-		No particular rights granted
54.03.02 APPLIED ART AND VERNACULAR ARTS	Applied Art	full-time	1	Literature	-	1	No particular rights granted
AND VERNACULAR ARTS			2	Russian	-	1	No particular rights granted
			3	Drawing	-		No particular rights granted
			4	Painting	-	1	No particular rights granted
			5	Composition	-	1	No particular rights granted
54.03.04 Restoration	Restoration of Fine and	full-time	1	Literature	-	1	No particular rights granted
	Applied Art Objects		2	Russian	-	1	No particular rights granted
			3	Painting	-	1	No particular rights granted
			4	Drawing	-		No particular rights granted
54.00.00 FINE AND APPLIED	ARTS — SPECIALIST'S						
54.05.02 Painting	Easel painting	full-time	1	Literature	-	Olympiads included in the lists approved by	No particular rights granted
			2	Russian	-	the Ministry of	No particular rights granted

			3	Painting	-	Education as of 31			No particular	rights granted				
			4	Drawing	-	August 2021 No 804, 27 August			No particular	rights granted				
			5	Composition	-	2020 No 1125.			No particular	rights granted				
54.05.02 Painting	Film and TV Costume	full-time	1	Literature	-	No. 658 dd. 30/08/2019	No particular rights granted							
	Designer		2	Russian	-	No. 32н dd.	No particular rights granted							
			3	Painting	-	28/08/2018 No. 866 dd.			No particular	rights granted				
			4	Drawing	-	30/08/2017	No particular rights granted							
			5	Composition	-	1			No particular	rights granted				
58.00.00 ASIAN AND AFRIC	CAN STUDIES — BACHEL	OR'S DEGRE	E											
58.03.01 Asian and African Studies	Arabic Philology	full-time	1	History	75	Olympiads specified in the lists approved by	I	-	History	=	Winners and awardees	admissions tests not required		
	Hebrew Studies	full-time				the Ministry of Education: No 804	II.	-	History	-	Winner	admissions tests not required		
	Indo-Aryan (Bengali) Philology	full-time				dated 31 August 2021, No 1125 dated 27	II.	-	History	-	Awardee	100 points for AT in Social Science		
	Iranian Philology	full-time				August 2020, No 658 dated 30 August 2019,	III.	-	History	-	Winners and awardees	100 points for AT in Social Science		
	History of Caucasus (Armenia)	full-time	2	Russian	76	No 32-n dated 28 August 2018, No 865	I-II	-	Russian	-	Winners and awardees	100 points for AT in Russian		
	History of China	full-time	3	Foreign Language	75	dated 30 August 2017	I-III	-	Foreign Language	-	Winners and awardees	100 points for AT in Russian		
	History of Korea	full-time									awaruces			
	History of Japan	full-time												
	Mongolian and Tibetan Philology	full-time												
	Turkic Philology	full-time	1											
	Languages and Cultures of East Africa (Hausa Language)	full-time												
	Languages, History and Culture of Ancient Egypt (Egyptology)	full-time												